

Université de Liège Faculté des Sciences Département de Mathématique

# On Generalized Hölder-Zygmund Spaces

Promoteur : Prof. S. Nicolay Co-promoteur : Prof. F. Bastin Composition du Jury : Prof. S. Jaffard Prof. J.-P. Schneiders Prof. J. Wengenroth Thèse présentée en vue de l'obtention du grade de Docteur en Sciences par : Damien KREIT

# Remerciements

J'adresse mes remerciements aux personnes qui m'ont aidé ou soutenu lors de la réalisation de cette thèse. En premier lieu, à mon promoteur de thèse, Monsieur le Professeur Samuel Nicolay, pour m'avoir offert sa confiance et l'opportunité de suivre la formation doctorale, mais aussi pour son support, sa disponibilité et ses qualités humaines de compréhension et d'écoute.

Merci à Madame le Professeur Françoise Bastin qui a accepté d'être ma co-promotrice, et à Messieurs les Professeurs Stéphane Jaffard, Jochen Wengenroth et Jean-Pierre Schneiders qui ont accepté de faire partie de mon jury de thèse. Je souhaite exprimer à Monsieur le Professeur Jochen Wengenroth ma plus profonde gratitude pour m'avoir donné le goût de la recherche lors de mes dernières années d'étude en master de sciences mathématiques. Sans lui, je ne me serais peut-être jamais orienté vers une thèse de doctorat, et je le remercie pour cette belle et riche expérience.

Un merci particulier à mon épouse Marie Blaffart, à mes parents Patrick et Krystyna Kreit, et à ma grand-mère Danuta Cybulska, qui n'ont jamais cessé de croire en moi et qui m'ont toujours fourni le courage et le soutien nécessaire pour poursuivre dans la voie que j'ai choisie. J'adresse une pensée particulière au reste de ma famille, et à ceux qui ne sont plus là.

Merci aussi à toutes les personnes que j'ai croisées durant mon parcours, et qui, d'une façon ou d'une autre, ont contribué à sa réussite : merci à Marianne Clausel, Julien Hamonier, Qidi Peng, Mehdi Madani, Rukiye Cavus,... ainsi qu'aux nombreuses autres personnes que je n'ai pas citées.

Ces dernières années furent une expérience de vie inoubliable que je n'hésiterai pas à renouveler.

## Abstract

To study the regularity of functions, many functional spaces have been introduced during the 20<sup>th</sup> century. Among them, let us mention the Hölder-Zygmund spaces  $\Lambda^{\alpha}(\mathbb{R}^d)$  and the Besov spaces  $B_{p,q}^{\alpha}(\mathbb{R}^d)$  where  $\alpha > 0$  somehow indicates the regularity of their elements  $(p, q \in ]0, +\infty]$ ). The Hölder-Zygmund spaces are particular cases of Besov spaces in the sense that  $\Lambda^{\alpha}(\mathbb{R}^d) = B_{\infty\infty}^{\alpha}(\mathbb{R}^d)$ .

A generalization of Besov spaces has been introduced in the middle of the seventies and is still studied nowadays. This new type of space allows a deepest study of the regularity of functions (see e.g. [50]). In this thesis, we start from this generalization in order to introduce a generalization of Hölder-Zygmund spaces.

The first aim of this thesis is to show that most classical properties of Hölder-Zygmund spaces can be transposed to their generalized version. Among others, a complete characterization of these spaces in terms of wavelet coefficients is proved, which opens their use in the context of the signal analysis.

The second aim of this thesis is to introduce a generalized version of the pointwise Hölder spaces similarly to their global version. We then show that most properties of the global spaces can be transposed to their generalized pointwise version.

Finally, we study the regularity of some financial stochastic processes.

## Résumé

Afin d'étudier la régularité des fonctions, divers espaces fonctionnels ont été introduits au  $XX^{\text{ème}}$  siècle. Parmis eux, citons les espaces de Hölder-Zygmund  $\Lambda^{\alpha}(\mathbb{R}^d)$  et les espaces de Besov  $B_{p,q}^{\alpha}(\mathbb{R}^d)$ , où  $\alpha > 0$  indique d'une certaine manière la régularité de leurs éléments  $(p, q \in ]0, +\infty]$ ). Les espaces de Hölder-Zygmund sont des cas particuliers d'espaces de Besov, en ceci que  $\Lambda^{\alpha}(\mathbb{R}^d) = B_{\infty\infty}^{\alpha}(\mathbb{R}^d)$ .

Une généralisation des espaces de Besov a été introduite durant les années 1970 et est encore étudiée aujourd'hui. Ce nouveau type d'espace permet une étude plus approfondie de la régularité (voir par ex. [50]). Dans cette thèse, nous partons de cette généralisation pour introduire une généralisation des espaces de Hölder-Zygmund.

Le premier objectif de cette thèse est de démontrer que la plupart des propriétés classiques des espaces de Hölder-Zygmund peuvent être transposées à leur version généralisée. Entre autre, une caractérisation complète de ces espaces en terme de coefficients en ondelettes est démontrée, ce qui ouvre leur utilisation dans le contexte de l'analyse du signal.

Le second objectif de cette thèse est d'introduire une version généralisée des espaces de Hölder ponctuels, de manière similaire à leur version globale. Nous démontrons ensuite que la plupart des propriétés et caractérisations des espaces globaux peuvent être transposées à leur version ponctuelle généralisée.

Diverses applications d'étude de régularité de processus stochastiques financiers seront entre autres illustrées.

# Contents

In	troduction	1
1	Origins of the study of generalized Hölder-Zygmund spaces – State of the Art         1.1       Background: reminder of basic concepts	
2	Characterizations and basic properties of generalized Hölder-Zygmund spaces $\Lambda^{\sigma,\alpha}(\mathbb{R}^d)$ 2.1 A reminder of the characterizations of Hölder-Zygmund spaces $\Lambda^{\alpha}(\mathbb{R}^d)$ 2.2 Some preliminary results	<b>27</b> 29 31 33 34 36 37 43 50
3	<ul> <li>2.10 Generalized Hölder-Zygmund spaces Λ<sup>σ,α</sup>(ℝ<sup>a</sup>) and generalized (real) interpolation of Sobolev spaces</li> <li>2.11 A weak result of Lions-Peetre type for spaces Λ<sup>σ,α</sup>(ℝ<sup>d</sup>)</li> <li>Some applications of the generalized Hölder-Zygmund spaces</li> <li>3.1 Generalized Hölder exponents</li> <li>3.1.1 Preliminary results</li> <li>3.1.2 Decreasing generalized Hölder spaces</li> </ul>	53 58 <b>60</b> 60 60 63

#### CONTENTS

	3.2	Another definition of Hölder-Zygmund spaces	67									
	3.3	The uniform irregular Hölder spaces $I^{\alpha}(\mathbb{R}^d)$ expressed as a particular case										
		of generalized Hölder spaces $\Lambda^{\sigma,\alpha}(\mathbb{R}^d)$	68									
	3.4	Application to financial models	71									
4	Characterizations and properties of pointwise generalized Hölder-Zygmund											
	spa	ces	75									
	4.1	Definition of pointwise generalized Hölder-Zygmund spaces $\Lambda^M_{\sigma,N}(x_0)$ and										
		$\Lambda^{\sigma,M}(x_0)$	75									
	4.2	A characterization of pointwise generalized Hölder-Zygmund spaces in terms										
		of finite differences	77									
	4.3	A characterization of pointwise generalized Hölder-Zygmund spaces in terms										
		of Taylor decomposition	78									
	4.4	A characterization of pointwise generalized Hölder-Zygmund spaces in terms										
		of wavelet coefficients	82									
	4.5	A characterization of pointwise generalized Hölder-Zygmund spaces in terms										
		of the convolution product	86									
	4.6	An application of generalized Hölder-Zygmund spaces: the Takagi function	89									
	4.7	Generalized pointwise Hölder exponent	90									
<b>5</b>	App	pendix	95									
	5.1	About subadditive sequences	95									
	5.2	Some inequalities from S. Bernstein	97									
	5.3	About the Littlewood-Paley decomposition	97									
	5.4	Some reminders about wavelets	98									
	5.5	Some reminders about the interpolation theory	100									

vi

#### Some basic notations

- $\mathbb{N}_0 = \{0, 1, 2, 3, ...\}$  denotes the set of natural numbers.
- $\mathbb{N}^* = \{1, 2, 3, ...\}.$
- $\mathbb{Z}$  denotes the set of integers.
- $\mathbb{R}$  denotes the set of real numbers.
- $\mathbb{R}^d = \{(x_1, ..., x_d) : x_i \in \mathbb{R} \mid \forall i \in \{1, ..., d\}\}$  denotes the *d*-dimensional euclidean space  $(d \in \mathbb{N}^*)$ .
- $\mathbb{N}_0^d = \{(\alpha_1, ..., \alpha_d) : \alpha_i \in \mathbb{N}_0 \quad \forall i \in \{1, ..., d\}\}$  denotes the set of multi-indices  $(d \in \mathbb{N}^*)$ .
- B(x, R) denotes the open ball with center  $x \in \mathbb{R}^d$  and radius R > 0.
- $B(x, \leq R)$  denotes the closed ball with center  $x \in \mathbb{R}^d$  and radius R > 0.
- $\lfloor x \rfloor = \sup\{m \in \mathbb{Z} : m \le x\}$  denotes the floor of  $x \in \mathbb{R}$ .
- $\lceil x \rceil = \inf\{m \in \mathbb{Z} : m \ge x\}$  denotes the ceiling of  $x \in \mathbb{R}$ .
- $(x)_{+} = \max\{x, 0\}$  denotes the positive part of a real number  $x \in \mathbb{R}$ .
- $\alpha$ ! denotes the value  $\alpha$ ! =  $\alpha_1$ !... $\alpha_d$ ! if  $\alpha \in \mathbb{N}_0^d$  is a multi-index.
- $|\alpha|$  denotes the value  $\alpha_1 + \ldots + \alpha_d$  if  $\alpha \in \mathbb{N}_0^d$  is a multi-index.
- $\binom{m}{j} = \frac{m!}{(m-j)!j!}$  where  $m, j \in \mathbb{N}_0$  and  $m \ge j$ .
- C(A) denotes the space of continuous functions defined on  $A \subseteq \mathbb{R}^d$ .
- $C^p(\Omega)$   $(p \in \mathbb{N}_0 \cup \{\infty\})$  denotes the space of functions which are *p*-times continuously differentiable on  $\Omega$  (where  $\Omega$  is an open set of  $\mathbb{R}^d$ ).
- $D(\Omega)$  denotes the subspace of  $C^{\infty}(\Omega)$  made from compactly supported functions on  $\Omega \subseteq \mathbb{R}^d$ .
- $L^{p}(A)$  denotes the space of measurable functions on A satisfying  $\|f\|_{L^{p}(A)} = \left(\int_{A} |f(x)|^{p} dx\right)^{1/p} < \infty$  (where  $p \in ]0, \infty[$  and A is a measurable set of  $\mathbb{R}^{d}$ ).
- $L^{\infty}(A)$  denotes the space of measurable functions on A satisfying  $||f||_{L^{\infty}(A)} = \sup_{ppA} |f| < \infty$  (where A is a measurable set of  $\mathbb{R}^{d}$ ).
- $||f||_E = \sup_{x \in E} |f(x)|$  where f is a function defined on  $E \subseteq \mathbb{R}^d$ .

#### CONTENTS

- $L^p = L^p(\mathbb{R}^d)$  if  $p \in ]0, \infty]$ .
- $l^p = l^p(\mathbb{N}_0)$  denotes the space of sequences  $(a_n)_{n \in \mathbb{N}_0}$  such that  $\|(a_n)_{n \in \mathbb{N}_0}\|_{l^p} = (\sum_{n=0}^{\infty} |a_n|^p)^{1/p} < \infty \ (p \in [0, \infty[).$
- $l^{\infty} = l^{\infty}(\mathbb{N}_0)$  denotes the space of sequences  $(a_n)_{n \in \mathbb{N}_0}$  such that  $||(a_n)_{n \in \mathbb{N}_0}||_{l^{\infty}} = \sup_{n \in \mathbb{N}_0} |a_n| < \infty$ .
- $D'(\mathbb{R}^d)$  denotes the space of distributions on  $\mathbb{R}^d$ .
- $\mathcal{S}(\mathbb{R}^d) = \mathcal{S}$  denotes the Schwartz space, composed of all rapidly decreasing infinitely differentiable functions on  $\mathbb{R}^d$ .
- $\mathcal{S}'(\mathbb{R}^d) = \mathcal{S}'$  denotes the topological dual of the space  $\mathcal{S}$ , i.e. the space of all tempered distributions on  $\mathbb{R}^d$ .
- $\mathcal{F}f$  denotes the Fourier transform of the distribution  $f \in \mathcal{S}'(\mathbb{R}^d)$ . If the function f belongs to  $L^1(\mathbb{R}^d)$ , this expression is equal to  $\mathcal{F}f(\xi) = (2\pi)^{-d/2} \int_{\mathbb{R}^d} e^{-ix\xi} f(x) dx$ .
- $\mathcal{F}^{-1}f$  denotes the inverse Fourier transform of the distribution  $f \in \mathcal{S}'(\mathbb{R}^d)$ . If the function f belongs to  $L^1(\mathbb{R}^d)$ , this expression is equal to  $\mathcal{F}^{-1}f(\xi) = (2\pi)^{-d/2} \int_{\mathbb{R}^d} e^{+ix\xi} f(x) dx$ .

### Introduction

The Hölder condition has been first introduced by Otto Ludwig Hölder, and is more than 130 years old. It is formally defined as follows: a real or complex-valued function f satisfies a Hölder condition if there exist C > 0 and  $\alpha \in ]0, 1[$  such that

$$|f(x) - f(y)| \le C|x - y|^{\alpha},$$

for all x, y belonging to the domain of f. If such an inequality is satisfied, one says that f belongs to the Hölder space of index  $\alpha$ . It is easy to define these spaces for  $\alpha \geq 1$ . They are used in many areas such as the theory of partial differential equation, harmonic analysis, stochastic differential equations and function spaces (see e.g. [4, 29, 48, 56, 118]). More recently, it gave birth to the multifractal analysis theory (see e.g. [13, 64, 67]).

The concept of Hölder-Zygmund space  $\Lambda^{\alpha}(\mathbb{R}^d)$  ( $\alpha > 0$ ) is directly linked with the Hölder condition and can be used to measure the global regularity through the parameter  $\alpha$ . Let  $W_{\alpha}$  be the Weierstraß function given by

$$W_{\alpha}(x) = \sum_{j=0}^{+\infty} 2^{-j\alpha} \cos(2^j \pi x),$$

where  $\alpha \in ]0, 1[$ . This function and the Brownian motion share a particular property: these functions are continuous but nowhere differentiable. A natural question that arises is how to determine which one of them is more regular. Using Hölder-Zygmund spaces, one can answer the question depending on the value of the parameter  $\alpha$ . One can prove that the Weierstraß function is more regular than the Brownian motion if and only if  $\alpha \geq 1/2$ .

Hölder-Zygmund spaces are particular cases of Besov spaces  $B_{p,q}^{\alpha}(\mathbb{R}^d)$  where  $0 < p, q \leq +\infty$  and  $\alpha > 0$  ([118]). Besov spaces were defined in 1959 by O. V. Besov in [18, 19]. A large number of references are now dedicated to them, see [20, 21, 117, 118, 119, 120, 121, 122]. These spaces were generalized in the middle of the seventies by several authors, with different starting points and in different contexts. They were first considered by the Russian school, and are still studied today in connection with embeddings, limiting embeddings, entropy numbers, probability theory and theory of stochastic processes (see e.g. [25, 46, 47, 85, 99]).

Meanwhile, S. Jaffard and Y. Meyer proposed a generalization of Hölder-Zygmund spaces in [65] for several purposes, such as Sobolev embeddings results, the study of regularity and multifractal purpose. This new type of Hölder-Zygmund spaces relies on the

#### CONTENTS

concept of modulus of continuity. Our approach is to propose a unified view. It consists in defining generalized Hölder-Zygmund spaces  $\Lambda^{\sigma,\alpha}(\mathbb{R}^d)$  as a particular case of generalized Besov spaces through the use of admissible sequences. This generalizes S. Jaffard's approach and helps to stand back from the different theories to obtain a better understanding of the underlying mechanisms behind the spaces. The generalized Hölder-Zygmund spaces  $\Lambda^{\sigma,\alpha}(\mathbb{R}^d)$  allow a more precise study of the regularity of a function, as illustrated by the following example. The Brownian motion is  $\alpha$ -regular with  $\alpha < \frac{1}{2}$ , in the sense that it belongs to the space  $\Lambda^{\alpha}(\mathbb{R}^d) = \Lambda^{\sigma,\alpha}(\mathbb{R}^d)$  with  $\sigma_j = 2^{-j\alpha}$ . Using a result of A. Khintchine ([72]), this sequence can be replaced by  $2^{-j/2}\sqrt{\log(j)}$ , giving a better characterization of the regularity of the Brownian motion. The concepts of Besov spaces and Hölder-Zygmund spaces, as well as their generalized versions, are recalled in chapter 1.

As generalized Hölder-Zygmund spaces  $\Lambda^{\sigma,\alpha}(\mathbb{R}^d)$  measure the regularity of functions, it is natural to wonder what links these spaces and classical concepts of regularity, like  $C^N(\mathbb{R}^d)$  spaces. For this purpose, the second chapter is dedicated to the study of the characterizations of the spaces  $\Lambda^{\sigma,\alpha}(\mathbb{R}^d)$ . It is shown that the properties of classical Hölder-Zygmund spaces can be transposed to their generalized version. In particular, elements in the generalized Hölder-Zygmund spaces can be characterized through approximation with polynomials, smooth functions obtained from a convolution product, a Taylor decomposition formula, wavelet coefficients and Littlewood-Paley decomposition. We also show that those spaces can be obtained through a (generalized) real interpolation of Sobolev spaces. The results of this chapter have been published in [76, 77] (except theorem 115).

An important concept associated to Hölder spaces is the Hölder exponent of some function f, which measures the global regularity of f. It is defined as follows: if  $f \in L^{\infty}(\mathbb{R}^d)$ , the Hölder exponent of f is given by

$$H_f = \sup\{\alpha > 0 : f \in \Lambda^{\alpha}(\mathbb{R}^d)\}.$$

In chapter 3, we explain how to transpose this concept to the generalized Hölder spaces, so we can define a concept of generalized Hölder exponents. As opposed to Hölder spaces which allow to measure the regularity of functions through the Hölder exponent, the uniform irregular spaces somehow measure the irregularity of a function ([31, 32, 33]). We show in the same chapter that those spaces can be expressed in terms of generalized Hölder-Zygmund spaces. Finally, we discuss some considerations related to financial models. Only results of section 3.1 have been published in [76, 77].

Up to now, we have considered the study of global regularity through the generalized Hölder-Zygmund spaces  $\Lambda^{\sigma,\alpha}(\mathbb{R}^d)$ . There also exists a pointwise version of the Hölder-Zygmund spaces, which gives some indication about the regularity at a given point  $x_0$ . These spaces are denoted by  $\Lambda^{\alpha}(x_0)$  and we can define a pointwise version of the Hölder exponent. It is formally defined as follows: if  $f \in L^{\infty}_{loc}(\mathbb{R}^d)$ , the Hölder exponent of f at  $x_0$ is given by

$$h_f(x_0) = \sup\{\alpha > 0 : f \in \Lambda^{\alpha}(x_0)\}.$$

#### CONTENTS

As Hölder-Zygmund spaces have been generalized through the use of admissible sequences, it is therefore natural to do the same for pointwise Hölder spaces. We introduce the generalized pointwise Hölder spaces in chapter 4 and we show that main properties of classical pointwise spaces are kept. In particular, we obtain a wavelet characterization of those spaces as well as a Taylor decomposition formula for their elements. Then, we study the pointwise and the global regularity of the Takagi function. We also show how to define a pointwise generalized Hölder exponent using these new spaces. Results of this chapter have been published in [78].

Several necessary theories and results are recalled in the appendix.

# Chapter 1

# Origins of the study of generalized Hölder-Zygmund spaces – State of the Art

#### **1.1** Background: reminder of basic concepts

The Besov spaces are the central point of the study of this thesis. These spaces allow one to measure the smoothness of functions, and are a useful tool to solve partial differential equations. They were first introduced in the 1960's and are defined here below.

Let us introduce some notations linking spaces  $l^q$  and  $L^p(\mathbb{R}^d)$ . If  $0 < p, q \leq \infty$  and  $(f_j)_{j \in \mathbb{N}_0}$  is a sequence of Lebesgue-measurable functions on  $\mathbb{R}^d$ , we set

$$\|(f_j)_{j \in \mathbb{N}_0} | l^q (L^p(\mathbb{R}^d)) \| = \left( \sum_{j=0}^{\infty} \|f_j\|_{L^p(\mathbb{R}^d)}^q \right)^{1/q} \quad \text{if } q \neq \infty$$

and

$$||(f_j)_{j\in\mathbb{N}_0}|l^{\infty}(L^p(\mathbb{R}^d))|| = \sup_{j\in\mathbb{N}_0} ||f_j||_{L^p(\mathbb{R}^d)}$$
 else.

To define Besov spaces, we need to introduce particular sequences of infinitely differentiable compactly supported functions which constitute a smooth resolution of unity.

**Definition 1.** Let  $\Phi(\mathbb{R}^d)$  denote the set of sequences  $(\varphi_j)_{j \in \mathbb{N}_0} \subset \mathcal{S}(\mathbb{R}^d)$  satisfying the following properties:

- 1.  $\operatorname{supp}\varphi_0 \subseteq \{\xi \in \mathbb{R}^d : |\xi| \le 2\};$
- 2. supp $\varphi_j \subseteq \{\xi \in \mathbb{R}^d : 2^{j-1} \le |\xi| \le 2^{j+1}\}, \quad j \in \mathbb{N}^*;$
- 3.  $\sup_{\xi \in \mathbb{R}^d} |D^{\alpha} \varphi_j(\xi)| \le c_{\alpha} 2^{-j|\alpha|}, \quad j \in \mathbb{N}_0, \alpha \in \mathbb{N}_0^d;$
- 4.  $\sum_{j=0}^{\infty} \varphi_j(\xi) = 1, \qquad \xi \in \mathbb{R}^d.$

H. Triebel introduced the definition of Besov spaces in [118] as:

**Definition 2** (Besov spaces). Let  $(\varphi_j)_{j \in \mathbb{N}_0} \in \Phi(\mathbb{R}^d)$  and  $0 < p, q \leq \infty, s \in \mathbb{R}$ . We define the *Besov space*  $B^s_{p,q}(\mathbb{R}^d)$  by

$$B_{p,q}^{s}(\mathbb{R}^{d}) = \{ f \in \mathcal{S}'(\mathbb{R}^{d}) : \|f\|_{B_{p,q}^{s}(\mathbb{R}^{d})} := \|(2^{js}\mathcal{F}^{-1}(\varphi_{j}\mathcal{F}f))_{j\in\mathbb{N}_{0}}|l^{q}(L^{p}(\mathbb{R}^{d}))\| < \infty \}$$

- **Remark 3.** 1. This definition does not depend on the chosen sequence  $(\varphi_j)_{j \in \mathbb{N}_0} \in \Phi(\mathbb{R}^d)$  (see e.g. [117, 118]).
  - 2. The distributions  $\mathcal{F}^{-1}(\varphi_j \mathcal{F} f)$   $(j \in \mathbb{N}_0)$  are associated to analytic functions defined on  $\mathbb{R}^d$  by the Paley-Wiener theorem. So, they belong to  $C^{\infty}(\mathbb{R}^d)$ .
  - 3. The spaces  $B_{p,q}^s(\mathbb{R}^d)$  are quasi-Banach spaces and Banach spaces if  $p, q \ge 1$  ([119]).

These spaces have many interesting properties, but they will not be discussed in further detail here. We refer to [43, 118, 119, 121] for a rather complete study.

Let us introduce Hölder spaces. We need to introduce the concept of finite difference. This concept is developed further in section 1.7 because it is one of the most important aspects of this thesis.

**Definition 4.** Let  $x, h \in \mathbb{R}^d$  and a function f defined on  $\mathbb{R}^d$ . We define the *(forward)* finite difference of order  $m \in \mathbb{N}^*$  of f by

$$\Delta_h^1 f(x) := f(x+h) - f(x)$$
  
$$\Delta_h^m f(x) := \Delta_h^1 \Delta_h^{m-1} f(x) \qquad (m \in \mathbb{N}^*).$$

If the function f is continuously differentiable in a neighbourhood of  $x \in \mathbb{R}^d$ , we have (by definition)

$$\lim_{\substack{h \to 0 \\ h \neq 0}} \frac{\Delta_{he_i}^1 f(x)}{h} = D_{x_i} f(x).$$

More generally, we have the following result ([73], lemma 4.4). This result can be easily proved using the definition of a derivative and permuting and reducing all the limits to a single one, thanks to the regularity assumption.

**Lemma 5.** If f is a n-times continuously differentiable function on a neighbourhood of  $x \in \mathbb{R}^d$ , then we have

$$\lim_{\substack{h \to 0\\h \in \mathbb{R}_0}} \frac{\Delta_{he_i}^n f(x)}{h^n} = D_{x_i}^n f(x)$$

The Hölder-Zygmund spaces can be interpreted as a continuum between the classical spaces  $C^n(\mathbb{R}^d)$  of *n*-times continuously differentiable functions on  $\mathbb{R}^d$ . In this way, they allow a more precise measurement of the global regularity of functions.

**Definition 6** (Hölder-Zygmund spaces). Let  $\alpha > 0$ . We define the Hölder-Zygmund space  $\Lambda^{\alpha}(\mathbb{R}^d)$  by

$$\Lambda^{\alpha}(\mathbb{R}^d) = \{ f \in L^{\infty}(\mathbb{R}^d) : \sup_{j \in \mathbb{N}_0} 2^{j\alpha} \sup_{|h| \le 2^{-j}} \|\Delta_h^{\lfloor \alpha \rfloor + 1} f\|_{L^{\infty}} < \infty \}.$$
(1.1)

The spaces  $\Lambda^n(\mathbb{R}^d)$  do not coincide with the spaces  $C^n(\mathbb{R}^d)$ , but each function belonging to  $\Lambda^{n+\varepsilon}(\mathbb{R}^d)$  with  $\varepsilon > 0$  is *n*-times continuously differentiable ([73]). These spaces have many interesting properties. For example, these spaces can be characterized by Taylor decomposition of their elements and can even be characterized by wavelet coefficients<sup>1</sup> ([93]).

- **Remark 7.** 1. Some authors ask the continuity of the elements of the Hölder spaces in the definition (see e.g. [73]). This is not necessary: indeed, we will see that functions of Hölder spaces can be modified on a set of null measure so that they become continuous (proposition 80).
  - 2. These spaces are Banach spaces with the norm defined by

$$\|f\|_{\Lambda^{\alpha}(\mathbb{R}^d)} = \|f\|_{L^{\infty}} + \sup_{j \in \mathbb{N}_0} \sup_{0 \neq |h| \leq 2^{-j}} \frac{\|\Delta_h^{\lfloor \alpha \rfloor + 1} f\|_{L^{\infty}}}{|h|^{\alpha}}$$

but they are not separable. They are indeed isomorphic to  $l^{\infty}$  ([118], p.87, Remark 1).

The Hölder spaces are closely linked to Besov spaces. Indeed, they are particular cases of Besov spaces as stated by the following result [118].

**Theorem 8.** Let  $\alpha > 0$ . We have

$$B^{\alpha}_{\infty,\infty}(\mathbb{R}^d) = \Lambda^{\alpha}(\mathbb{R}^d)$$

with equivalent norms.

To conclude this introduction to Besov and Hölder spaces, let us note that Besov spaces  $B_{p,q}^{\alpha}(\mathbb{R}^d)$  in general can be expressed in terms of finite differences. This fact is expressed by the next result ([117]).

**Theorem 9.** Let  $0 < p, q \leq +\infty$ ,  $\alpha > d(\frac{1}{\inf\{1,p\}} - 1)$  and  $M \in \mathbb{N}_0$  such that  $M > \alpha$ . We have

$$B_{p,q}^{\alpha}(\mathbb{R}^d) = \left\{ f \in \mathcal{S}'(\mathbb{R}^d) : \|f\|_{B_{p,q}^{\alpha}}^{(1),M} < +\infty \right\}$$
$$= \left\{ f \in \mathcal{S}'(\mathbb{R}^d) : \|f\|_{B_{p,q}^{\alpha}}^{(2),M} < +\infty \right\}$$

<sup>&</sup>lt;sup>1</sup>The concept of wavelet is developed further in section 5.4.

where

$$\|f\|_{B^{\alpha}_{p,q}}^{(1),M} = \|f\|_{L^{p}} + \left(\int_{\mathbb{R}^{d}} |h|^{-\alpha q} \sup_{|y| \le |h|} \|\Delta^{M}_{y} f\|_{p}^{q} \frac{dh}{|h|^{d}}\right)^{1/q}$$

and

$$\|f\|_{B^{\alpha}_{p,q}}^{(2),M} = \|f\|_{L^{p}} + \left(\int_{\mathbb{R}^{d}} |h|^{-\alpha q} \|\Delta_{h}^{M}f\|_{p}^{q} \frac{dh}{|h|^{d}}\right)^{1/q}$$

are equivalent quasi-norms to  $||f||_{B^{\alpha}_{p,q}}$  (with the usual modification if  $q = \infty$ ).

#### 1.2 An overview of admissible sequences

The core of generalized Besov and Hölder spaces relies on the notion of admissible sequence. We propose to study this concept separately before entering the definition of the generalized spaces.

**Definition 10.** A sequence  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$  of real positive numbers is called an *admissible* sequence if there exists two positive constants  $d_0$  and  $d_1$  such that

$$d_0\sigma_j \le \sigma_{j+1} \le d_1\sigma_j, \qquad j \in \mathbb{N}_0.$$
(1.2)

In the following, we will only consider admissible sequences which are not identically zero. This implies in particular that no element can be equal to 0.

For an admissible sequence  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$ , let

$$\underline{\sigma}_j := \inf_{k \ge 0} \frac{\sigma_{j+k}}{\sigma_k} \quad \text{and} \quad \overline{\sigma}_j := \sup_{k \ge 0} \frac{\sigma_{j+k}}{\sigma_k}, \quad j \in \mathbb{N}_0.$$

The lower and upper Boyd indices are respectively defined by

$$\underline{s}(\sigma) := \lim_{j \to +\infty} \frac{\log_2(\underline{\sigma}_j)}{j} \quad \text{and} \quad \overline{s}(\sigma) := \lim_{j \to +\infty} \frac{\log_2(\overline{\sigma}_j)}{j}.$$

Let us show that the two previous limits exist and are finite. For  $\overline{s}(\sigma)$ , this results from the fact that the sequence  $(\log_2(\overline{\sigma}_j))_{j \in \mathbb{N}_0}$  is subadditive and from the following lemma (see lemma 189 in the appendix for a proof of this lemma).

**Lemma 11** (Fekete (1923)). For every subadditive sequence  $(a_n)_{n \in \mathbb{N}^*}$ , the limit  $\lim_{n \to +\infty} \frac{a_n}{n}$  exists and is equal to  $\inf_{n \in \mathbb{N}^*} \frac{a_n}{n}$  (the limit can be equal to  $-\infty$ ).

Indeed, if  $j, l \in \mathbb{N}_0$  then

$$\frac{\sigma_{j+k+l}}{\sigma_k} = \frac{\sigma_{j+k+l}}{\sigma_{k+l}} \frac{\sigma_{k+l}}{\sigma_k} \qquad \forall k \in \mathbb{N}_0$$

which implies

$$\sup_{k \in \mathbb{N}_0} \frac{\sigma_{j+k+l}}{\sigma_k} \le \sup_{k \in \mathbb{N}_0} \frac{\sigma_{j+k+l}}{\sigma_{k+l}} \sup_{k \in \mathbb{N}_0} \frac{\sigma_{k+l}}{\sigma_k} \le \sup_{k \in \mathbb{N}_0} \frac{\sigma_{j+k}}{\sigma_k} \sup_{k \in \mathbb{N}_0} \frac{\sigma_{k+l}}{\sigma_k}$$

hence the conclusion. The same proof can be used to show that  $\underline{s}(\sigma)$  is well defined (we use the analogous result of lemma 11 for superadditive sequences).

The Boyd index  $\overline{s}(\sigma)$  of an admissible sequence  $\sigma$  describes the asymptotic behaviour of  $\overline{\sigma}_j$ ; similarly, the index  $\underline{s}(\sigma)$  describes the asymptotic behaviour of  $\underline{\sigma}_j$ . We notice that for  $\varepsilon > 0$ , there exist two positive constants  $c_1 = c_1(\varepsilon)$  and  $c_2 = c_2(\varepsilon)$  such that

$$c_1 2^{(\underline{s}(\sigma)-\varepsilon)j} \leq \underline{\sigma}_j \leq \frac{\sigma_{j+k}}{\sigma_k} \leq \overline{\sigma}_j \leq c_2 2^{(\overline{s}(\sigma)+\varepsilon)j}, \qquad j,k \in \mathbb{N}_0.$$

$$(1.3)$$

Conversely,  $\underline{s}(\sigma)$  and  $\overline{s}(\sigma)$  are respectively the biggest and the lowest real numbers satisfying inequalities (1.3) for every  $\varepsilon > 0$ .

**Remark 12.** Let us remark that the supremum of the constants  $d_0 > 0$  satisfying (1.2) also satisfies this inequality. It corresponds to  $\underline{\sigma}_1$ . Similarly, the infinum of constants  $d_1 > 0$  satisfying (1.2) corresponds to  $\overline{\sigma}_1$ . If there exists a constant  $d_0 > 1$  satisfying (1.2), then this implies that  $\sigma_j \to +\infty$  (if the sequence is not identically null). Similarly, if there exists a constant  $d_1 < 1$  satisfying (1.2), then  $\sigma_j \to 0$ . For Boyd indices: if  $\overline{s}(\sigma) < 0$ , then  $\sigma_j \to 0$ ; if  $\underline{s}(\sigma) > 0$ , then  $\sigma_j \to +\infty$ .

The following results are immediate.

**Lemma 13.** Let  $\sigma = (\sigma_i)_{i \in \mathbb{N}_0}$  and  $\gamma = (\gamma_i)_{i \in \mathbb{N}_0}$  be two admissible sequences.

- 1. The sequence  $\sigma + \gamma = (\sigma_i + \gamma_i)_{i \in \mathbb{N}_0}$  is admissible;
- 2. the sequence  $r\sigma = (r\sigma_i)_{i \in \mathbb{N}_0}$  is admissible for every positive real number r;
- 3. the sequence  $\sigma \gamma = (\sigma_i \gamma_i)_{i \in \mathbb{N}_0}$  is admissible;
- 4. the sequence  $\sigma^{-1} = (\sigma_i^{-1})_{i \in \mathbb{N}_0}$  is admissible.

**Lemma 14.** Let  $\alpha \in \mathbb{R}$  and  $\sigma$  be an admissible sequence. The new sequence  $\sigma^{\alpha}$  is admissible and

1. if  $\alpha > 0$ , then we have

$$\underline{s}(\sigma^{\alpha}) = \alpha \underline{s}(\sigma) \quad and \quad \overline{s}(\sigma^{\alpha}) = \alpha \overline{s}(\sigma);$$

2. if  $\alpha < 0$ , then we have

$$\underline{s}(\sigma^{\alpha}) = \alpha \overline{s}(\sigma) \quad and \quad \overline{s}(\sigma^{\alpha}) = \alpha \underline{s}(\sigma).$$

*Proof.* The proof is immediate: we start from inequalities (1.3) for the sequences  $\sigma$  and  $\sigma^{\alpha}$ , and we raise them to the power  $\alpha$  and  $1/\alpha$  respectively.

**Example 15.** The following example will be fundamental for the basis of this thesis. It is closely linked to classical Hölder spaces. Let  $\alpha \in \mathbb{R}$  and  $\sigma_j := 2^{j\alpha}$  for every  $j \in \mathbb{N}^*$ . Because the sequence  $\sigma$  satisfies

$$2^{\alpha}\sigma_j = \sigma_{j+1},$$

it is an admissible sequence and we have

$$\underline{\sigma}_j = 2^{j\alpha} = \overline{\sigma}_j$$

so that

$$\underline{s}(\sigma) = \overline{s}(\sigma) = \alpha.$$

**Example 16.** The sequence  $\sigma$  defined by  $\sigma_j = 2^{2^j}$   $(j \in \mathbb{N}^*)$  is not admissible. To show that, we proceed by reductio ad absurdum. As

$$\sigma_{j+1} = \sigma_j^2$$

the inequalities (1.2) should imply  $\sigma_j \leq C$  for some constant. This example gives an idea about how to interpret the definition of admissible sequence: inequalities (1.2) imply some limits on the speed of convergence and divergence.

The following examples come from [6] and [49].

**Example 17.** A function  $\Phi : [0, 1] \rightarrow ]0, +\infty[$  is a *slowly varying function* if it is (Lebesgue-) measurable and satisfies

$$\lim_{x \to 0} \frac{\Phi(\lambda x)}{\Phi(x)} = 1, \qquad \forall \lambda > 0.$$

For all  $s \in \mathbb{R}$ , the sequence  $\sigma = (2^{sj}\Phi(2^{-j}))_{j\in\mathbb{N}_0}$  is admissible and satisfies  $\underline{s}(\sigma) = \overline{s}(\sigma) = s$ . An example of slowly varying function is given by  $\Phi(x) = |\log(x)|$  (defined on some interval ]0, a[ with a < 1 and extended outside of this interval by a constant).

**Example 18.** Let  $(j_n)_{n \in \mathbb{N}_0}$  the increasing sequence defined by

$$j_0 = 0, \quad j_1 = 1, \quad j_{2n} = 2j_{2n-1} - j_{2n-2}, \quad j_{2n+1} = 2^{j_{2n}}, \quad n \in \mathbb{N}^*.$$

We set a sequence  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$  by

$$\sigma_j = \begin{cases} 2^{j_{2n}} & \text{if } j_{2n} \le j \le j_{2n+1} \\ 2^{j_{2n}} 4^{(j-j_{2n+1})} & \text{if } j_{2n+1} \le j < j_{2n+2}. \end{cases}$$

The sequence  $\sigma$  is admissible : by proceeding by case-to-case, we remark that

 $\sigma_j \le \sigma_{j+1} \le 4\sigma_j \qquad \forall j \in \mathbb{N}_0 \,.$ 

The sequence  $\sigma$  oscillates infinitely between  $(j)_{j\in\mathbb{N}_0}$  and  $(2^j)_{j\in\mathbb{N}_0}$ , i.e. we have

$$j \le \sigma_j \le 2^j \quad \forall j \in \mathbb{N}_0$$

and there exist infinitely many indices  $n \in \mathbb{N}_0$  and  $m \in \mathbb{N}_0$  such that  $\sigma_n = n$  and  $\sigma_m = 2^m$  respectively.

**Example 19.** Let  $s \in \mathbb{R}$  and  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$  be the sequence defined in the previous example. Then the new sequence

$$\gamma_j = 2^{js} \sigma_j$$

is admissible (lemma 13). Moreover, the sequence  $\gamma$  oscillates between  $(j2^{js})_{j\in\mathbb{N}_0}$  and  $(2^{j(s+1)})_{j\in\mathbb{N}_0}$ , i.e.

$$j2^{js} \le \gamma_j \le 2^{j(s+1)} \quad \forall j \in \mathbb{N}_0$$

where the left inequality is attained as well as the right one for infinitely many indices j.

The reader should note that in section 2.7 we provide a method to construct an admissible sequence such that the Boyd indices can take any given values (this method consists in adapting the previous example).

Before concluding this section, let us consider some classical terminologies linked to admissible sequences.

**Definition 20.** A sequence  $\sigma = (\sigma_j)_{j \in \mathbb{N}^*}$  of positive real numbers is called

1. almost increasing if there is a positive constant  $d_0$  such that

 $d_0\sigma_j \leq \sigma_k$  for all j,k such that  $0 \leq j \leq k$ .

2. strongly increasing if it is almost increasing and, in addition, there is a natural number  $k_0$  such that

 $2\sigma_j \leq \sigma_k$  for all j, k such that  $j + k_0 \leq k$ .

3. of bounded growth if there is a positive constant  $d_1$  such that

$$\sigma_{j+1} \leq d_1 \sigma_j$$
 for all  $j \in \mathbb{N}_0$ .

**Example 21.** The following examples illustrate these notions.

- 1. The sequence defined by  $\sigma_j = 2^{j\alpha}(1+j)^{\beta}$  (where  $\alpha > 0, \beta \in \mathbb{R}$ ) is strongly increasing and of bounded growth.
- 2. The sequence defined by  $\sigma_j = j!$  is strongly increasing, but not of bounded growth.
- 3. The sequence defined by  $\sigma_i = j$  is not strongly increasing but of bounded growth.

Let us note that each almost increasing and of bounded growth sequence is admissible, but the converse is false as it is shown by the admissible sequence  $(2^{-j\alpha})_{j\in\mathbb{N}_0}$  for  $\alpha > 0$ .

#### **1.3 Generalized Besov Spaces**

In this section, we will present some of the generalizations of Besov spaces that have been proposed in the literature for more than 30 years. One of the most general definitions until now is the one given by the definition 31, presented at the end of this section.

**Definition 22.** The set  $\mathcal{B}$  is defined by

$$\mathcal{B} = \left\{ \phi \in C(]0, \infty[) : \phi > 0, \quad \phi(1) = 1 \quad \text{and} \quad \sup_{s > 0} \frac{\phi(ts)}{\phi(s)} < \infty \quad \forall t > 0 \right\}.$$

**Definition 23.** Let *E* be a quasi-normed space,  $\phi \in \mathcal{B}$  and  $0 < q \leq \infty$ . We define

$$l^{q,\phi}(E) = \{ (a_j)_{j \in \mathbb{N}_0} \in E^{\mathbb{N}_0} : \| (\phi(2^j)a_j)_j \|_{l^q(E)} < \infty \}.$$

This is a quasi-normed space.

**Definition 24.** Let  $(\varphi_j)_{j \in \mathbb{N}_0} \in \Phi(\mathbb{R}^d)$  be a sequence of functions. For a chosen function  $\phi \in \mathcal{B}, 0 and <math>0 < q \leq \infty$ , we define

$$B_{p,q}^{\phi}(\mathbb{R}^{d}) = \{ f \in \mathcal{S}'(\mathbb{R}^{d}) : \|f|B_{p,q}^{\phi}(\mathbb{R}^{d})\| := \|(\mathcal{F}^{-1}(\varphi_{j}\mathcal{F}f))_{j\in\mathbb{N}_{0}}|l^{q,\phi}(L^{p}(\mathbb{R}^{d}))\| < \infty \}.$$

- **Remark 25.** 1. This definition does not depend on the chosen sequence  $(\varphi_j)_{j \in \mathbb{N}_0}$  (see e.g. [6]).
  - 2. Moreover, the distributions  $\mathcal{F}^{-1}(\varphi_j \mathcal{F} f)$   $(j \in \mathbb{N}_0)$  are associated with analytic functions defined on  $\mathbb{R}^d$  by the Paley-Wiener theorem. So, they belong to  $C^{\infty}(\mathbb{R}^d)$ .

Another (identical) definition is expressed in terms of admissible sequences. The idea is to replace the terms  $\varphi(2^j)$  by  $\sigma_j$ .

**Definition 26** (Generalized Besov spaces). Let  $(\varphi_j)_{j \in \mathbb{N}_0} \in \Phi(\mathbb{R}^d)$ ,  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$  be an admissible sequence and  $0 < p, q \leq \infty$ . The generalized Besov space  $B_{p,q}^{\sigma}$  is defined by

$$B_{p,q}^{\sigma}(\mathbb{R}^{d}) = \{ f \in \mathcal{S}'(\mathbb{R}^{d}) : \|f|B_{p,q}^{\sigma}(\mathbb{R}^{d})\| := \|(\sigma_{j}\mathcal{F}^{-1}(\varphi_{j}\mathcal{F}f)))_{j\in\mathbb{N}_{0}}|l^{q}(L^{p}(\mathbb{R}^{d}))\| < +\infty \}.$$

**Remark 27.** Let  $\alpha \in \mathbb{R}$ . By considering  $\phi(2^j) = 2^{\alpha j} = \sigma_j$   $(j \in \mathbb{N}^*)$ , we find the classical Besov spaces ([7]).

We note that the definition of the spaces  $B_{p,q}^{\sigma}$  is equivalent to the one of  $B_{p,q}^{\phi}$ . Indeed, the following result has been demonstrated in [6].

**Proposition 28.** Let  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$  be an admissible sequence and  $0 < p, q \leq \infty$ . There exists a function  $\phi_{\sigma} \in \mathcal{B}$  such that

$$B_{p,q}^{\phi_{\sigma}} = B_{p,q}^{\sigma}$$

and so that  $\phi_{\sigma}(2^j) = \sigma_j$  for all  $j \in \mathbb{N}^*$ .

#### GENERALIZED BESOV SPACES

*Proof.* Let  $\sigma$  be an admissible sequence. By considering the new sequence  $\sigma'$  defined by  $\sigma'_0 = 1$ ,  $\sigma'_n = \sigma_n \forall n \ge 1$  (which does not change anything on the associated space  $B^{\sigma}_{p,q}$ ), we can suppose that the first term of the sequence is 1. Then we construct a function  $\phi_{\sigma} \in \mathcal{B}$  in the following way:

$$\phi_{\sigma}(t) = \begin{cases} \frac{\sigma_{j+1} - \sigma_j}{2^j} (t - 2^j) + \sigma_j, & t \in [2^j, 2^{j+1}[, j \in \mathbb{N}_0]\\ 1, & t \in ]0, 1[. \end{cases}$$

So,  $\phi_{\sigma}(2^j) = \sigma_j$  for all  $j \in \mathbb{N}^*$ , which is sufficient to conclude.

The next result gives the converse part.

**Lemma 29.** Let  $\phi \in \mathcal{B}$  et  $0 < p, q \leq \infty$ . The sequence  $\sigma_{\phi}$  defined by  $\sigma_{\phi,j} = \phi(2^j)$   $(j \in \mathbb{N}_0)$  is admissible and such that

$$B_{p,q}^{\phi} = B_{p,q}^{\sigma_{\phi}}.$$

*Proof:* Indeed, we have for all  $j \in \mathbb{N}_0$ 

$$\frac{\phi(22^j)}{\phi(2^j)} \le \sup_{s>0} \frac{\phi(2s)}{\phi(s)} < +\infty$$

and

$$\frac{\phi(2^{-1}2^j)}{\phi(2^j)} \le \sup_{s>0} \frac{\phi(2^{-1}s)}{\phi(s)} < +\infty.$$

The definition given in [50, 97] generalizes those definitions. They allow a change of scale in the support of the functions  $\varphi_j$ , where the scale is associated with a strongly increasing sequence. Let  $N = (N_j)_{j \in \mathbb{N}_0}$  be a strongly increasing sequence, J a natural number different from zero and  $k_0$  the natural number associated to the strongly increasing sequence N. We define the associate covering  $\Omega^{N,J} = (\Omega_j^{N,J})_{j \in \mathbb{N}_0}$  of  $\mathbb{R}^d$  by

$$\Omega_j^{N,J} = \left\{ \xi \in \mathbb{R}^d : |\xi| \le N_{j+Jk_0} \right\} \quad \text{if} \quad j = 0, 1, ..., Jk_0 - 1,$$

and

$$\Omega_j^{N,J} = \left\{ \xi \in \mathbb{R}^d : N_{j-Jk_0} \le |\xi| \le N_{j+Jk_0} \right\} \quad \text{if} \quad j = Jk_0, Jk_0 + 1, \dots$$

As for classical Besov spaces, we define a smooth resolution of unity associated to the covering  $(\Omega_j^{N,J})_{j\in\mathbb{N}_0}$ .

**Definition 30.** We let  $\Phi^{N,J}(\mathbb{R}^d)$  denotes the set of sequences  $(\varphi_j^{N,J})_{j\in\mathbb{N}_0} \subset \mathcal{S}(\mathbb{R}^d)$  satisfying the following properties:

- 1.  $\varphi_i^{N,J}(\xi) \ge 0$  for all  $\xi \in \mathbb{R}^d, j \in \mathbb{N}_0;$
- 2.  $\operatorname{supp}\varphi_j^{N,J} \subseteq \Omega_j^{N,J}$  for all  $j \in \mathbb{N}_0$ ;

#### A CENTRAL RESULT

3. for all  $\alpha \in \mathbb{N}_0^d$ , there exists a constant  $c_\alpha > 0$  such that for all  $j \in \mathbb{N}_0$  we have  $|D^{\alpha}\varphi_j^{N,J}(\xi)| \leq c_\alpha (1+|\xi|^2)^{-|\alpha|/2}$  for all  $\xi \in \mathbb{R}^d$ ;

4. there exists a constant  $c_{\varphi} > 0$  such that

$$0 < \sum_{j=0}^{\infty} \varphi_j^{N,J}(\xi) = c_{\varphi} < \infty \quad \text{for all} \quad \xi \in \mathbb{R}^d.$$

The construction (and so the existence) of such a sequence of functions is tricky. We refer to [50] for such constructions.

**Definition 31.** Let  $N = (N_j)_{j \in \mathbb{N}_0}$  be a strongly increasing sequence and of bounded growth. Let  $J \in \mathbb{N}^*$ ,  $(\varphi_j^{N,J})_{j \in \mathbb{N}_0} \in \Phi^{N,J}(\mathbb{R}^d)$  and  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$  be an admissible sequence. Let  $0 < p, q \leq \infty$ . The generalized Besov space  $B_{p,q}^{\sigma,N}$  is defined by

$$B_{p,q}^{\sigma,N}(\mathbb{R}^d) = \{ f \in \mathcal{S}'(\mathbb{R}^d) : \|f|B_{p,q}^{\sigma,N}(\mathbb{R}^d)\| := \|(\sigma_j \mathcal{F}^{-1}(\varphi_j^{N,J} \mathcal{F}f)))_{j \in \mathbb{N}_0} |l^q(L^p(\mathbb{R}^d))\| < +\infty \}.$$

As for classical Besov spaces, the distributions  $\mathcal{F}^{-1}(\varphi_j^{N,J}\mathcal{F}f)$  are associated with functions belonging to  $C^{\infty}(\mathbb{R}^d)$ . The assumption on N implies in particular that N is an admissible sequence. One can easily adapt the usual proofs for the classical properties of Besov spaces in [118] ([50]), to prove the independence of the definition from the choice of the sequence  $(\varphi_j^{N,J})_{j\in\mathbb{N}_0}$ , the inclusion of the Schwartz space into those spaces and so on.

**Remark 32.** If we consider the sequence  $N_j = 2^j$   $(j \in \mathbb{N}_0)$ , definition 31 is equivalent to definition 26 so that we have  $B_{p,q}^{\sigma,(2^j)_j} = B_{p,q}^{\sigma}$ .

The generalized Besov spaces have been studied since the mid of 1970's by many authors using many different approaches. Several references about these ones can be found in [50] with many historical explanations about them.

#### 1.4 A central result

In this section, we introduce the generalized Hölder-Zygmund spaces that are studied in this thesis. We explain their origins and some first reasons why these have interesting properties.

The following result is the core of our work. It has been proved in [97].

**Theorem 33.** Let  $0 < p, q \leq \infty$ ,  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$  and  $N = (N_j)_{j \in \mathbb{N}_0}$  be two admissible sequences such that  $\underline{N}_1 > 1$  and  $\underline{s}(\sigma^{-1})\overline{s}(N)^{-1} > n(1/p-1)_+$ . Let  $M \in \mathbb{N}_0$  such that  $M > \overline{s}(\sigma^{-1})\underline{s}(N)^{-1}$ . We have  $B_{p,q}^{\sigma^{-1},N} =$ 

$$\left\{ f \in L^{\max(1,p)} : \|f\| B_{p,q}^{\sigma^{-1},N} \|^{(M)} := \|f\|_{L^p} + \left( \sum_{j=0}^{\infty} \sigma_j^{-q} \left( \sup_{|h| \le N_j^{-1}} \|\Delta_h^M f\|_{L^p} \right)^q \right)^{1/q} < \infty \right\}$$

(with an obvious modification if  $q = \infty$ ), in the sense of equivalent quasi-norms.

#### A CENTRAL RESULT

If we consider the particular case  $N_j = 2^j$  and  $p = q = +\infty$ , then  $N_1 = 2$  and  $\overline{s}(N) = \underline{s}(N) = 1$ , so that the previous result can be restated in this case as follows.

**Corollary 34.** Let  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$  be an admissible sequence such that  $\underline{s}(\sigma^{-1}) > 0$  and let  $M \in \mathbb{N}_0$  such that  $M > \overline{s}(\sigma^{-1})$ . We have

$$B_{\infty,\infty}^{\sigma^{-1}} = \left\{ f \in L^{\infty} : \|f\|_{B_{\infty,\infty}^{\sigma}}\|^{(M)} := \|f\|_{L^{\infty}} + \sup_{j \in \mathbb{N}_{0}} \sigma_{j}^{-1} \sup_{|h| \le 2^{-j}} \|\Delta_{h}^{M} f\|_{L^{\infty}} < \infty \right\}$$

in the sense of equivalent quasi-norms.

Let us remark the immediate analogy with the classical Hölder-Zygmund spaces: it consists of replacing the sequence  $2^{-j\alpha}$  of definition 6 by an arbitrary admissible sequence  $\sigma$  which satisfies the conditions of corollary 34. This means replacing the following control of the finite difference

$$\sup_{|h| \le 2^{-j}} \|\Delta_h^M f\|_{L^\infty} \le C 2^{-j\alpha}$$

by the more general control

$$\sup_{|h| \le 2^{-j}} \|\Delta_h^M f\|_{L^{\infty}} \le C\sigma_j.$$

Theorem 33 shows that an even more generalized control is possible: it consists of replacing the ball of dyadic radius on which we consider the finite difference  $(h \in B(0, 2^{-j}))$  by a ball of a more general radius linked to an arbitrary admissible sequence  $(h \in B(0, N_j^{-1}))$ . The control on the finite difference under that last generalization is then

$$\sup_{|h| \le N_j^{-1}} \|\Delta_h^M f\|_{L^{\infty}} \le C\sigma_j.$$

The next step is to study the conditions needed for the sequence  $\sigma$  to be able to apply the corollary 34. We provide some results that can help the reader to interpret these conditions and that are often used in the sequel.

**Lemma 35.** Let  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$  be an admissible sequence. If  $\underline{s}(\sigma^{-1}) > 0$ , then there exists C > 0 such that

$$\sum_{j=J}^{+\infty} \sigma_j \le C\sigma_J \quad \forall J \in \mathbb{N}_0 \,.$$

*Proof.* Let  $\varepsilon > 0$  such that  $\underline{s}(\sigma^{-1}) - \varepsilon > 0$ . Using inequalities (1.3), we obtain

$$\sum_{j=J}^{+\infty} \sigma_j = \sum_{j=0}^{+\infty} \sigma_{J+j} \le C \sum_{j=0}^{+\infty} \sigma_J 2^{-(\underline{s}(\sigma^{-1})-\varepsilon)j} \le C\sigma_J$$

for all  $J \in \mathbb{N}_0$ .

	_	_	

#### A CENTRAL RESULT

So the condition  $\underline{s}(\sigma^{-1}) > 0$  does not only ask for the sequence  $\sigma$  to converge to 0, but also asks the convergence to be "as quick" as some arbitrary power of 2. This is confirmed by inequalities (1.3) which implies

$$\sigma_i < C2^{-(\underline{s}(\sigma^{-1})-\varepsilon)j}$$

(where the constant C depends on  $\varepsilon$ ) where  $\varepsilon$  can be taken as arbitrarily small. Moreover, the Boyd index gives the power of 2 that should be used. One can easily prove the following result.

**Lemma 36.** Let  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$  be an admissible sequence. It satisfies  $\underline{s}(\sigma) > 0$  if and only if there exist  $\varepsilon > 0$  and C > 0 such that  $C2^{\varepsilon j} \leq \underline{\sigma}_j$  for all  $j \in \mathbb{N}^*$ .

The next result allows for the interpretation of the condition  $M > \overline{s}(\sigma^{-1})$  in the corollary 34.

**Lemma 37.** Let  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$  be an admissible sequence. If  $M \in \mathbb{N}_0$  satisfies  $M > \overline{s}(\sigma^{-1})$ , then there exists C > 0 such that

$$\sum_{j=0}^{J} 2^{jM} \sigma_j \le C 2^{JM} \sigma_J \quad \forall J \in \mathbb{N}_0 \,.$$

*Proof.* Let  $\varepsilon > 0$  such that  $\overline{s}(\sigma^{-1}) + \varepsilon < M - \varepsilon$ . Using inequalities (1.3), we obtain

$$\frac{\sigma_k}{\sigma_{j+k}} \le C 2^{jM} 2^{-\varepsilon j} \quad \forall j,k \in \mathbb{N}_0$$

which implies for  $J \in \mathbb{N}_0$ 

$$2^{-JM}\sigma_J^{-1}\sum_{j=0}^J 2^{jM}\sigma_j \le C\sum_{j=0}^J 2^{-\varepsilon(J-j)} \le C$$

hence the conclusion.

So the condition  $M > \overline{s}(\sigma^{-1})$  means that the growth of the sequence  $2^{jM}\sigma_j$  must be fast enough so that the term  $2^{JM}\sigma_J$  is still "comparable" to the sum of the previous elements of the sequence  $(\{2^M\sigma_1, 2^{2M}\sigma_2..., 2^{JM}\sigma_J\})$ . The natural M we have to choose to apply corollary 34 must be large enough so that  $2^{jM}\sigma_j$  grows sufficiently fast. We can also note that inequalities (1.3) imply in particular that  $C \leq 2^{jM}\sigma_j$  for all  $j \in \mathbb{N}_0$ .

that inequalities (1.3) imply in particular that  $C \leq 2^{jM} \sigma_j$  for all  $j \in \mathbb{N}_0$ . Another important characterization of spaces  $B_{p,q}^{\sigma,N}$  is proved in [97] (Theorem 3.1). This characterization states that any element of  $B_{p,q}^{\sigma,N}$  can be approximated by functions with compactly supported Fourier transforms. Let  $0 and N be an admissible sequence such that <math>\underline{N}_1 > 1$ . We easily show that N is a strongly increasing sequence, and so there exists a natural number  $k_0 \in \mathbb{N}_0$  such that

 $N_k \ge 2N_j \quad \forall k, j \quad \text{satisfying} \quad k \ge j + k_0.$ 

We set

$$\mathcal{U}_p^N = \{a = (a_j)_{j \in \mathbb{N}_0} : a_j \in \mathcal{S}' \cap L^p, \operatorname{supp} \mathcal{F} a_j \subset \{y : |y| \le N_{j+k_0}\}, j \in \mathbb{N}_0\}.$$

**Theorem 38.** Let  $0 < p, q \leq \infty$ ,  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$  and  $N = (N_j)_{j \in \mathbb{N}_0}$  be two admissible sequences such that  $\underline{N}_1 > 1$  and  $\underline{s}(\sigma)\overline{s}(N)^{-1} > n(1/p-1)_+$ . We have

$$B_{p,q}^{\sigma,N} = \left\{ f \in \mathcal{S}' : \exists a = (a_j)_j \in \mathcal{U}_p^N \text{ such that } f = \lim_{k \to +\infty} a_k \text{ in } \mathcal{S}' \text{ and} \\ \|f|B_{p,q}^{\sigma,N}\|^a := \|a_0\|_{L^p} + \|(\sigma_k(f-a_k))_{k \in \mathbb{N}^*}|l^q(L^p)\| < \infty \right\}.$$

Moreover,

$$||f|B_{p,q}^{\sigma,N}||^X := \inf ||f|B_{p,q}^{\sigma,N}||^a$$

(where the infimum is taken over all admissible systems  $a \in \mathcal{U}_p^N$ ), is an equivalent quasinorm in  $B_{p,q}^{\sigma,N}$ .

If we consider  $N_j = 2^j$  and  $p = q = +\infty$ , this result can be restated in this case as follows:

**Corollary 39.** Let  $\sigma = (\sigma_i)_{i \in \mathbb{N}_0}$  be an admissible sequence such that  $\underline{s}(\sigma) > 0$ . Then

$$B_{\infty,\infty}^{\sigma} = \left\{ f \in \mathcal{S}' : \exists a = (a_j)_j \in \mathcal{U}_{\infty} \text{ such that } f = \lim_{k \to +\infty} a_k \text{ in } \mathcal{S}' \text{ and} \\ \|f|B_{\infty,\infty}^{\sigma}\|^a := \|a_0\|_{L^{\infty}} + \sup_{k \in \mathbb{N}^*} \sigma_k \|f - a_k\|_{L^{\infty}} < \infty \right\}$$

where

$$\mathcal{U}_{\infty} = \left\{ a = (a_j)_{j \in \mathbb{N}_0} : a_j \in \mathcal{S}' \cap L^{\infty}, supp\mathcal{F}a_j \subset \{y : |y| \le 2^{j+1}\}, j \in \mathbb{N}_0 \right\}.$$

#### 1.5 Definition of generalized Hölder-Zygmund spaces

In this section, we introduce the definition of generalized Hölder-Zygmund spaces studied in this thesis. We show also that these spaces generalize a particular type of space studied from the wavelet<sup>2</sup> point of view some years ago by Stéphane Jaffard, Yves Meyer and Marianne Clausel ([65, 30]). Indeed, these authors already noticed that the useful wavelet characterization known for classical Hölder spaces could be generalized to a more broad scope which can be used in practice. It is remarkable that the path they follow to generalize Hölder spaces and the completely different path followed through the generalization of Besov point of view (by Hans-Gerd Leopold, Walter Farkas, Susana Moura, Alexandre Almeida, ...) merge into one unique notion several years after their work, when restricted to the Hölder case.

Let us give a definition of the generalized Hölder-Zygmund spaces. This definition relies on the principles exposed in the previous section.

<sup>&</sup>lt;sup>2</sup>Wavelets are a mathematical tool used to study signals in mathematical analysis, in engineering and physical domains (see [89] for a good overview on the subject). This concept is introduced in section 5.4.

**Definition 40.** Let  $\alpha > 0$ ,  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$  and  $N = (N_j)_{j \in \mathbb{N}_0}$  be two admissible sequences. The generalized Hölder-Zygmund space  $\Lambda_{\sigma,N}^{\alpha}(\mathbb{R}^d)$  is defined by

$$\Lambda^{\alpha}_{\sigma,N}(\mathbb{R}^d) = \{ f \in L^{\infty}(\mathbb{R}^d) : \sup_{j \in \mathbb{N}_0} \sigma_j^{-1} \sup_{|h| \le N_j^{-1}} \|\Delta_h^{\lfloor \alpha \rfloor + 1} f\|_{L^{\infty}} < \infty \}.$$

If  $N_j = 2^j$   $(j \in \mathbb{N}_0)$ , then we note  $\Lambda^{\sigma,\alpha}(\mathbb{R}^d)$  instead of  $\Lambda^{\alpha}_{\sigma,N}(\mathbb{R}^d)$  to simplify notations.

Remark 41. The application

$$f \mapsto |f|_{\Lambda_{\sigma,N}^{\alpha}} = \sup_{j \in \mathbb{N}_0} \sigma_j^{-1} \sup_{|h| \le N_j^{-1}} \|\Delta_h^{\lfloor \alpha \rfloor + 1} f\|_{L^{\infty}}$$

is a semi-norm on the space  $\Lambda^{\alpha}_{\sigma,N}(\mathbb{R}^d)$ . The application

$$f \mapsto \|f\|_{\Lambda^{\alpha}_{\sigma,N}} = \|f\|_{L^{\infty}} + |f|_{\Lambda^{\alpha}_{\sigma,N}}$$

is trivially a norm on this space. Moreover, the space  $(\Lambda_{\sigma,N}^{\alpha}, \|.\|_{\Lambda_{\sigma,N}^{\alpha}})$  is complete. Let  $(f_j)_{j \in \mathbb{N}_0}$  be a Cauchy sequence on this space. Because the space  $L^{\infty}$  is complete, the sequence  $f_j$  converges in  $L^{\infty}$  to a function f. We can choose a subsequence  $(f_{k(j)})_{j \in \mathbb{N}_0}$  of the Cauchy sequence such that

$$|f_{k(j+1)} - f_{k(j)}|_{\Lambda^{\alpha}_{\sigma N}} < 2^{-j} \qquad \forall j \in \mathbb{N}_0$$

We set  $g_1 = f_{k(1)}$  and  $g_{j+1} = f_{k(j+1)} - f_{k(j)}$   $(j \in \mathbb{N}^*)$ , so that we obtain  $f = \sum_{j=1}^{+\infty} g_j$ . We find

$$|f - f_{k(j)}|_{\Lambda^{\alpha}_{\sigma,N}} \le \sum_{l=j+1}^{+\infty} |g_l|_{\Lambda^{\alpha}_{\sigma,N}} \le \sum_{l=j+1}^{+\infty} 2^{-l+1}.$$

As a consequence,  $||f - f_{k(j)}||_{\Lambda^{\alpha}_{\sigma,N}} \to 0$  if  $j \to +\infty$ . Proposition 4.4.4 of [106] leads to the following result:

**Proposition 42.** Let  $\alpha > 0$ ,  $\sigma$  and N be two admissible sequences. The space  $(\Lambda_{\sigma,N}^{\alpha}, \|.\|_{\Lambda_{\sigma,N}^{\alpha}})$  is a Banach space.

**Remark 43.** The choice of the order of the finite difference should take into account the speed of convergence of the sequence  $\sigma$ , so that the spaces are not reduced to the space of constant functions. Indeed, we prove in section 2.3 that if  $m+1 < \underline{s}(\sigma^{-1})$  (where  $m \in \mathbb{N}_0$ ), then the space  $\Lambda^{\sigma,m}$  is composed of constant functions.

As already mentioned, another generalized version of the classical Hölder-Zygmund spaces has been introduced in [65] et [30], which relies on the concept of moduli of continuity. Let us introduce that concept.

**Definition 44** (Modulus of continuity). A non-decreasing function non identically null  $\omega$  defined on  $[0, +\infty]$  is a *modulus of continuity* if it satisfies the two following conditions:

1.  $\omega(0) = 0$ 

2. there exists a constant  $C_{\omega} > 0$  such that

$$\omega(2t) \le C_{\omega}\omega(t), \quad \forall t \in [0, +\infty[. \tag{1.4})$$

Obviously, such a function cannot be equal to 0 at any point distinct of 0. The generalized version of Hölder-Zygmund space linked to this concept is the following.

**Definition 45.** Let  $\alpha > 0$  and  $\omega$  be a modulus of continuity. We say that a function  $f \in L^{\infty}(\mathbb{R}^d)$  belongs to the generalized Hölder-Zygmund space  $\Lambda^{\omega,\alpha}(\mathbb{R}^d)$  if there exists a constant C > 0 such that

$$\sup_{h|\leq 2^{-j}} \|\Delta_h^{\lfloor \alpha \rfloor + 1} f\|_{L^{\infty}} \leq C\omega(2^{-j}) \qquad \forall j \in \mathbb{N}_0.$$

Since only the behavior of  $\omega$  near the origin really matters, we could define a modulus of continuity using germ functions. The purpose of the remaining part of this section is to study the link between these spaces and the generalized Hölder-Zygmund spaces based on admissible sequences.

Let  $\omega$  be a modulus of continuity. We define  $\sigma_j := \omega(2^{-j})$  for all  $j \in \mathbb{N}_0$ . The sequence  $\sigma$  so defined is admissible and shows that the generalized Hölder-Zygmund spaces linked to moduli of continuity are a particular case of the ones concerned by the definition 40. The reader might wonder what the exact link is between moduli of continuity and admissible sequences. The two following immediate results give an answer to that question.

**Lemma 46.** Let  $(\sigma_j)_{j \in \mathbb{N}_0}$  be an admissible sequence. There exists a modulus of continuity  $\omega$  such that  $\sigma_j = \omega(2^{-j}) \ \forall j \in \mathbb{N}_0$  if and only if  $(\sigma_j)_{j \in \mathbb{N}_0}$  is a non-increasing sequence.

**Lemma 47.** Let  $(\sigma_j)_{j \in \mathbb{N}_0}$  be an admissible sequence. There exists a modulus of continuity  $\omega$  continuous at 0 and such that  $\sigma_j = \omega(2^{-j}) \ \forall j \in \mathbb{N}_0$  if and only if  $(\sigma_j)_{j \in \mathbb{N}_0}$  is a non-increasing sequence that converges to 0.

*Proof:* We define the modulus of continuity by  $\omega(r) := \sigma_j$  if  $r \in [2^{-j}, 2^{-(j-1)}]$   $(j \in \mathbb{N}_0)$ , and we extend it by a constant over the interval  $[1, +\infty[$ .

Our considerations of the section 1.4 can now be restated as follow.

**Proposition 48.** 1. Let  $\sigma$  and N be two admissible sequences such that  $\underline{N}_1 > 1$  and  $\underline{s}(\sigma^{-1})\overline{s}(N)^{-1} > 0$ . We have

$$B^{\sigma^{-1},N}_{\infty,\infty}(\mathbb{R}^d) = \Lambda^{\overline{s}(\sigma^{-1})\underline{s}(N)^{-1}}_{\sigma,N}(\mathbb{R}^d) = \Lambda^{M-1}_{\sigma,N}(\mathbb{R}^d)$$

for all  $M \in \mathbb{N}_0$  such that  $M > \overline{s}(\sigma^{-1})\underline{s}(N)^{-1}$ .

2. Let  $\sigma$  be an admissible sequence such that  $\underline{s}(\sigma^{-1}) > 0$ . We have

$$B^{\sigma^{-1}}_{\infty,\infty}(\mathbb{R}^d) = \Lambda^{\sigma,\overline{s}(\sigma^{-1})}(\mathbb{R}^d) = \Lambda^{\sigma,M-1}(\mathbb{R}^d)$$

for all  $M \in \mathbb{N}_0$  such that  $M > \overline{s}(\sigma^{-1})$ .

3. Let  $\omega$  be a modulus of continuity. We set  $\sigma = (\omega(2^{-j}))_{j \in \mathbb{N}^*}$ . If  $\underline{s}(\sigma^{-1}) > 0$ , then

$$B^{\sigma^{-1}}_{\infty,\infty}(\mathbb{R}^d) = \Lambda^{\omega,\overline{s}(\sigma^{-1})}(\mathbb{R}^d).$$

A modulus of continuity satisfies the conditions of this result if, for example, there exist C > 1 and  $J \in \mathbb{N}^*$  such that

$$\omega(2^{-j}) \ge C\omega(2^{-(j+1)}) \qquad \forall j \ge J.$$

This condition is often realized for interesting moduli of continuity or admissible sequences (i.e. the ones that give interesting properties to the associated spaces). For example, this is the case for the sequence  $\sigma_j = \omega(2^{-j}) := 2^{-j\alpha}$  ( $\alpha > 0$ ) which corresponds to the classical Hölder-Zygmund spaces. To end this section, we show some examples of admissible sequences which are moduli of continuity for which the proposition 48 can or cannot be applied.

**Example 49.** Let 0 < a < 1. We define a modulus of continuity by

$$\omega(r) = \begin{cases} |\log_2(r)|^{-1} & \text{si } r \in ]0, a[\\ |\log_2(a)|^{-1} & \text{si } r \ge a. \end{cases}$$

The sequence  $\sigma = (\omega(2^{-j})^{-1})_{j \in \mathbb{N}_0}$  is such that  $\underline{s}(\sigma) = 0$ .

**Example 50.** We look back at the example 18. The sequence  $\sigma$  is non-decreasing so  $\sigma^{-1}$  defines a modulus of continuity by lemma 46. For all  $j \in \mathbb{N}_0$ , we have

$$\underline{\sigma}_j = \inf_{k \ge 0} \frac{\sigma_{j+k}}{\sigma_k} \ge 1$$

and by considering  $n \in \mathbb{N}_0$  sufficiently large so that  $j_{2n} + j \leq j_{2n+1}$ , we obtain  $\sigma_{j+j_{2n}} = \sigma_{j_{2n}} = 2^{j_{2n}}$ . So

 $\underline{\sigma}_i = 1$ 

which implies  $\underline{s}(\sigma) = 0$ .

**Example 51.** We look back at the example 19 with s > 0. The sequence  $\gamma$  is nondecreasing and  $\gamma^{-1}$  defines a modulus of continuity. Moreover, by lemma 52 which is proved below, we have  $\underline{s}(\gamma) \geq s > 0$  and corollary 48 can be applied.

**Lemma 52.** Let  $\sigma$  and  $\gamma$  be two admissible sequences. We have

$$\underline{s}(\sigma\gamma) \ge \underline{s}(\sigma) + \underline{s}(\gamma)$$

and

$$\overline{s}(\sigma\gamma) \le \overline{s}(\sigma) + \overline{s}(\gamma)$$

*Proof.* For every  $j \in \mathbb{N}_0$ , we have

$$\underline{\sigma\gamma}_{j} = \inf_{k \ge 0} \frac{\sigma_{j+k}\gamma_{j+k}}{\sigma_{k}\gamma_{k}} \ge \underline{\sigma}_{j}\underline{\gamma}_{j}$$

which implies

$$\frac{\log_2(\underline{\sigma}\underline{\gamma}_j)}{j} \ge \frac{\log_2(\underline{\sigma}_j)}{j} + \frac{\log_2(\underline{\gamma}_j)}{j}.$$

This proves the first inequality. The second one can be proved in the same way.

Before ending this section, let us provide generalizations of lemmata 37 and 35 (these results do not appear in [76, 77]).

**Lemma 53.** Let  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$  and  $N = (N_j)_{j \in \mathbb{N}_0}$  be two admissible sequences such that  $\underline{s}(N) > 0$ . If  $M \in \mathbb{N}_0$  is such that  $M > \overline{s}(\sigma^{-1})\underline{s}(N)^{-1}$ , then there exists a constant C > 0 such that

$$\sum_{j=0}^{J} N_j^M \sigma_j \le C N_J^M \sigma_J \quad \forall J \in \mathbb{N}_0 \,.$$

*Proof.* Let  $\varepsilon > 0$  such that  $\overline{s}(\sigma^{-1}) + \varepsilon < M(\underline{s}(N) - \varepsilon) - \varepsilon$ . Using inequalities (1.3) we obtain

$$\frac{\sigma_k}{\sigma_{j+k}} \le C \left(\frac{N_{j+k}}{N_k}\right)^M 2^{-\varepsilon j} \quad \forall j, k \in \mathbb{N}_0$$

which implies, for  $J \in \mathbb{N}_0$ ,

$$N_J^{-M} \sigma_J^{-1} \sum_{j=0}^J N_j^M \sigma_j \le C \sum_{j=0}^J 2^{-\varepsilon(J-j)} \le C$$

which allows to conclude.

One can proceed similarly to prove the following result.

**Lemma 54.** Let  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$  and  $N = (N_j)_{j \in \mathbb{N}_0}$  be two admissible sequences such that  $\overline{s}(N) > 0$ . If  $L \in \mathbb{N}_0$  is such that  $L < \underline{s}(\sigma^{-1})\overline{s}(N)^{-1}$ , then there exists a constant C > 0 such that

$$\sum_{j=J}^{+\infty} N_j^L \sigma_j \le C N_J^L \sigma_J \quad \forall J \in \mathbb{N}_0 \,.$$

#### 1.6 Some last remarks about generalized Besov and Hölder-Zygmund spaces

The purpose of this section is to discuss some last points about generalized Besov spaces. After this section, we will only be interested in generalized Hölder-Zygmund spaces. The first remark states that some results and references connect classical and generalized Besov spaces with wavelets. These comments are interesting as we will consider characterizations of generalized Hölder spaces in terms of wavelet coefficients in section 2.9. Secondly, as stated in the proposition 48, generalized Besov spaces coincide with generalized Hölder spaces. It is thus natural to ask whether it is the case in general. The second remark gives a negative answer to that question. Finally, the last remark somehow connects the generalized Besov spaces with the classical Hölder-Zygmund spaces, and consequently with the usual notion of continuity.

**Remark 55.** A characterization of classical Besov spaces  $B_{p,q}^s$   $(s \in \mathbb{R}, 0 < p, q \leq +\infty)$ in terms of wavelet decomposition has been proved in [120] (see also [35, 40, 41]). This result has been generalized to Weighted Besov spaces in [61], and recently to spaces  $B_{p,q}^{\sigma}$ (0 in [6]. The existence of a topological isomorphism betweenthe generalized Besov spaces and sequences spaces was also proved in the last reference.Other characterizations of these spaces were proved in [27]. In section 2.9, we show a simple $characterization of space <math>B_{\infty,\infty}^{\sigma}$  in terms of wavelet coefficients under some assumptions on the admissible sequence  $\sigma$ .

**Remark 56.** The goal of this section is to show that if  $\sigma$  is an admissible sequence with  $\underline{s}(\sigma) = 0$ , then the spaces  $B_{\infty,\infty}^{\sigma}$  can be different from every generalized Hölder-Zygmund space  $\Lambda^{\psi,\alpha}$  (for all arbitrary admissible sequence  $\psi$ ). By considering the sequence  $\sigma_j^{-1} = \frac{1}{j}$  which corresponds to the admissible sequence given by example 49, we know that  $\underline{s}(\sigma) = 0$  and proposition 3.13 of [28] implies that

$$B^{\sigma}_{\infty,\infty} \not\hookrightarrow L^{\infty},$$

where the symbol " $\hookrightarrow$ " means that the corresponding embedding is continuous. In particular, the space  $B^{\sigma}_{\infty,\infty}$  can not be written as a generalized Hölder-Zygmund space.

This fact shows that the condition  $\underline{s}(\sigma) > 0$  of theorem 33 is necessary and that the concept of generalized Besov space can not always be directly linked to some generalized Hölder-Zygmund spaces.

**Remark 57.** The following result was proved in [6] (proposition 4).

**Proposition 58.** Let  $\phi$ ,  $\psi \in \mathcal{B}$ ,  $0 , <math>0 < q_0, q_1 \leq \infty$ . If  $(\frac{\phi(2^j)}{\psi(2^j)})_{j \in \mathbb{N}_0} \in l^{\min\{q_1,1\}}$ , then

$$B_{p,q_0}^{\psi}(\mathbb{R}^d) \hookrightarrow B_{p,q_1}^{\phi}(\mathbb{R}^d).$$

We obtain the following result from the previous one.

**Proposition 59.** Let  $0 < p, q_0 \leq \infty$ ,  $1 \leq q_1 \leq \infty$  and  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$  be an admissible sequence such that  $\underline{s}(\sigma) > 0$ . For every  $0 < \epsilon < \underline{s}(\sigma)$ , we have

$$B^{\sigma}_{p,q_0}(\mathbb{R}^d) \hookrightarrow B^{\underline{s}(\sigma)-\varepsilon}_{p,q_1}(\mathbb{R}^d).$$

*Proof.* Let  $0 < \varepsilon < \underline{s}(\sigma)$ . By inequalities (1.3), there exists C > 0 such that

$$c2^{(\underline{s}(\sigma)-\varepsilon/2)j}\sigma_0 \le \sigma_j \qquad \forall j \in \mathbb{N}_0$$

which implies

$$\sum_{j=0}^{+\infty} 2^{j(\underline{s}(\sigma)-\varepsilon)} \sigma_j^{-1} < \infty$$

Proposition 58 allows to conclude.

**Corollary 60.** Let  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$  be an admissible sequence such that  $\underline{s}(\sigma) > 0$ . There exists s > 0 such that

$$B^{\sigma}_{\infty,\infty}(\mathbb{R})^d \hookrightarrow B^s_{\infty,\infty}(\mathbb{R}^d) = \Lambda^s(\mathbb{R}^d).$$

In particular, the elements of  $B^{\sigma}_{\infty,\infty}(\mathbb{R}^d)$  coincide almost everywhere with a continuous function defined on  $\mathbb{R}^d$ .

#### **1.7** Some reminders on finite differences

The concept of finite differences is at the core of the future developments done throughout this thesis. In this regard, we need to recall the two different types of finite differences as well as their main properties that are used at a later stage.

The concept of forward finite difference has already been introduced in the section 1.1.

**Example 61.** We find the following expressions

$$\Delta_h^2 f(x) = f(x+2h) - 2f(x+h) + f(x),$$

$$\Delta_h^3 f(x) = f(x+3h) - 3f(x+2h) + 3f(x+h) - f(x)$$

for the (forward) finite difference of order 2 and 3 respectively.

**Remark 62.** We can easily check recursively that we have the following expression for the finite differences:

$$\Delta_h^m f(x) := \sum_{j=0}^m (-1)^{m-j} \binom{m}{j} f(x+jh).$$

In the literature, there exists another definition of finite difference which is called "central" ([68]).

**Definition 63.** The central finite difference of order  $m \in \mathbb{N}^*$  associated to f is defined recursively by

$$\begin{split} \delta_h^1 f(x) &:= f(x + h/2) - f(x - h/2), \\ \delta_h^m f(x) &:= \delta_h^1 \delta_h^{m-1} f(x), \end{split}$$

where  $x, h \in \mathbb{R}^d$ .

**Example 64.** We find the expression

$$\delta_h^2 f(x) = f(x+h) - 2f(x) + f(x-h)$$

and

$$\delta_h^3 f(x) = f(x+3\frac{h}{2}) - 3f(x+\frac{h}{2}) + 3f(x-\frac{h}{2}) - f(x-3\frac{h}{2})$$

for the central finite difference of order 2 and 3 respectively.

**Remark 65.** As  $\delta_h^1 f(x) = \Delta_h^1 f(x - \frac{h}{2})$ , we can link the two concepts of finite difference together by the following formulae :

$$\delta_h^m f(x) = \Delta_h^m f(x - m\frac{h}{2})$$
 et  $\delta_h^m f(x + m\frac{h}{2}) = \Delta_h^m f(x).$ 

In particular we deduce the following formula for the central finite difference of f:

$$\delta_h^m f(x) = \sum_{j=0}^m (-1)^{m-j} \binom{m}{j} f(x + (j - \frac{m}{2})h).$$

**Remark 66.** In the following, we mostly consider the norm in the  $L^p(\mathbb{R}^d)$  space (for  $p \in [1, +\infty]$ ) of the finite difference of f. As

$$\|\Delta_h^m f\|_{L^p} = \|\delta_h^m f\|_{L^p}$$

for  $f \in L^p(\mathbb{R}^d)$  and  $h \in \mathbb{R}^d$ , we can use either central or forward finite differences and move from one concept to another depending on our needs without modifying the results of this thesis.

For the results of this thesis, we need to connect and control finite differences of different orders together. First, let us remark that it is easy to control finite differences of order m + 1 by finite differences of order m. Indeed, we have

$$|\Delta_h^{m+1} f(x)| = |\Delta_h^m f(x+h) - \Delta_h^m f(x)| \le 2 \sup_{y \in \{x,x+h\}} |\Delta_h^m f(y)|.$$

However, the converse is less trivial: it is more complicated to control finite differences of order m by finite differences of order m+1. To obtain such a result, we need two lemmata.

**Lemma 67.** Let  $m \in \mathbb{N}^*$  and f be a function defined on  $\mathbb{R}^d$ . We have

$$\Delta_{2h}^m f(x) = \sum_{j=0}^m \binom{m}{j} \Delta_h^m f(x+jh)$$

for all  $x, h \in \mathbb{R}^d$ .

*Proof.* We proceed by induction on m. For m = 1, this results immediately from

$$\Delta_{2h}^{1}f(x) = f(x+2h) - f(x+h) + f(x+h) - f(x) = \Delta_{h}^{1}f(x+h) + \Delta_{h}^{1}f(x).$$

We suppose that the result is true for m and we prove that it is still true for m + 1. We obtain successively

$$\begin{split} \Delta_{2h}^{m+1} f(x) &= \Delta_{2h}^{m} (\Delta_{2h}^{1}) f(x) \\ &= \sum_{j=0}^{m} \binom{m}{j} \Delta_{h}^{m} (\Delta_{2h}^{1} f)(x+jh) \\ &= \sum_{j=0}^{m} \binom{m}{j} \Delta_{h}^{m} (\Delta_{h}^{1} f(x+(j+1)h) + \Delta_{h}^{1} f(x+jh)) \\ &= \sum_{j=0}^{m} \left( \binom{m+1}{j+1} - \binom{m}{j+1} \right) \Delta_{h}^{m+1} f(x+(j+1)h) + \sum_{j=0}^{m} \binom{m}{j} \Delta_{h}^{m+1} f(x+jh) \\ &= \sum_{j=0}^{m} \binom{m}{j} \Delta_{h}^{m+1} f(x+jh) + \sum_{j=1}^{m} \left( \binom{m+1}{j} - \binom{m}{j} \right) \Delta_{h}^{m+1} f(x+jh) + \\ & \binom{m}{m} \Delta_{h}^{m+1} f(x+(m+1)h) \\ &= \binom{m+1}{0} \Delta_{h}^{m+1} f(x) + \binom{m+1}{m+1} \Delta_{h}^{m+1} f(x+(m+1)h) + \sum_{j=1}^{m} \binom{m+1}{j} \Delta_{h}^{m+1} f(x+jh) \\ &= \sum_{j=0}^{m+1} \binom{m+1}{j} \Delta_{h}^{m+1} f(x+jh). \end{split}$$

**Lemma 68.** Let  $k, m \in \mathbb{N}^*$  and f be a function defined on  $\mathbb{R}^d$ . We have

$$\Delta_{kh}^{m} f(x) = \sum_{i_1=0}^{k-1} \dots \sum_{i_m=0}^{k-1} \Delta_h^{m} f(x+i_1h+\dots+i_mh)$$

for all  $x, h \in \mathbb{R}^d$ .

*Proof.* We proceed by induction on m. For m = 1, there is nothing to prove. We suppose the result is true for m and we show that it is still true for m + 1. We obtain successively

$$\begin{split} \Delta_{kh}^{m+1} f(x) &= \Delta_{kh}^{1} \Delta_{kh}^{m} f(x) \\ &= \Delta_{kh}^{m} f(x+kh) - \Delta_{kh}^{m} f(x) \\ &= \sum_{i_{1}=0}^{k-1} \dots \sum_{i_{m}=0}^{k-1} \left( \Delta_{h}^{m} f(x+kh+i_{1}h+\ldots+i_{m}h) - \Delta_{h}^{m} f(x+i_{1}h+\ldots+i_{m}h) \right) \\ &= \sum_{i_{1}=0}^{k-1} \dots \sum_{i_{m}=0}^{k-1} \Delta_{kh}^{1} \Delta_{h}^{m} f(x+i_{1}h+\ldots+i_{m}h), \end{split}$$

so the conclusion by the case m = 1.

The next proposition is an important result that is used in section 3.1 to explicitly control finite differences of order m by finite differences of order m + 1.

**Proposition 69.** Let  $m \in \mathbb{N}^*$  and f be a function defined on  $\mathbb{R}^d$ . We have

$$|\Delta_h^m f(x+mh)| \le \frac{m}{2} \sup_{j \in \mathbb{N}_0} |\Delta_h^{m+1} f(x+jh)| + \frac{1}{2^m} |\Delta_{2h}^m f(x+(m-1)h)|$$

for all  $x, h \in \mathbb{R}^d$ .

*Proof.* We proceed by induction on m. For m = 1, we easily check that we have

$$\begin{aligned} |\Delta_h^1 f(x+h)| &= |f(x+2h) - f(x+h)| \\ &\leq \frac{1}{2} |f(x+2h) - 2f(x+h) + f(x)| + \frac{1}{2} |f(x+2h) - f(x)| \\ &\leq \frac{1}{2} |\Delta_h^2 f(x)| + \frac{1}{2} |\Delta_{2h}^1 f(x)| \end{aligned}$$

for all  $x, h \in \mathbb{R}^d$ . We suppose that the result is true for m and we prove that it is still true for m + 1. Let  $x, h \in \mathbb{R}^d$ . By the induction hypothesis, we have

$$\begin{split} |\Delta_h^{m+1} f(x+(m+1)h)| &= |\Delta_h^m (\Delta_h^1 f)((x+h)+mh)| \\ &\leq \frac{m}{2} \sup_{j \in \mathbb{N}_0} |\Delta_h^{m+1} (\Delta_h^1 f)((x+h)+jh)| + \\ &\qquad \frac{1}{2^m} |\Delta_{2h}^m (\Delta_h^1 f)((x+h)+(m-1)h)|. \end{split}$$

Let us remark that

$$\Delta_{2h}^m(\Delta_h^1 f) = \frac{1}{2} (\Delta_{2h}^m \Delta_{2h}^1 f - \Delta_{2h}^m \Delta_h^2 f).$$

#### SOME REMINDERS ON FINITE DIFFERENCES

By lemma 67, we have for all  $y \in \mathbb{R}^d$ 

$$\begin{aligned} |\Delta_{2h}^m \Delta_h^2 f(y)| &\leq \sum_{j=0}^m \binom{m}{j} |\Delta_h^{m+2} f(y+jh)| \\ &\leq 2^m \sup_{j \in \{0,\dots,m\}} |\Delta_h^{m+2} f(y+jh)|. \end{aligned}$$

So we obtain

$$|\Delta_{2h}^{m}(\Delta_{h}^{1}f)(y)| \leq \frac{1}{2} \left( |\Delta_{2h}^{m+1}f(y)| + \sup_{j \in \{0,\dots,m\}} |\Delta_{h}^{m+2}f(y+jh)| 2^{m} \right)$$

for all  $y \in \mathbb{R}^d$ . As a result, we find

$$|\Delta_h^{m+1} f(x + (m+1)h)| \le \left(\frac{m+1}{2}\right) \sup_{j \in \mathbb{N}_0} |\Delta_h^{m+2} f(x+jh)| + \frac{1}{2^{m+1}} |\Delta_{2h}^{m+1} f(x+mh)|,$$

which allows to conclude.

# Chapter 2

# Characterizations and basic properties of generalized Hölder-Zygmund spaces $\Lambda^{\sigma,\alpha}(\mathbb{R}^d)$

In this chapter we show that the main properties of the usual Hölder-Zygmund spaces are still satisfied in the generalized setting previously introduced. Consequently, such a generalization implies more flexibility for approximation purposes.

# 2.1 A reminder of the characterizations of Hölder-Zygmund spaces $\Lambda^{\alpha}(\mathbb{R}^d)$

Since the aim of the forthcoming subsections is to generalize properties related to the characterizations of the Hölder-Zygmund spaces  $\Lambda^{\alpha}(\mathbb{R}^d)$ , we first recall some of these properties. Proofs can be found in [42, 73, 84, 93] for example.

Let us first give a characterization in terms of convolution with smooth functions

**Theorem 70.** Let  $\alpha > 0$ . Then  $f \in \Lambda^{\alpha}(\mathbb{R}^d)$  if and only if there exists C > 0 and  $\Phi \in D(\mathbb{R}^d)$  so that, with  $\Phi_{\delta}(x) := \delta^{-d} \Phi(x/\delta)$ ,

$$\|f - f \star \Phi_{\delta}\|_{L^{\infty}} \le C\delta^{\alpha} \quad \forall \delta > 0.$$

Notation 71. Let  $\mathbb{P}_m$  denote the set of polynomials of degree less or equal to  $m \in \mathbb{N}_0$ . The following result shows a close link between Hölder spaces and polynomial approximations.

**Theorem 72.** Let  $\alpha > 0$  and  $f \in L^{\infty}(\mathbb{R}^d)$ . The following assertions are equivalent:

- 1.  $f \in \Lambda^{\alpha}(\mathbb{R}^d);$
- 2. there exist a constant C > 0 and  $J \in \mathbb{N}_0$  such that

$$\inf_{P \in \mathbb{P}_{\lfloor \alpha \rfloor}} \|f - P\|_{L^{\infty}(B(x, 2^{-j}))} \le C 2^{-j\alpha}, \quad \forall x \in \mathbb{R}^{d}, j \ge J.$$

#### CHARACTERIZATIONS OF $\Lambda^{\alpha}(\mathbb{R}^d)$

The next result establishes a strong connection between the somehow classical concept of regularity given by  $C^n$  spaces and the regularity expressed through Hölder spaces. Some authors (e.g. [73, 93]) even introduce these spaces in terms of such a characterization.

**Proposition 73.** Let  $\alpha > 0$ . A function  $f \in \Lambda^{\alpha}(\mathbb{R}^d)$  satisfies<sup>1</sup>

1. 
$$f \in C^{\lceil \alpha \rceil - 1}(\mathbb{R}^d),$$

- 2.  $D^{\nu}f \in L^{\infty}(\mathbb{R}^d) \ \forall |\nu| \leq \lceil \alpha \rceil 1,$
- 3.  $\sup_{|h| \le 2^{-j}} \|\Delta_h^{\lfloor \alpha \rfloor + 1 |\nu|} D^{\nu} f\|_{L^{\infty}} \le C 2^{-j(\alpha |\nu|)} \quad \forall j \in \mathbb{N}_0, |\nu| \le \lceil \alpha \rceil 1.$

Conversely, if  $f \in C^{\lceil \alpha \rceil - 1}(\mathbb{R}^d) \cap L^{\infty}(\mathbb{R}^d)$  satisfies

$$\sup_{|h| \le 2^{-j}} \|\Delta_h^{\lfloor \alpha \rfloor + 1 - |\nu|} D^{\nu} f\|_{L^{\infty}} \le C 2^{-j(\alpha - |\nu|)} \qquad \forall j \in \mathbb{N}_0, |\nu| = \lceil \alpha \rceil - 1$$

then  $f \in \Lambda^{\alpha}(\mathbb{R}^d)$ .

It is also well known that the Hölder-Zygmund spaces are closely related to the Taylor approximation.

**Theorem 74.** Let  $N \in \mathbb{N}_0$ ,  $\alpha > 0$  such that  $N < \alpha < N + 1$ . If  $f \in \Lambda^{\alpha}(\mathbb{R}^d)$ , then, for all  $x \in \mathbb{R}^d$ , we have

$$f(x+h) = \sum_{|\nu| \le N} D^{\nu} f(x) \frac{h^{\nu}}{|\nu|!} + R_N(x,h), \quad \forall h \in \mathbb{R}^d$$

where  $|R_N(x,h)| \leq C|h|^{\alpha}$ .

Conversely, if  $f \in L^{\infty}(\mathbb{R}^d) \cap C^N(\mathbb{R}^d)$  satisfies

$$f(x+h) = \sum_{|\nu| \le N} D^{\nu} f(x) \frac{h^{\nu}}{|\nu|!} + R_N(x,h) \qquad \forall x, h \in \mathbb{R}^d,$$

where  $|R_N(x,h)| \leq C|h|^{\alpha}$ , then  $f \in \Lambda^{\alpha}(\mathbb{R}^d)$ .

The next result is a characterization of Hölder-Zygmund spaces in terms of wavelet coefficients. The concept of wavelet and multiresolution analysis is recalled in section 5.4. A proof of this result can be found in [64, 93].

**Theorem 75.** Let  $\alpha > 0$  such that  $\alpha \notin \mathbb{N}_0$ . Given a multiresolution analysis of regularity  $r > \alpha$  with  $r \in \mathbb{N}_0$ , the following assertions are equivalent:

1.  $f \in \Lambda^{\alpha}(\mathbb{R}^d);$ 

<sup>&</sup>lt;sup>1</sup>The function f in the next expressions should be understood by the reader as a function which is equal almost everywhere to the original function f and which is  $(\lceil \alpha \rceil - 1)$ -times continuously differentiable. The existence of such a function is demonstrated in proposition 80.
#### SOME PRELIMINARY RESULTS

2. 
$$\exists C > 0: \begin{cases} |C_k| \leq C & \forall k \in \mathbb{Z}^d \\ |c_{j,k}^i| \leq C2^{-j\alpha} & \forall j \in \mathbb{N}_0, \forall i \in \{1, \dots, 2^d - 1\}, \forall k \in \mathbb{Z}^d, \end{cases}$$

where  $C_k$  and  $c_{j,k}^i$  denote wavelet coefficients of f associated respectively to the "father" and the "mother" wavelets (see section 5.4).

The following result shows that Hölder spaces can be written as a real interpolation of Sobolev spaces. Reminders about Sobolev spaces and the real interpolation theory are made in section 5.5. A proof of the following result can be found in [84].

**Theorem 76.** Let N, M be two natural numbers and  $\alpha > 0$  such that  $N < \alpha < M$  and  $\alpha = (1 - \theta)N + \theta M$ . Then

$$\Lambda^{\alpha}(\mathbb{R}^d) = [W_N^{\infty}, W_M^{\infty}]_{\theta, \infty, J} = [W_N^{\infty}, W_M^{\infty}]_{\theta, \infty, K}$$

### 2.2 Some preliminary results

The aim of this section is to present some basic results that are often used in the sequel.

Let  $\rho \in D(\mathbb{R}^d)$  be a function whose support is included in the closed ball  $B(0, \leq 1)$  and satisfies the following conditions:

- 1.  $0 \le \rho \le 1;$
- 2.  $\int_{\mathbb{R}^d} \rho(x) dx = 1;$
- 3.  $\rho$  is a radial function, i.e. we have  $|x| = |y| \Rightarrow \rho(x) = \rho(y)$ .

We set  $\rho_{\delta}(x) := \delta^{-d} \rho(x/\delta) \ \forall \delta > 0$ . A classical example of such a function is given by

$$\rho(x) = \begin{cases} e^{-1/(1-|x|^2)} & \text{if } |x| < 1, \\ 0 & \text{if } |x| \ge 1 \end{cases}$$

up to a normalisation factor.

**Proposition 77.** Let  $m \in \mathbb{N}$ ,  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$  be an admissible sequence and  $f \in L^1_{loc}(\mathbb{R})$  be a function such that  $\sup_{|h| \leq 2^{-j}} \|\Delta_h^m f\|_{L^{\infty}} \leq C\sigma_j \ \forall j \in \mathbb{N}_0$ . There exists  $\Phi \in D(\mathbb{R})$  such that

$$\sup_{\delta \le 2^{-j}} \| f \star \Phi_{\delta} - f \|_{L^{\infty}} \le C \sigma_j, \ \forall j \in \mathbb{N}_0.$$

*Proof.* We can increase m by 1, 2 or 3 units if necessary so that we can assume that m is equal to 2n where n is an odd integer. We set

$$\tilde{\Phi}(t) := \sum_{j=0}^{m/2-1} (-1)^j \binom{m}{j} \frac{1}{2j-m} \rho\left(\frac{t}{2j-m}\right),$$

where  $\binom{m}{j} = \frac{m!}{(m-j)!j!}$   $(m, j \in \mathbb{N}_0 \text{ and } m \ge j)$ . Let  $c_m := \int \tilde{\Phi}(t) dt = \frac{1}{2} \binom{m}{m/2}, \Phi(t) := \tilde{\Phi}(t)/c_m$ and  $\Phi_{\delta}(t) := \delta^{-1} \Phi(t/\delta)$ . We have

$$\begin{split} f \star \Phi_{\delta}(x) - f(x) &= \int f(x-t) \Phi_{\delta}(t) dt - f(x) \\ &= \int f(x-\delta t) \Phi(t) dt - f(x) \\ &= \frac{1}{c_m} \sum_{j=0}^{m/2-1} (-1)^j \binom{m}{j} \frac{1}{2j-m} \int f(x-\delta t) \rho\left(\frac{t}{2j-m}\right) dt - f(x) \\ &= \frac{1}{c_m} \sum_{j=0}^{m/2-1} (-1)^j \binom{m}{j} \int f(x-\delta(2j-m)t) \rho(t) dt - f(x) \\ &= \frac{1}{2c_m} \left( \sum_{\substack{j=0\\j \neq m/2}}^m (-1)^j \binom{m}{j} \int f(x-\delta(2j-m)t) \rho(t) dt - 2c_m f(x) \right) \\ &= \frac{1}{2c_m} \int \Delta_{2\delta t}^m f(x) \rho(t) dt. \end{split}$$

We conclude that

$$\sup_{\delta \le 2^{-j}} \|f \star \Phi_{\delta} - f\|_{L^{\infty}} \le \frac{2^m}{2c_m} C\sigma_j.$$

**Lemma 78.** Let  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$  be an admissible sequence. If  $f \in L^1_{loc}(\mathbb{R})$  is a function satisfying

$$\|f \star \rho_{2^{-j}} - f\|_{L^{\infty}(\mathbb{R})} \le C\sigma_j \quad \forall j \in \mathbb{N}_0,$$

then, for all  $k \in \mathbb{N}_0$ , we have

$$\|D^k \left(f \star \rho_{2^{-j}} - f \star \rho_{2^{-j+1}}\right)\|_{L^{\infty}(\mathbb{R})} \le C' 2^{jk} \sigma_j \qquad \forall j \in \mathbb{N}^*.$$

*Proof.* For  $\delta > 0$ , let us write

$$f \star \rho_{\delta} - f \star \rho_{2\delta} = \rho_{\delta} \star (f \star \rho_{\delta} - f \star \rho_{2\delta}) + \rho_{\delta} \star (f - f \star \rho_{\delta}) - \rho_{2\delta} \star (f - f \star \rho_{\delta})$$

One gets

$$\begin{aligned} |D^{k}\rho_{\delta} \star (f \star \rho_{\delta} - f \star \rho_{2\delta})| &\leq ||D^{k}\rho_{\delta}||_{L^{1}}||f \star \rho_{\delta} - f \star \rho_{2\delta}||_{L^{\infty}} \\ &\leq C_{1}\delta^{-k} \left(||f \star \rho_{\delta} - f||_{L^{\infty}} + ||f - f \star \rho_{2\delta}||_{L^{\infty}}\right). \end{aligned}$$

By considering  $\delta = 2^{-j}$ , the previous inequality leads to

$$|D^k \rho_{2^{-j}} \star (f \star \rho_{2^{-j}} - f \star \rho_{2^{-j+1}})| \le C 2^{jk} C(\sigma_j + \sigma_{j-1}) \le C 2^{jk} \sigma_j,$$

for all  $j \in \mathbb{N}^*$ . The two other terms in the decomposition of  $f \star \rho_{\delta}(x) - f \star \rho_{2\delta}(x)$  can be handled in the same way.

**Remark 79.** All the results from this section can easily be adapted to  $\mathbb{R}^d$ .

### 2.3 Generalized Hölder spaces $\Lambda^{\sigma,\alpha}(\mathbb{R}^d)$ and $C^k(\mathbb{R}^d)$ spaces

The next result binds the regularity of the elements of  $\Lambda^{\sigma,\alpha}(\mathbb{R}^d)$  to the classical notion of differentiability.

**Proposition 80.** Let  $m \in \mathbb{N}^*$ ,  $k \in \mathbb{N}_0$  and  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$  be an admissible sequence such that

$$\sum_{j=1}^{+\infty} 2^{jk} \sigma_j < \infty.$$

If  $f \in L^{\infty}(\mathbb{R}^d)$  satisfies

$$\sup_{|h| \le 2^{-j}} \|\Delta_h^m f\|_{L^{\infty}} \le C\sigma_j \quad \forall j \in \mathbb{N}_0,$$

then f is k-times continuously differentiable (in the sense that f coincides almost everywhere on  $\mathbb{R}^d$  with a k-times continuously differentiable function).

Proof. Let  $\Phi$  be the function defined by proposition 77 and  $f_1 := f \star \Phi_{2^{-1}}, f_j := f \star (\Phi_{2^{-j}} - \Phi_{2^{-j+1}})$  (j > 1). By proposition 77, we have  $||f_j||_{L^{\infty}} \leq C\sigma_j$  for all  $j \in \mathbb{N}^*$ , where the constant C does not depend on j. We thus get

$$\sum_{j=1}^k ||f_j||_{L^{\infty}} \le C \sum_{j=1}^k \sigma_j,$$

for all  $k \in \mathbb{N}^*$ . So, the series  $\sum_{j=1}^{+\infty} f_j$  converges uniformly on  $\mathbb{R}^d$  to a function which coincides almost everywhere with f. Moreover, we have

$$\|D^{\alpha}f_j\|_{L^{\infty}} \le C2^{jk}\sigma_j \quad \forall j \in \mathbb{N}^*, |\alpha| \le k.$$

Therefore, the series  $\sum_{j=1}^{\infty} D^{\alpha} f_j$  converges uniformly, which leads to the conclusion.  $\Box$ 

**Remark 81.** This result does not depend on the order m of the finite difference.

Under the assumptions of theorem 33, this result can be rewritten in the following way.

**Corollary 82.** Let  $k \in \mathbb{N}_0$ ,  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$  be an admissible sequence such that  $\underline{s}(\sigma^{-1}) > 0$ and

$$\sum_{j=1}^{+\infty} 2^{jk} \sigma_j < \infty.$$

We have

$$B^{\sigma^{-1}}_{\infty,\infty}(\mathbb{R}^d) = \Lambda^{\sigma,\overline{s}(\sigma^{-1})}(\mathbb{R}^d) \subseteq C^k(\mathbb{R}^d).$$

If  $\underline{s}(\sigma^{-1}) > 0$ , the last result is always satisfied for k = 0. In this case, the elements of  $\Lambda^{\sigma,\overline{s}(\sigma^{-1})}(\mathbb{R}^d)$  are continuous.

**Remark 83.** If we apply corollary 82 to the admissible sequence  $\sigma_j = 2^{-j\alpha}$  ( $\alpha > 0$ ), we recover the differentiability properties of classical Hölder-Zygmund spaces. Corollary 82 gives additional information about differentiability properties of those spaces. Indeed, it is well known that the elements of  $\Lambda^m(\mathbb{R}^d)$  are at least (m-1)-times continuously differentiable  $(m \in \mathbb{N}^*)$  and that the elements of  $\Lambda^{m+\epsilon}(\mathbb{R}^d)$  are *m*-times continuously differentiable for any  $\epsilon > 0$ . A question that naturally arises is the following: can we sensibly modify the admissible sequences of the usual Hölder spaces  $\Lambda^m(\mathbb{R}^d)$  so that its elements become *m*-times continuously differentiable? The answer is also given by corollary 82, which states that if we replace the sequence  $(2^{-jm})_j$  with  $\sigma_j^{(m)} = 2^{-jm}j^{-1}\log(j)^{-(1+\epsilon)}$   $(j \in \mathbb{N}_0, j > 1,$  $\epsilon > 0$ ), then we have  $\Lambda^{\sigma^{(m)},m}(\mathbb{R}^d) \subset C^m(\mathbb{R}^d)$ . This is a direct consequence of the study of Bertrand series, which is recalled in the next result:

**Lemma 84.** Let  $\alpha, \beta \in \mathbb{R}$ . The series

$$\sum_{j\geq 2} \frac{1}{j^{\alpha}(\log(j))^{\beta}}$$

converges if and only if  $\alpha > 1$  or  $(\alpha = 1 \text{ and } \beta > 1)$ 

**Remark 85.** Let  $m \in \mathbb{N}_0$ . If  $\sigma$  is an admissible sequence satisfying

$$\sum_{j=1}^{+\infty} 2^{j(m+1)} \sigma_j < \infty, \tag{2.1}$$

then the generalized Hölder-Zygmund space  $\Lambda^{\sigma,m}(\mathbb{R}^d)$  is composed of constant functions.

Let us prove this assertion. If  $f \in \Lambda^{\sigma,m}(\mathbb{R}^d)$ , then the function f belongs to  $C^{m+1}(\mathbb{R}^d)$ by proposition 80. By lemma 5, we find

$$\frac{|\Delta_{2^{-j}e_i}^{m+1}f(x)|}{2^{-j(m+1)}} \to |D_{x_i}^{m+1}f(x)|$$

and

$$\frac{\Delta_{2^{-j}e_i}^{m+1}f(x)|}{2^{-j(m+1)}} \le C2^{j(m+1)}\sigma_j \to 0 \text{ if } j \to +\infty,$$

so that  $D_{x_i}^{m+1}f = 0$  for all  $i \in \{1, \ldots, d\}$ . For all  $j \in \{1, \ldots, d\}$ , the function f can be written as

$$f(x_1, ..., x_j, ..., x_d) = \sum_{i=0}^m a_{x_1, ..., x_{j-1}, x_{j+1}, ..., x_d}^{(i)} x_j^i.$$

As  $f \in L^{\infty}(\mathbb{R}^d)$ , we have

$$f(x_1, ..., x_j, ..., x_d) = a_{x_1, ..., x_{j-1}, x_{j+1}, ..., x_d}^{(0)}$$

so that  $D_{x_i}f = 0$ . As a conclusion, the function f is constant.

The condition (2.1) implies that the sequence  $(\sigma_j)_{j\in\mathbb{N}^*}$  converges faster to 0 than  $(2^{-j(m+1)})_{j\in\mathbb{N}^*}$ . In particular, if  $m+1 < \underline{s}(\sigma^{-1})$   $(m \in \mathbb{N}_0)$ , then the spaces  $\Lambda^{\sigma,m}$  are composed of constant functions.

# 2.4 A characterization of spaces $\Lambda^{\sigma,\alpha}(\mathbb{R}^d)$ in terms of convolution product

The aim of this section is to generalize theorem 70 to spaces  $\Lambda^{\sigma,\alpha}(\mathbb{R}^d)$ . The following result gives the converse of proposition 77.

**Proposition 86.** Let  $m \in \mathbb{N}^*$ ,  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$  be an admissible sequence satisfying

$$\forall J \in \mathbb{N}_0, \quad \sum_{j=0}^J 2^{jm} \sigma_j \le C 2^{Jm} \sigma_J \tag{2.2}$$

and

$$\forall J \in \mathbb{N}_0, \quad \sum_{j=J}^{+\infty} \sigma_j \le C\sigma_J. \tag{2.3}$$

If  $f \in L^{\infty}(\mathbb{R}^d)$  is a function for which there exists  $\Phi \in D(\mathbb{R}^d)$  satisfying

$$\|f \star \Phi_{2^{-j}} - f\|_{L^{\infty}} \le C\sigma_j \quad \forall j \in \mathbb{N}_0,$$

then we have

$$\sup_{|h| \le 2^{-j}} \|\Delta_h^m f\|_{L^{\infty}} \le C\sigma_j \quad \forall j \in \mathbb{N}_0.$$

*Proof.* We keep the same notations as the ones introduced in the proof of proposition 80. We know that  $\Delta_h^m f = \sum_{j=1}^{+\infty} \Delta_h^m f_j$  with uniform convergence on  $\mathbb{R}^d$ . For all  $N \in \mathbb{N}^*$ , we have

$$\begin{split} \|\Delta_{h}^{m}f\|_{L^{\infty}} &\leq \sum_{j=1}^{N} \|\Delta_{h}^{m}f_{j}\|_{L^{\infty}} + \sum_{j=N+1}^{+\infty} \|\Delta_{h}^{m}f_{j}\|_{L^{\infty}} \\ &\leq C\sum_{j=1}^{N} |h|^{m} \sup_{|\alpha|=m} \|D^{\alpha}f_{j}\|_{L^{\infty}} + \sum_{j=N+1}^{+\infty} 2^{m} \|f_{j}\|_{L^{\infty}} \\ &\leq C|h|^{m} \sum_{j=1}^{N} 2^{jm}\sigma_{j} + C\sum_{j=N+1}^{+\infty} \sigma_{j}, \end{split}$$

 $\mathbf{SO}$ 

$$\|\Delta_h^m f\|_{L^{\infty}} \le C(1+|h|^m 2^{Nm})\sigma_N.$$

We get

$$\sup_{|h| \le 2^{-N}} \|\Delta_h^m f\|_{L^{\infty}} \le C\sigma_N \qquad \forall N \in \mathbb{N}_0,$$

which ends the proof.

The following corollary is the main result of this section.

Corollary 87. (D.K., S. Nicolay) Let  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$  be an admissible sequence such that  $\underline{s}(\sigma^{-1}) > 0$ . We have

$$B_{\infty,\infty}^{\sigma^{-1}}(\mathbb{R}^d) = \Lambda^{\sigma,\overline{s}(\sigma^{-1})}(\mathbb{R}^d) = \left\{ f \in L^{\infty}(\mathbb{R}^d) : \exists \Phi \in D(\mathbb{R}^d) \quad \sup_{j \in \mathbb{N}_0} \left( \sigma_j^{-1} \sup_{\delta \le 2^{-j}} \| f \star \Phi_\delta - f \|_{L^{\infty}} \right) < \infty \right\}.$$
(2.4)

Moreover, the norm

$$\|f\|_{L^{\infty}} + \inf \left\{ \sup_{j \in \mathbb{N}_0} \left( \sigma_j^{-1} \sup_{\delta \le 2^{-j}} \|f \star \Phi_{\delta} - f\|_{L^{\infty}} \right) \right\},\$$

where the infimum is taken on the set of functions  $\Phi \in D(\mathbb{R}^d)$  satisfying (2.4) and  $\sup_{|\alpha|=\lfloor \overline{s}(\sigma^{-1})\rfloor+1} \|D^{\alpha}\Phi\|_{L^1(\mathbb{R}^d)} \leq 2^{\lfloor \overline{s}(\sigma^{-1})\rfloor+1}$ , is equivalent to  $\|f\|_{\Lambda^{\sigma,\overline{s}(\sigma^{-1})}}$ .

*Proof.* This is a consequence of propositions 48, 77, 86 and from lemmata 35 and 53.  $\Box$ 

## 2.5 A polynomial characterization of spaces $\Lambda^{\sigma,\alpha}(\mathbb{R}^d)$

The following result shows that generalized Hölder-Zygmund spaces and polynomials approximations are closely linked. It is a generalization of theorem 72.

**Theorem 88.** Let  $m \in \mathbb{N}^*$ ,  $f \in L^{\infty}(\mathbb{R}^d)$  be a continuous function on  $\mathbb{R}^d$  and  $(\sigma_j)_{j \in \mathbb{N}_0}$  be an admissible sequence. The following assertions are equivalent:

34

### A POLYNOMIAL CHARACTERIZATION OF SPACES $\Lambda^{\sigma,\alpha}(\mathbb{R}^d)$

- 1. there exists a constant C > 0 such that  $\sup_{|h| < 2^{-j}} \|\Delta_h^m f\|_{L^{\infty}} \leq C\sigma_j \ \forall j \in \mathbb{N}_0$ ,
- 2. there exist a constant C > 0 and a natural number J such that

$$\inf_{P\in\mathbb{P}_{m-1}} \|f-P\|_{L^{\infty}(B(x,2^{-j}))} \le C\sigma_j,$$

 $\forall x \in \mathbb{R}^d, j \ge J.$ 

Notation 89. We define the following notation:

$$B_h(x_0, r) := \{ x \in B(x_0, r) : [x, x + mh] \subset B(x_0, r) \},\$$

where [a, b] refers to the segment in  $\mathbb{R}^d$  joining the points a and b.

We need the following classical result ([24]).

**Theorem 90.** Let  $m \in \mathbb{N}^*$  and  $f \in L^{\infty}(\mathbb{R}^d)$ . There exists a constant C > 0 (which depends only on m and d) such that for all  $x_0 \in \mathbb{R}^d$  and r > 0, we have

$$\inf_{P \in \mathbb{P}_{m-1}} \|f - P\|_{L^{\infty}(B(x_0, r))} \le C \sup_{|h| \le r} \|\Delta_h^m f\|_{L^{\infty}(B_h(x_0, r))}.$$

Proof of theorem 88. Implication  $1 \Rightarrow 2$  is a consequence of theorem 90. Let us prove that  $2 \Rightarrow 1$ . We remark that for all  $x \in \mathbb{R}^d$  and  $j \ge J$ , there exists a polynomial  $P \in \mathbb{P}_{m-1}$  such that

$$\sup_{y \in B(x,2^{-j})} |f(y) - P(y)| \le C\sigma_j.$$

For any polynomial  $P \in \mathbb{P}_{m-1}$ , we have

$$\begin{aligned} |\Delta_h^m f(x)| &= |\Delta_h^m (f - P)(x)| \\ &\leq 2^m \sup_{y \in \{x, \dots, x+mh\}} |f(y) - P(y)|. \end{aligned}$$

Then, for  $|h| \leq 2^{-(j+m)}$ , we have

$$\begin{aligned} |\Delta_h^m f(x)| &\leq 2^m C \sigma_j \\ &\leq 2^m C d_0^{-m} \sigma_{j+m}, \end{aligned}$$

which ends the proof.

One gets the following corollary.

**Corollary 91.** (D.K., S. Nicolay) Let  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$  be an admissible sequence such that  $\underline{s}(\sigma^{-1}) > 0$ . If  $M \in \mathbb{N}_0$  is such that  $M > \overline{s}(\sigma^{-1})$ , then

$$B_{\infty,\infty}^{\sigma^{-1}}(\mathbb{R}^d) = \Lambda^{\sigma,\overline{s}(\sigma^{-1})}(\mathbb{R}^d) = \left\{ f \in L^{\infty}(\mathbb{R}^d) : \sup_{x \in \mathbb{R}^d} \left( \sup_{j \in \mathbb{N}_0} \left( \sigma_j^{-1} \inf_{P \in \mathbb{P}_{M-1}} \|f - P\|_{L^{\infty}(B(x,2^{-j}))} \right) \right) < \infty \right\}.$$

Moreover, the semi-norm  $\sup_{x \in \mathbb{R}^d} \left( \sup_{j \in \mathbb{N}_0} \left( \sigma_j^{-1} \inf_{P \in \mathbb{P}_{M-1}} \|f - P\|_{L^{\infty}(B(x,2^{-j}))} \right) \right)$  is equivalent to  $\|f\|_{\Lambda^{\sigma,\overline{s}(\sigma^{-1})}}$ .

### 2.6 A characterization of spaces $\Lambda^{\sigma,\alpha}(\mathbb{R}^d)$ in terms of derivatives

The goal of this section is to present a characterization of the generalized Hölder spaces  $\Lambda^{\sigma,\bar{s}(\sigma^{-1})}(\mathbb{R}^d)$  in terms of derivatives of their elements. The following result generalizes proposition 73.

**Proposition 92.** (D.K., S. Nicolay) Let  $\sigma$  be an admissible sequence and N, M be two natural numbers such that  $N < \underline{s}(\sigma^{-1}) \leq \overline{s}(\sigma^{-1}) < M$ . If f belongs to  $\Lambda^{\sigma,\overline{s}(\sigma^{-1})}(\mathbb{R}^d)$ , then we have

- 1.  $f \in C^N(\mathbb{R}^d)$ ,
- 2.  $D^{\nu}f \in L^{\infty}(\mathbb{R}^d) \ \forall |\nu| \leq N,$
- 3.  $\sup_{|h| \le 2^{-j}} \|\Delta_h^{M-|\nu|} D^{\nu} f\|_{L^{\infty}} \le C \sigma_j 2^{j|\nu|} \quad \forall j \in \mathbb{N}_0, |\nu| \le N.$

Conversely, if  $f \in C^N(\mathbb{R}^d) \cap L^\infty(\mathbb{R}^d)$  satisfies

$$\sup_{|h| \le 2^{-j}} \|\Delta_h^{M-|\nu|} D^{\nu} f\|_{L^{\infty}} \le C\sigma_j 2^{j|\nu|} \qquad \forall j \in \mathbb{N}_0, |\nu| = N,$$

then  $f \in \Lambda^{\sigma,\overline{s}(\sigma^{-1})}(\mathbb{R}^d)$ .

*Proof.* Let  $f \in \Lambda^{\sigma,\overline{s}(\sigma^{-1})}(\mathbb{R}^d)$ . Using the same notations as in proposition 80, we have

$$\sum_{j=1}^{+\infty} D^{\nu} f_j = D^{\nu} f \text{ (uniformly)} \quad \forall |\nu| \le N.$$

By lemma 78, we get

$$\sum_{j=1}^{+\infty} \|D^{\nu} f_j\|_{L^{\infty}} \le C \sum_{j=1}^{+\infty} 2^{jN} \sigma_j < +\infty,$$

which proves the two first assertions. Let  $\nu \in \mathbb{N}_0^d$  be a multi-index such that  $|\nu| \leq N$ ,  $h \in \mathbb{R}^d$  and  $J \in \mathbb{N}_0$  such that  $|h| \leq 2^{-J}$ . By the mean value theorem, we have

$$\begin{split} \|\Delta_{h}^{M-|\nu|}D^{\nu}f\|_{L^{\infty}} &\leq \sum_{j=1}^{J} \|\Delta_{h}^{M-|\nu|}D^{\nu}f_{j}\|_{L^{\infty}} + \sum_{j=J+1}^{+\infty} \|\Delta_{h}^{M-|\nu|}D^{\nu}f_{j}\|_{L^{\infty}} \\ &\leq \sum_{j=1}^{J} |h|^{M-|\nu|} \sup_{|\alpha|=M-|\nu|} \|D^{\alpha+\nu}f_{j}\|_{L^{\infty}} + C\sum_{j=J+1}^{+\infty} |h|^{N-|\nu|} \sup_{|\alpha|=N-|\nu|} \|D^{\alpha+\nu}f_{j}\|_{L^{\infty}} \\ &\leq C\sum_{j=1}^{J} |h|^{M-|\nu|} 2^{Mj}\sigma_{j} + C\sum_{j=J+1}^{+\infty} |h|^{N-|\nu|}\sigma_{j} 2^{Nj} \\ &\leq C2^{J|\nu|}\sigma_{J}. \end{split}$$

Let us prove the converse. Let  $|h| \leq 2^{-j}$ ; by the mean value theorem, we have

$$\begin{split} \|\Delta_h^M f\|_{L^{\infty}} &\leq C |h|^N \sup_{|\nu|=N} \|\Delta_h^{M-N} D^{\nu} f\|_{L^{\infty}} \\ &\leq C 2^{-jN} 2^{jN} \sigma_j = C \sigma_j. \end{split}$$

Theorem 33 leads to the conclusion.

We can explain the last result in the following way: the value of  $\underline{s}(\sigma^{-1})$  characterizes the number of times a function of  $\Lambda^{\sigma,\overline{s}(\sigma^{-1})}$  is differentiable. On the other hand, the value of  $\overline{s}(\sigma^{-1})$  is linked to the order of the finite difference.

The same proof as the one given in proposition 92 leads to the following result, which ends our study of the generalized Hölder spaces in terms of derivatives.

**Lemma 93.** Let  $N \in \mathbb{N}_0$ ,  $\sigma$  be an admissible sequence and  $f \in C^N(\mathbb{R}^d) \cap L^{\infty}(\mathbb{R}^d)$ . If there exists a natural number M > N such that

$$\sup_{|h| \le 2^{-j}} \|\Delta_h^{M-N} D^{\nu} f\|_{L^{\infty}} \le C\sigma_j 2^{jN} \quad \forall |\nu| = N$$

then  $f \in \Lambda^{\sigma, M-1}(\mathbb{R}^d)$ .

The next result is a characterization of the generalized Hölder-Zygmund spaces in terms of derivatives.

Corollary 94. (D.K., S. Nicolay) Let  $\sigma$  be an admissible sequence and N, M be two natural numbers such that  $N < \underline{s}(\sigma^{-1}) \leq \overline{s}(\sigma^{-1}) < M$ . Then

$$B_{\infty,\infty}^{\sigma^{-1}}(\mathbb{R}^d) = \Lambda^{\sigma,\overline{s}(\sigma^{-1})}(\mathbb{R}^d) = \{ f \in L^{\infty}(\mathbb{R}^d) \cap C^N(\mathbb{R}^d) :$$
$$\sup_{|h| \le 2^{-j}} \|\Delta_h^{M-N} D^{\nu} f\|_{L^{\infty}} \le C\sigma_j 2^{jN} \quad \forall j \in \mathbb{N}_0, |\nu| = N \}.$$
(2.5)

## 2.7 A Characterization of spaces $\Lambda^{\sigma,\alpha}(\mathbb{R}^d)$ in terms of Taylor decomposition

The first goal of this section is to present a particular case of admissible sequences called strong admissible sequences, which rely on strong assumptions about the convergence to 0. This concept has already been introduced by S. Jaffard and Y. Meyer ([65]) and by M. Clausel ([30]) in the particular case of moduli of continuity (in section 1.5, we have seen that a modulus of continuity leads to an admissible sequence, whereas the converse is false). Examples of such sequences are then provided to bring a better understanding of the underlying conditions.

Secondly, a characterization of spaces  $\Lambda^{\sigma,\overline{s}(\sigma^{-1})}(\mathbb{R}^d)$  in terms of a Taylor decomposition is presented, under the assumption that the admissible sequence  $\sigma$  is strong. This characterization refines the polynomial approximation of those spaces obtained in the section 2.5:

the characterization given by corollary 91 gives an approximation of functions in terms of polynomials, but the choice of those polynomials depends on the scale j. The characterization presented here not only allows to break away from this dependence on the scale j, but also gives the expression of the polynomial that approximates f. This polynomial is expressed only in terms of derivatives of the function f.

First, we define the concept of strong admissible sequence.

**Definition 95.** An admissible sequence  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$  is strong of order  $N \in \mathbb{N}^*$  if it satisfies

$$\sum_{j=0}^{J} 2^{Nj} \sigma_j \le C 2^{NJ} \sigma_J, \tag{2.6}$$

$$\sum_{j=J}^{+\infty} 2^{(N-1)j} \sigma_j \le C 2^{(N-1)J} \sigma_J$$
(2.7)

for all  $J \in \mathbb{N}_0$ .

Before providing some examples of strong admissible sequences in the sequel, let us study the implications of being a strong admissible sequences on the Boyd indices. Those preliminary results lead to a better understanding of the underlying conditions. They also lead to an easier method for constructing such sequences.

The following result shows that the concept of strong admissible sequence implies strong conditions on the values of the Boyd indices  $\underline{s}(\sigma^{-1})$  and  $\overline{s}(\sigma^{-1})$ .

**Lemma 96.** If  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$  is a strong admissible sequence of order  $N \in \mathbb{N}^*$ , then

$$N-1 \le \underline{s}(\sigma^{-1}) \le \overline{s}(\sigma^{-1}) \le N.$$

*Proof.* We have

$$2^{Nk}\sigma_k \le \sum_{l=0}^{j+k} 2^{Nl}\sigma_l \le C2^{N(j+k)}\sigma_{j+k},$$

so that

$$\frac{\sigma_k}{\sigma_{j+k}} \le C2^{Nj}$$

which implies

$$\lim_{j \to +\infty} \frac{\log_2(\sup_k \frac{\sigma_k}{\sigma_{j+k}})}{j} \le N$$

and  $\overline{s}(\sigma^{-1}) \leq N$ . Similarly, we have

$$2^{(N-1)(j+k)}\sigma_{j+k} \le \sum_{l=k}^{+\infty} 2^{(N-1)l}\sigma_l \le C2^{(N-1)k}\sigma_k$$

and

$$\frac{2^{(N-1)j}}{C} \le \frac{\sigma_k}{\sigma_{j+k}},$$

so  $N-1 \leq \underline{s}(\sigma^{-1})$ .

	-	-	1

The following lemma gives some partial converse to the previous result.

**Lemma 97.** Let  $N \in \mathbb{N}^*$ . If  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$  is an admissible sequence satisfying

$$N-1 < \underline{s}(\sigma^{-1})$$
 and  $\overline{s}(\sigma^{-1}) < N$ 

then  $\sigma$  is a strong admissible sequence of order N.

*Proof.* Inequalities (2.6) and (2.7) follow from lemmata 53 and 54.  $\Box$ 

The following result clarifies the concept of strong admissible sequence.

**Lemma 98.** If  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$  is a strong admissible sequence of order  $N \in \mathbb{N}^*$ , then there exist two constants  $C_1, C_2 > 0$  such that

$$C_1 2^{-jN} \le \sigma_j \le C_2 2^{-j(N-1)} \qquad \forall j \in \mathbb{N}_0$$

As a consequence, we have

$$\Lambda^{N}(\mathbb{R}^{d}) \hookrightarrow \Lambda^{\sigma,\overline{s}(\sigma^{-1})} \hookrightarrow \Lambda^{N-1}(\mathbb{R}^{d}).$$

*Proof.* The second inequality follows from the definition. Let us prove the first one. We have

$$2^{0}\sigma_{0} \leq \sum_{j=0}^{J} 2^{Nj}\sigma_{j} \leq C2^{NJ}\sigma_{J} \qquad \forall J \in \mathbb{N}_{0}$$

The conclusion can be deduced from lemma 96.

As a result, generalized Hölder spaces associated to strong admissible sequences "lie between" classical Hölder spaces superscripted with two consecutive natural numbers.

**Example 99.** Let us give two examples of strong admissible sequences.

- 1. The admissible sequence  $\sigma_j = 2^{-j\alpha}$  where  $\alpha \in ]0, +\infty[\backslash \mathbb{N}_0 \text{ is strong of order } \lfloor \alpha \rfloor + 1.$
- 2. If  $\alpha \in ]0, +\infty[\backslash \mathbb{N}_0 \text{ and } \beta \in \mathbb{R}$ , the admissible sequence  $\sigma_j = 2^{-j\alpha} j^\beta$  is strong of order  $\lfloor \alpha \rfloor + 1$ .

Example 100. Let us now give two examples of admissible sequences that are not strong.

1. If  $\alpha \in \mathbb{N}^*$ , then the sequence  $\sigma_j = 2^{-j\alpha}$  is not strong of order  $\alpha + 1$ . It is not even strong for any order  $N \in \mathbb{N}^*$ . Indeed, in the case where  $\alpha + 1 \leq N$ , we have

$$\sum_{j=J}^{+\infty} 2^{(N-1)j} \sigma_j = +\infty$$

and condition (2.7) cannot be satisfied. If  $\alpha > N$ , then

$$2^{NJ}\sigma_J \to 0$$
 if  $J \to +\infty$ 

and inequality (2.6) should imply  $\sum_{j=0}^{J} 2^{Nj} \sigma_j \to 0$  if  $J \to \infty$ , which is absurd. Finally, if  $\alpha = N$ , we obtain

$$\sum_{j=0}^{J} 2^{Nj} \sigma_j = J$$

so that inequality (2.6) can not be satisfied.

2. Similarly, one can easily show that the sequence  $\sigma_j = 2^{-j\alpha} j^\beta$  with  $\alpha \in \mathbb{N}^*$  and  $\beta \in \mathbb{R}$  is not strong for any natural number N.

**Example 101.** Let us consider the examples described in section 1.2 in order to determine if they are strong or not. If the Boyd indices are known, then lemmata 96, 97 and 98 can be used to give an answer to the question.

1. Let us consider example 18. The sequence  $\psi = \sigma^{-1}$  converges to 0. Let us prove that  $\psi$  is not strong. If  $j \in \mathbb{N}^*$ , as  $\psi$  is constant for all the indices in the intervals of type  $[j_{2n}, j_{2n+1}] \cap \mathbb{N}_0$ , we have

$$\inf_{k\geq 0} \left(\frac{\psi_k}{\psi_{j+k}}\right) = 1$$

so that  $\underline{s}(\psi^{-1}) = 0$ . We can easily check that we have

$$\sup_{k \ge 0} \left( \frac{\psi_k}{\psi_{j+k}} \right) = 2^{2j}$$

so that  $\overline{s}(\psi^{-1}) = 2$ . Hence the conclusion follows from lemma 96.

2. Let us consider example 19. The sequence  $\eta_j = 2^{-js} \psi_j$  satisfies

$$\inf_{k \ge 0} \left( \frac{\eta_k}{\eta_{j+k}} \right) = 2^{js} \inf_{k \ge 0} \left( \frac{\psi_k}{\psi_{j+k}} \right) = 2^{js}$$

and

$$\sup_{k\geq 0} \left(\frac{\eta_k}{\eta_{j+k}}\right) = 2^{js} 2^{2j}$$

This implies  $\underline{s}(\eta^{-1}) = s$  and  $\overline{s}(\eta^{-1}) = s + 2$ , which leads to the conclusion.

3. We can modify the previous example in order to construct an admissible sequence which satisfies the conditions of lemma 97. This example has been published in [80]. Let  $s_0 \ge 0$  and  $s_1 > 0$ . Let  $(j_n)_{n \in \mathbb{N}_0}$  be defined as in example 18. We define  $\sigma$  by induction in the following way: we put  $\sigma_0 = 1$  and

$$\sigma_{j+1} = \begin{cases} \sigma_j 2^{s_0} & \text{if } j_{2n} \le j < j_{2n+1}, \\ \sigma_j 2^{s_0+s_1} & \text{if } j_{2n+1} \le j < j_{2n+2} \end{cases}$$

Let us prove that  $\underline{s}(\sigma) = s_0$  and  $\overline{s}(\sigma) = s_0 + s_1$ . Let us first remark that the sequence can be rewritten as

$$\sigma_j = 2^{s_0 j} \psi_j$$

### CHARACTERIZATION OF $\Lambda^{\sigma,\alpha}(\mathbb{R}^d)$ IN TERMS OF TAYLOR DECOMPOSITION 41

where  $\psi$  is an admissible sequence satisfying

$$\psi_{j+1} = \begin{cases} \psi_j & \text{if } j_{2n} \le j < j_{2n+1}, \\ \psi_j 2^{s_1} & \text{if } j_{2n+1} \le j < j_{2n+2} \end{cases}$$

The following lemma allows to determine the Boyd indices of  $\sigma$  from the ones of  $\psi$ .

**Lemma 102.** Let  $\alpha \in \mathbb{R}$  and  $\sigma$  be an admissible sequence. We have

$$\underline{s}(2^{\alpha j}\sigma_j) = \alpha + \underline{s}(\sigma) \quad and \quad \overline{s}(2^{\alpha j}\sigma_j) = \alpha + \overline{s}(\sigma).$$

*Proof.* This is an immediate consequence of the definition of the Boyd indices.  $\Box$ 

Let us now determine the Boyd indices of the sequence  $\psi$ . Let  $j \in \mathbb{N}_0$ ; for some sufficiently large value of  $n \in \mathbb{N}_0$ , we have  $j_{2n} \leq j_{2n} + j < j_{2n+1}$ , which implies

$$\underline{\psi}_{j} = \inf_{k \ge 0} \frac{\psi_{j+k}}{\psi_{k}} = \frac{\psi_{j_{2n}+j}}{\psi_{j_{2n}}} = 1.$$

So, we have  $\underline{s}(\psi) = 0$ . Similarly, for a fixed value  $j \in \mathbb{N}_0$ , there exists a natural number *n* sufficiently large so that  $j_{2n+1} \leq j_{2n+1} + j < j_{2n+2}$ , which implies

$$\overline{\psi}_j = \sup_{k \ge 0} \frac{\psi_{j+k}}{\psi_k} = \frac{\psi_{j_{2n+1}+j}}{\psi_{j_{2n+1}}} = 2^{s_1 j}.$$

We thus have  $\overline{s}(\psi) = s_1$ . The Boyd indices of the sequence  $\sigma$  are therefore  $\underline{s}(\sigma) = s_0$ and  $\overline{s}(\sigma) = s_0 + s_1$ .

An interesting consequence can be deduced from this example: let  $N \in \mathbb{N}^*$ ; for any  $\alpha, \beta$  satisfying  $N - 1 < \alpha \leq \beta < N$ , we can construct a strong admissible sequence  $\sigma$  for which  $\underline{s}(\sigma^{-1}) = \alpha$  and  $\overline{s}(\sigma^{-1}) = \beta$ .

**Remark 103.** The same proof as in proposition 92 applied to strong admissible sequences of order  $L \in \mathbb{N}^*$  shows that proposition 92 is still true for  $L - 1 \leq \underline{s}(\sigma^{-1}) \leq \overline{s}(\sigma^{-1}) \leq L$  (where the limit values are allowed).

Let us generalize theorem 74 to generalized Hölder-Zygmund spaces. We need a lemma to prove this result.

**Lemma 104.** Let  $f \in C^k(\mathbb{R}^d)$  where  $k \in \mathbb{N}_0$ . We have

$$f(x+h) = \sum_{|\alpha| \le k} D^{\alpha} f(x) \frac{h^{\alpha}}{|\alpha|!} + R_k(x,h) \frac{|h|^k}{k!} \qquad \text{for all } x, \ h \in \mathbb{R}^d$$

where

$$|R_k(x,h)| \le \sum_{|\alpha|=k} \sup_{|l|\le |h|} \|\Delta_l^1 D^{\alpha} f\|_{L^{\infty}}.$$

*Proof.* By considering the function g defined on  $\mathbb{R}$  by  $g: t \mapsto f(x+th)$ , we can assume that d = 1. Using integration by parts, we find

$$\begin{split} f(x+h) - f(x) &= \int_0^h Df(x+t)dt \\ &= Df(x+t)(t-h)|_0^h - \int_0^h D^2 f(x+t)(t-h)dt \\ &= Df(x)h - \int_0^h D^2 f(x+t)(t-h)dt \\ &= \cdots \\ &= \sum_{j=1}^{k-1} D^j(x)\frac{h^j}{j!} + (-1)^{k-1} \int_0^h D^k f(x+t)\frac{(t-h)^{k-1}}{(k-1)!}dt \\ &= \sum_{j=1}^{k-1} D^j(x)\frac{h^j}{j!} + (-1)^{k-1} \int_0^h D^k f(x)\frac{(t-h)^{k-1}}{(k-1)!}dt + \\ &\qquad (-1)^{k-1} \int_0^h (D^k f(x+t) - D^k f(x))\frac{(t-h)^{k-1}}{(k-1)!}dt \\ &= \sum_{j=1}^k D^j(x)\frac{h^j}{j!} + (-1)^{k-1} \int_0^h (D^k f(x+t) - D^k f(x))\frac{(t-h)^{k-1}}{(k-1)!}dt, \end{split}$$

which leads to the conclusion if we set

$$R_k(x,h) = \frac{(-1)^{k-1} \int_0^h (D^k f(x+t) - D^k f(x)) \frac{(t-h)^{k-1}}{(k-1)!} dt}{\int_0^h \frac{|t-h|^{k-1}}{(k-1)!} dt}.$$

The following result gives the Taylor approximation of elements in  $\Lambda^{\sigma,\overline{s}(\sigma^{-1})}(\mathbb{R}^d)$ .

**Corollary 105.** (D.K., S. Nicolay) Let  $\sigma$  be a strong admissible sequence of order  $N \in \mathbb{N}^*$ . If  $f \in \Lambda^{\sigma,\overline{s}(\sigma^{-1})}(\mathbb{R}^d)$ , then, for all  $x \in \mathbb{R}^d$ , we have

$$f(x+h) = \sum_{|\nu| \le N-1} D^{\nu} f(x) \frac{h^{\nu}}{|\nu|!} + R_{N-1}(x,h) \frac{|h|^{N-1}}{(N-1)!}, \quad \forall h \in \mathbb{R}^d$$

where  $|R_{N-1}(x,h)| \leq C\sigma_j 2^{j(N-1)}, \forall |h| \leq 2^{-j}$ . Conversely, if  $f \in L^{\infty}(\mathbb{R}^d) \cap C^{N-1}(\mathbb{R}^d)$  satisfies

$$f(x+h) = \sum_{|\nu| \le N-1} D^{\nu} f(x) \frac{h^{\nu}}{|\nu|!} + R_{N-1}(x,h) \frac{|h|^{N-1}}{(N-1)!} \qquad \forall x,h \in \mathbb{R}^d$$
(2.8)

with  $\sup_{x,|h|\leq 2^{-j}} |R_{N-1}(x,h)| \leq C\sigma_j 2^{j(N-1)} \ \forall j \in \mathbb{N}^*$ , then  $f \in \Lambda^{\sigma,\overline{s}(\sigma^{-1})}(\mathbb{R}^d)$ .

*Proof.* Let  $f \in \Lambda^{\sigma,\overline{s}(\sigma^{-1})}(\mathbb{R}^d)$ . We know that  $f \in C^{N-1}(\mathbb{R}^d)$ . By lemma 104, we get

$$f(x+h) = \sum_{|\nu| \le N-1} D^{\nu} f(x) \frac{h^{\nu}}{|\nu|!} + R_{N-1}(x,h) \frac{|h|^{N-1}}{(N-1)!},$$

where  $|R_{N-1}(x,h)| \leq C \sup_{\substack{|l| \leq |h| \ |\nu| = N-1}} \|\Delta_l^1 D^{\nu} f\|_{L^{\infty}}$ . Proposition 92 leads to the conclusion. 

The converse result is a consequence of theorem 88.

#### A characterization of spaces $\Lambda^{\sigma,\alpha}(\mathbb{R}^d)$ in terms of Littlewood-2.8Paley decomposition

The goal of this section is to prove a characterization of the generalized Hölder spaces in terms of the Littlewood-Paley decomposition. The results obtained in this section are a particular case of theorem 33. However, since they can easily be obtained from the previous results, they are presented here in order to have a consistent theory without using results from [97]. Moreover, they allow a better understanding of the hidden mechanisms behind the theory.

Let us recall the definition of Littlewood-Paley decomposition. Let  $\hat{\varphi}$  be a function of the Schwartz space such that

$$\hat{\varphi}(\xi) = 1$$
 if  $|\xi| \le \frac{1}{2}$   
 $\hat{\varphi}(\xi) = 0$  if  $|\xi| \ge 1$ 

and

$$\hat{\psi}(\xi) := \hat{\varphi}(\xi/2) - \hat{\varphi}(\xi).$$

We set  $S_j(f) = \mathcal{F}^{-1}(\hat{\varphi}(2^{-j}\xi)\mathcal{F}f)$  and  $\Delta_j(f) := S_{j+1}(f) - S_j(f) = \mathcal{F}^{-1}(\hat{\psi}(2^{-j}\xi)\mathcal{F}f)$  for all  $f \in \mathcal{S}'(\mathbb{R}^d)$ . Such a definition is motivated by the Bernstein inequalities (see appendix). It gives the following Littlewood-Paley decomposition<sup>2</sup>: we have

$$Id = S_0 + \Delta_0 + \Delta_1 + \dots,$$

with convergence in  $\mathcal{S}'(\mathbb{R}^d)$ .

Since  $\psi(x) = 2^d \varphi(2x) - \varphi(x)$   $(x \in \mathbb{R}^d)$ , the function  $\psi$  has a vanishing integral. If the function f belongs to the space  $L^p$  for some  $p \in [1,\infty]$ , the functions  $S_i f$  and  $\Delta_i f$  can be interpreted as a convolution product between f and a regular function that belongs to the Schwartz space. Indeed, one can prove<sup>3</sup> that functions  $S_i(f)$  and  $\Delta_i(f)$  belong to the space  $L^p(\mathbb{R}^d)$  and that

$$S_j(f) = 2^{jd} \varphi(2^j \cdot) \star f$$
 and  $\Delta_j(f) = 2^{jd} \psi(2^j \cdot) \star f$ 

for all  $j \in \mathbb{Z}$ .

 $<sup>^{2}</sup>$ A proof of this result is given by proposition 191 in the appendix.

<sup>&</sup>lt;sup>3</sup>See e.g. lemma 192 for a proof.

**Remark 106.** The support of the Fourier transform of  $S_j(f)$  is included in the ball  $B(0, \leq 2^j)$  and the support of the Fourier transform of  $\Delta_j f$  is included in the annulus  $B(0, \leq 2^{j+1}) \setminus B(0, < 2^{j-1})$ . As a consequence, we have the following result: if  $f \in L^p(\mathbb{R}^d)$   $(p \in [1, +\infty])$  then the functions  $S_j(f)$  and  $\Delta_j(f)$  belong to  $C^{\infty}(\mathbb{R}^d)$  and satisfy

$$||D^{\alpha}S_{j}(f)||_{L^{p}} \leq 2^{j|\alpha|}||S_{j}(f)||_{L^{p}}$$
 and  $||D^{\alpha}\Delta_{j}(f)||_{L^{p}} \leq 2^{(j+1)|\alpha|}||\Delta_{j}f||_{L^{p}} \quad \forall \alpha \in \mathbb{N}_{0}^{d}$ .

This is a consequence of classical Bernstein's inequalities (see e.g. [93], p.32).

**Remark 107.** The function  $\varphi$  satisfies  $\int_{\mathbb{R}^d} \varphi(x) dx > 0$  and  $\int_{\mathbb{R}^d} x^{\alpha} \varphi(x) dx = 0$  for all  $\alpha \in \mathbb{N}_0^d \setminus \{0\}$ . Indeed, as  $D^{\alpha} \mathcal{F} \varphi(0) = 0$  for  $\alpha \neq 0$ , we have

$$0 = \mathcal{F}(x^{\alpha}\varphi(x))(0) = \int_{\mathbb{R}^d} x^{\alpha}\varphi(x)dx.$$

As  $\psi(x) = 2^d \varphi(2x) - \varphi(x)$ , the moments of  $\psi$  are vanishing, i.e.

$$\int_{\mathbb{R}^d} x^{\alpha} \psi(x) dx = 0 \quad \forall \alpha \in \mathbb{N}_0^d$$

For more information about the Littlewood-Paley decomposition, the reader can refer to [84].

The proof of the following result is essentially inspired from proposition  $1.1.^4$  in [65].

**Proposition 108.** Let  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$  be a strong admissible sequence of order  $N \in \mathbb{N}^*$  and  $f \in L^{\infty}(\mathbb{R}^d)$ . If

$$\|\Delta_j f\|_{L^{\infty}} \le C\sigma_j \quad \forall j \in \mathbb{N}_0$$

then  $f \in \Lambda^{\sigma,\overline{s}(\sigma^{-1})}(\mathbb{R}^d)$ .

*Proof.* By the Bernstein inequalities, we have  $\Delta_j f \in C^{\infty}(\mathbb{R}^d)$  and

$$\|D^{\alpha}\Delta_{j}f\|_{L^{\infty}} \leq C2^{|\alpha|j}\sigma_{j}, \quad \forall |\alpha| \leq N, \forall j \in \mathbb{N}_{0}.$$

So,  $S_0 f + \sum_{j=0}^{+\infty} \Delta_j f$  converges uniformly on  $\mathbb{R}^d$ . Let  $x_0 \in \mathbb{R}^d$ ; we set

$$P_j(x - x_0) = \sum_{|\alpha| \le N-1} \frac{(x - x_0)^{\alpha}}{|\alpha|!} D^{\alpha} \Delta_j f(x_0) \quad \forall j \in \mathbb{N}_0 \cup \{-1\}$$
(2.9)

(where  $\Delta_{-1} = S_0$ ) and

$$P(x - x_0) = \sum_{j \ge -1} P_j(x - x_0).$$
(2.10)

Obviously, the assumption on  $\sigma$  implies that those polynomials are well-defined and of a degree less or equal to N-1. Let  $x \in \mathbb{R}^d$  and  $j_0 \in \mathbb{N}^*$  such that

$$2^{-j_0} \le |x - x_0| < 2^{-j_0 + 1}.$$

 $<sup>^{4}</sup>$ The present proof also corrects some small mistakes found in [65], such as the introduction of polynomials in the integrals.

We have

$$|f(x) - P(x - x_0)| \le \sum_{j \le j_0} |\Delta_j f(x) - P_j(x - x_0)| + \sum_{j > j_0} |\Delta_j f(x) - P_j(x - x_0)|.$$

The Taylor formula gives a bound for the first term:

$$\sum_{j \le j_0} |x - x_0|^N \sup_{|\alpha| = N} \|D^{\alpha} \Delta_j f\|_{L^{\infty}}$$
$$\le C 2^{-Nj_0} \sum_{j \le j_0} 2^{Nj} \sigma_j$$
$$\le C \sigma_{j_0}.$$

The second term is bounded by

$$\sum_{j>j_0} \left( \|\Delta_j f\|_{L^{\infty}} + \sum_{|\alpha| \le N-1} |x - x_0|^{|\alpha|} \|D^{\alpha} \Delta_j f\|_{L^{\infty}} \right)$$
  
$$\leq \sum_{j>j_0} C \left( \sigma_j + \sum_{|\alpha| \le N-1} |x - x_0|^{|\alpha|} 2^{|\alpha|j} \sigma_j \right)$$
  
$$\leq C 2^{-(N-1)j_0} \sum_{j>j_0} 2^{(N-1)j} \sigma_j$$
  
$$\leq C \sigma_{j_0}.$$

The previous result can be improved significantly: the next result shows that we can get rid of the assumption about the strong admissible sequence.

**Proposition 109.** Let  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$  be an admissible sequence such that  $\underline{s}(\sigma^{-1}) > 0$  and  $f \in L^{\infty}(\mathbb{R}^d)$ . If

$$\|\Delta_j f\|_{L^{\infty}} \le C\sigma_j \quad \forall j \in \mathbb{N}_0$$

then  $f \in \Lambda^{\sigma,\overline{s}(\sigma^{-1})}(\mathbb{R}^d)$ .

*Proof.* Let  $M > \overline{s}(\sigma^{-1})$  be a natural number. We know that  $\underline{s}(\sigma^{-1}) > 0$ , so that

$$f = S_0(f) + \sum_{j \in \mathbb{N}_0} \Delta_j f$$
 uniformly on  $\mathbb{R}^d$ .

Let  $x_0 \in \mathbb{R}^d$  and  $J \in \mathbb{N}_0$ ; let us set

$$P_j(x - x_0) = \sum_{|\alpha| \le M - 1} \frac{(x - x_0)^{\alpha}}{|\alpha|!} D^{\alpha} \Delta_j f(x_0) \quad \forall j \in \mathbb{N}_0 \cup \{-1\}$$
(2.11)

with  $\Delta_{-1} = S_0$  and

$$P_{x_0,J}(x-x_0) = \sum_{j=-1}^{J} P_j(x-x_0).$$
(2.12)

It is a polynomial of degree less or equal to M-1. Let  $x \in \mathbb{R}^d$  be such that  $|x-x_0| \leq 2^{-J}$ . We have

$$|f(x) - P_{x_0,J}(x - x_0)| \le \left| \sum_{j=-1}^{J} \left( \Delta_j f(x) - \sum_{|\alpha| \le M-1} \frac{(x - x_0)^{\alpha}}{|\alpha|!} D^{\alpha} \Delta_j f(x_0) \right) \right| + \left| \sum_{j=J+1}^{+\infty} \Delta_j f(x) \right|.$$

Since  $\underline{s}(\sigma^{-1}) > 0$ , the second term is bounded by  $C\sigma_J$ . By the Taylor formula, the first term is bounded by

$$\sum_{j=-1}^{J} |x - x_0|^M \sup_{|\alpha| = M} \|D^{\alpha} \Delta_j f\|_{L^{\infty}}$$
$$\leq C 2^{-JM} \sum_{j=-1}^{J} 2^{jM} \sigma_j$$
$$\leq C \sigma_J,$$

thanks to lemma 53, where the constant C is independent of x and J. Theorem 88 allows to conclude.  $\Box$ 

**Remark 110.** The previous proof gives interesting information about the spaces  $\Lambda^{\sigma,\overline{s}(\sigma^{-1})}(\mathbb{R}^d)$ . Indeed, equalities (2.11) and (2.12) give the polynomials that can be used in corollary 91. They are strongly linked to the Littlewood-Paley decomposition. Moreover, when the sequence  $\sigma$  is strong of order  $N \in \mathbb{N}^*$ , the polynomial given by (2.9) and (2.10) is independent of the scale  $j \in \mathbb{N}_0$ .

The following result essentially uses ideas exposed in [65].

**Proposition 111.** Let  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$  be a strong admissible sequence of order  $N \in \mathbb{N}^*$  and  $f \in L^{\infty}(\mathbb{R}^d)$ . If  $f \in \Lambda^{\sigma,\overline{s}(\sigma^{-1})}(\mathbb{R}^d)$  then

$$\|\Delta_j f\|_{L^{\infty}} \le C\sigma_j \quad \forall j \in \mathbb{N}_0.$$

Proof. By corollary 105 and remark 107, we have

$$\begin{aligned} |\Delta_{j}f(x)| &= \left| \int_{\mathbb{R}^{d}} f(x+t) 2^{jd} \psi(2^{j}t) dt \right| \\ &\leq \int_{\mathbb{R}^{d}} |R_{N-1}(x,t)| \frac{|t|^{N-1}}{(N-1)!} 2^{jd} |\psi(2^{j}t)| dt \\ &\leq C \int_{\mathbb{R}^{d}} \sup_{\substack{x,|l| \leq |t| \\ |\nu| = N-1}} \|\Delta_{l}^{1} D^{\nu} f\|_{L^{\infty}} \frac{|t|^{N-1}}{(N-1)!} 2^{jd} |\psi(2^{j}t)| dt \\ &\leq C \int_{\mathbb{R}^{d}} \sup_{\substack{x,|l| \leq |t|/2^{j} \\ |\nu| = N-1}} \|\Delta_{l}^{1} D^{\nu} f\|_{L^{\infty}} 2^{-j(N-1)} |t|^{N-1} |\psi(t)| dt. \end{aligned}$$

One has

$$\int_{\substack{|t| \le 1 \\ |\nu| = N-1}} \sup_{\substack{x, |l| \le |t|/2^j \\ |\nu| = N-1}} \|\Delta_l^1 D^{\nu} f\|_{L^{\infty}} 2^{-j(N-1)} |t|^{N-1} |\psi(t)| dt \le C\sigma_j$$

and, for  $m \in \mathbb{N}_0$ ,

$$\int_{2^{m} \le |t| \le 2^{m+1}} \sup_{\substack{x, |l| \le |t|/2^{j} \\ |\nu| = N-1}} \|\Delta_{l}^{1} D^{\nu} f\|_{L^{\infty}} 2^{-j(N-1)} |t|^{N-1} |\psi(t)| dt$$

$$\leq C_{M} \int_{2^{m} \le |t| \le 2^{m+1}} \sup_{\substack{x, |l| \le |t|/2^{j} \\ |\nu| = N-1}} \|\Delta_{l}^{1} D^{\nu} f\|_{L^{\infty}} 2^{-j(N-1)} |t|^{N-1} \frac{1}{(1+|t|)^{M}} dt$$
(where  $M \in \mathbb{N}^{*}$  can be chosen arbitrarily large)

$$\leq C_M \int_{\frac{1}{2} \leq |t| \leq 1} \sup_{\substack{|l| \leq 2^{m+1-j} \\ |\nu| = N-1}} \|\Delta_l^1 D^{\nu} f\|_{L^{\infty}} 2^{m(N+d-1)} 2^{-j(N-1)} 2^{-mM} dt$$

$$\leq C_M 2^{m(N+d)} 2^{-j(N-1)} 2^{-mM} \int_{\frac{1}{2} \leq |t| \leq 1} \sup_{\substack{|l| \leq 2^{-j} \\ |\nu| = N-1}} \|\Delta_l^1 D^{\nu} f\|_{L^{\infty}} dt$$

$$\leq C_M 2^{m(N+d-M)} \sigma_j$$

(where a classical property of the finite differences has been applied in the last but one line, see lemma 67). Putting these inequalities together gives the desired result.  $\Box$ 

The previous proof relies on the Taylor expansion of elements of generalized Hölder spaces. We now give another proof slightly less intuitive and more technical. It leads to a more general result.

**Proposition 112.** Let  $N \in \mathbb{N}_0$  and  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$  be an admissible sequence such that

$$N < \underline{s}(\sigma^{-1}) \le \overline{s}(\sigma^{-1}) < N + 2.$$

If  $f \in \Lambda^{\sigma,\overline{s}(\sigma^{-1})}(\mathbb{R}^d)$  then

$$\|\Delta_j f\|_{L^{\infty}} \le C\sigma_j \quad \forall j \in \mathbb{N}_0$$

*Proof.* We can suppose that the function  $\varphi$  defining the Littlewood-Paley decomposition is an even function. Indeed, one can easily show that this result does not depend on the considered function  $\varphi$  (one can use the same proof as for the independence of the choice of the unit partition for the classical Besov spaces, see [118] or [50]). As a consequence, the function  $\psi$  is also even. In proposition 92, we have  $f \in C^N(\mathbb{R}^d)$  and

$$D^{\alpha}f \in L^{\infty}(\mathbb{R}^d)$$
 and  $\sup_{|h| \le 2^{-j}} \|\Delta_h^2 D^{\alpha}f\|_{L^{\infty}} \le C\sigma_j 2^{jN} \quad \forall |\alpha| \le N.$ 

We note that

$$\Delta_j D_{y_k} f(x) = \mathcal{F}^{-1}(iy_k \hat{\psi}(2^{-j}\xi) \mathcal{F} f)(x)$$
$$= D_{y_k} \Delta_j f(x)$$

for all  $k \in \{1, ..., d\}$ . By induction, we find

$$\Delta_j D^{\alpha} f(x) = D^{\alpha} \Delta_j f(x) \quad \forall |\alpha| \le N.$$

Then, we have

$$\begin{split} \Delta_j D^{\alpha} f(x) &= 2^{jd} \int_{\mathbb{R}^d} D^{\alpha} f(x-y) \psi(2^j y) dy \\ &= \frac{2^{jd}}{2} \int_{\mathbb{R}^d} (D^{\alpha} f(x+y) - 2D^{\alpha} f(x) + D^{\alpha} f(x-y)) \psi(2^j y) dy \end{split}$$

because  $\psi$  is even with vanishing integrals. One of the Bernstein's inequalities states that

$$\|\Delta_j f\|_{L^{\infty}} \le C 2^{-jN} \sup_{|\alpha|=N} \|D^{\alpha} \Delta_j f\|_{L^{\infty}}$$

(see e.g. [84], prop.3.2 p.24), so that

$$\begin{split} \|\Delta_{j}f\|_{L^{\infty}} &\leq C2^{-jN}2^{jd} \int_{\mathbb{R}^{d}} \sup_{|\alpha|=N} \|\Delta_{y}^{2}D^{\alpha}f\|_{L^{\infty}} |\psi(2^{j}y)| dy \\ &\leq C2^{-jN} \int_{\mathbb{R}^{d}} \sup_{|\alpha|=N} \|\Delta_{2^{-j}y}^{2}D^{\alpha}f\|_{L^{\infty}} |\psi(y)| dy. \end{split}$$

On one hand, we have

$$\int_{|y|\leq 1} \sup_{|\alpha|=N} \|\Delta_{2^{-j}y}^2 D^{\alpha} f\|_{L^{\infty}} |\psi(y)| dy \leq C\sigma_j 2^{jN}$$

and, on the other hand, we get

$$\begin{split} \int_{2^{m} \leq |y| \leq 2^{m+1}} \sup_{|\alpha| = N} \|\Delta_{2^{-j}y}^{2} D^{\alpha} f\|_{L^{\infty}} |\psi(y)| dy \\ &\leq \int_{2^{m} \leq |y| \leq 2^{m+1}} \sup_{\substack{|\alpha| = N \\ |h| \leq 2^{-j}}} \|\Delta_{2^{m+1}h}^{2} D^{\alpha} f\|_{L^{\infty}} |\psi(y)| dy \\ &\leq C 2^{2(m+1)} \int_{2^{m} \leq |y| \leq 2^{m+1}} \sup_{\substack{|\alpha| = N \\ |h| \leq 2^{-j}}} \|\Delta_{h}^{2} D^{\alpha} f\|_{L^{\infty}} |\psi(y)| dy \\ &\qquad (\text{lemma 67}) \\ &\leq C_{M} 2^{2m} 2^{Nj} \sigma_{j} \int_{2^{m} \leq |y| \leq 2^{m+1}} \frac{1}{(1+|y|)^{M}} dy \\ &\leq C_{M} 2^{2m} 2^{Nj} \sigma_{j} 2^{md} 2^{-mM} \end{split}$$

for  $M \in \mathbb{N}_0$  sufficiently large. Hence the conclusion follows by taking e.g. M = d + 3.  $\Box$ In particular, we have obtained the following results:

**Corollary 113.** Let  $N \in \mathbb{N}_0$  and let  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$  be an admissible sequence such that

$$N < \underline{s}(\sigma^{-1}) \le \overline{s}(\sigma^{-1}) < N + 2.$$

Let  $f \in L^{\infty}(\mathbb{R}^d)$ ; the following assertions are equivalent:

- 1.  $f \in \Lambda^{\sigma,\overline{s}(\sigma^{-1})}(\mathbb{R}^d);$
- 2.  $\exists C > 0 : \|\Delta_j f\|_{L^{\infty}} \le C\sigma_j \quad \forall j \in \mathbb{N}_0.$

**Corollary 114.** Let  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$  be a strong admissible sequence of order  $N \in \mathbb{N}^*$  and  $f \in L^{\infty}(\mathbb{R}^d)$ . The following assertions are equivalent:

- 1.  $f \in \Lambda^{\sigma,\overline{s}(\sigma^{-1})}(\mathbb{R}^d);$
- 2.  $\exists C > 0 : \|\Delta_j f\|_{L^{\infty}} \le C\sigma_j \quad \forall j \in \mathbb{N}_0.$

# 2.9 A characterization of spaces $\Lambda^{\sigma,\alpha}(\mathbb{R}^d)$ in terms of wavelet coefficients

The goal of this section is to generalize theorem 75. Reminders about wavelets are done in section 5.4.

If we keep the same notations as the ones introduced in section 2.8, we note that  $\Delta_j f(x) = 2^{jd} \int_{\mathbb{R}^d} f(t) \psi(2^j(x-t)) dt$ . So,

$$\Delta_j f(k2^{-j}) = 2^{jd} \int_{\mathbb{R}^d} f(t)\psi(k-2^jt)dt$$

for all  $k \in \mathbb{Z}^d$  and  $j \in \mathbb{N}_0$ . One can therefore conclude that the Littlewood-Paley decomposition seems to be very similar to a wavelet decomposition. Indeed, the wavelet coefficients of f somehow looks like a discretization of the functions  $\Delta_j(f)$  at points  $k2^{-j}$ . Since Hölder spaces can be characterized in terms of the Littlewood-Paley decomposition, it is natural to ask whether those spaces admit a wavelet characterization as well.

The aim of this section is to obtain a wavelet characterization of the generalized Hölder-Zygmund spaces. For that purpose, we follow some ideas expressed in [64, 93]. We consider the Lemarié-Meyer wavelets (where the functions  $\phi$  and  $\psi^i$  belong to the Schwartz space, [83]) and the Daubechies wavelets (where the functions  $\phi$  and  $\psi^i$  are compactly supported and can be taken arbitrarily regular<sup>5</sup>, [37]).

### **Theorem 115.** (D.K., S. Nicolay) Let $\sigma$ be an admissible sequence such that $\underline{s}(\sigma^{-1}) > 0$ .

1. Let us consider the Daubechies wavelets. If  $f \in \Lambda^{\sigma,\overline{s}(\sigma^{-1})}(\mathbb{R}^d)$ , then there exists C > 0 such that

$$\begin{cases} |C_k| \le C & \forall k \in \mathbb{Z}^d \\ |c_{j,k}^i| \le C\sigma_j & \forall j \in \mathbb{N}_0, \forall i \in \{1, \dots, 2^d - 1\}, \forall k \in \mathbb{Z}^d. \end{cases}$$
(2.13)

If the assumption  $\underline{s}(\sigma^{-1}) > 0$  is replaced by  $\sigma$  is a strong admissible sequence of order  $N \in \mathbb{N}^*$ , then this result holds for the Lemarié-Meyer wavelets.

2. Conversely, if  $f \in L^{\infty}_{loc}(\mathbb{R}^d)$  and (2.13) holds, then  $f \in \Lambda^{\sigma,\overline{s}(\sigma^{-1})}(\mathbb{R}^d)$ .

*Proof.* Let  $f \in \Lambda^{\sigma,\overline{s}(\sigma^{-1})}(\mathbb{R}^d)$  and let us prove that (2.13) holds. Let us consider the Daubechies wavelets. Let  $M \in \mathbb{N}_0$  and  $j_0 \in \mathbb{N}_0$  such that  $M > \overline{s}(\sigma^{-1})$  and

$$supp \psi^{i} \subseteq B(0, \leq 2^{j_{0}}) \quad \forall i \in \{1, ..., 2^{d} - 1\}.$$

We have

$$|C_k| = \left| \int_{\mathbb{R}^d} f(x)\phi(x-k)dx \right| \le C ||f||_{L^{\infty}}.$$

<sup>&</sup>lt;sup>5</sup>Using the notation of the following, we consider that the multiresolution analysis is at least of regularity  $r > \overline{s}(\sigma^{-1})$ .

Using corollary 91, for  $k \in \mathbb{Z}^d$  and  $j \geq j_0$ , let  $P_{k/2^j, j-j_0}$  be a polynomial of degree less or equal to M-1 such that

$$\|f(\cdot) - P_{k/2^{j}, j-j_{0}}(\cdot - k/2^{j})\|_{L^{\infty}(B(k/2^{j}, 2^{-(j-j_{0})}))} \leq C\sigma_{j-j_{0}}.$$

One gets

$$\begin{aligned} |c_{j,k}^{i}| &= 2^{jd} \left| \int_{\mathbb{R}^{d}} f(x)\psi^{i}(2^{j}x-k)dx \right| \\ &= 2^{jd} \left| \int_{B\left(k/2^{j},2^{-(j-j_{0})}\right)} \left(f(x) - P_{k/2^{j},j-j_{0}}(x-k/2^{j})\right)\psi^{i}(2^{j}x-k)dx \right| \\ &\leq C\sigma_{j} \sup_{i\in\{1,\dots,2^{d}-1\}} \|\psi^{i}\|_{L^{1}(\mathbb{R}^{d})}. \end{aligned}$$

Let us consider the Lemarié-Meyer wavelets and let  $\sigma$  be a strong admissible sequence of order  $N \in \mathbb{N}^*$ . We have  $|C_k| \leq C ||f||_{L^{\infty}}$  and

$$\begin{aligned} |c_{j,k}^{i}| &= 2^{jd} \left| \int_{\mathbb{R}^{d}} f\left(\frac{k}{2^{j}} + (x - \frac{k}{2^{j}})\right) \psi^{i}(2^{j}x - k) dx \right| \\ &= 2^{jd} \int_{\mathbb{R}^{d}} |R_{N-1}\left(\frac{k}{2^{j}}, x - \frac{k}{2^{j}}\right)| \frac{|x - \frac{k}{2^{j}}|^{N-1}}{(N-1)!} |\psi^{i}(2^{j}x - k)| dx \\ &= 2^{jd} \int_{\mathbb{R}^{d}} |R_{N-1}\left(\frac{k}{2^{j}}, y\right)| \frac{|y|^{N-1}}{(N-1)!} |\psi^{i}(2^{j}y)| dy \\ &\leq C \int_{\mathbb{R}^{d}} \sup_{\substack{|h| \leq |y|/2^{j} \\ |\alpha| = N-1}} \|\Delta_{h}^{1} D^{\alpha} f\|_{L^{\infty}} |y|^{N-1} 2^{-j(N-1)} |\psi^{i}(y)| dy \end{aligned}$$

by using the same notations as the ones used in corollary 105. A similar proof as the one used in proposition 111 allows to conclude.

Now, suppose that we have

$$|C_k| \le C$$
 and  $|c_{j,k}^i| \le C\sigma_j$   $\forall j \in \mathbb{N}_0, \forall i \in \{1, \dots, 2^d - 1\}, \forall k \in \mathbb{Z}^d.$ 

We need to check that conditions of proposition 92 are satisfied. For that purpose, we will use some of the ideas exposed in [64]. Let  $N, M \in \mathbb{N}_0$  such that

$$N < \underline{s}(\sigma^{-1}) \le \overline{s}(\sigma^{-1}) < M \le r.$$

Let us denote

$$f_{-1}(x) = \sum_{k \in \mathbb{Z}^d} C_k \phi(x-k)$$

 $\quad \text{and} \quad$ 

$$f_j(x) = \sum_{i=1}^{2^d - 1} \sum_{k \in \mathbb{Z}^d} c_{j,k}^i \psi^i (2^j x - k)$$

for all  $j \in \mathbb{N}_0$ . This last series converges uniformly on every compact set, because of the wavelet coefficient assumption and because of the decay of  $\phi$  and  $\psi$ . Indeed, one has

$$\sum_{i=1}^{2^{d}-1} \sum_{|k| \le L} |c_{j,k}^{i} \psi^{i}(2^{j}x-k)| \le C\sigma_{j} \sum_{|k| \le L} \frac{1}{(1+|2^{j}x-k|)^{2^{d+1}}} \le C\sigma_{j} \quad (\forall L > 0),$$

where the constant C is uniform on x and j. The sequence of functions

$$\left(\sum_{i=1}^{2^{d}-1} \sum_{|k| \le L} |c_{j,k}^{i} \psi^{i}(2^{j}x-k)|\right)_{L \in \mathbb{N}_{0}}$$

is a sequence of continuous functions on  $\mathbb{R}^d$ , increasing and converging to a continuous and bounded function on  $\mathbb{R}^d$  (for the Daubechies wavelets, the continuity comes from the fact that the series can be reduced to a finite sum on each ball of  $\mathbb{R}^d$ ; for the Meyer wavelets, we can check the continuity by using the convergence of the sequences and their rapidly decreasing properties). By applying a classical theorem of Dini, this series converges uniformly on every compact set of  $\mathbb{R}^d$ . So, the limit  $f_j$  is well-defined and has the same regularity as the wavelets. Let us write

$$g(x) = \sum_{j=-1}^{+\infty} f_j(x).$$

For all  $j \geq -1$ , we have

 $|f_j(x)| \le C\sigma_j$ 

and the series g converges uniformly on  $\mathbb{R}^d$  to a function that belongs to  $L^{\infty}(\mathbb{R}^d)$ . So, one gets f = g. Similarly, by using the decay properties of  $D^{\beta}\psi^i$  and  $D^{\beta}\phi$ , we obtain

$$|D^{\beta}f_j(x)| \le C2^{|\beta|j}\sigma_j, \quad \forall |\beta| \le M.$$

So, we can differentiate the series  $\sum_j f_j$  term by term up to order N. This proves that  $f \in C^N(\mathbb{R}^d)$  and  $|D^{\beta}f(x)| \leq C$  for all  $|\beta| \leq N$ . Let  $\alpha \in \mathbb{N}_0^d$  such that  $|\alpha| = N$ , and let  $h \in \mathbb{R}^d$  and  $j_0 \in \mathbb{N}_0$  be such that  $|h| < 2^{-j_0}$ . We have

$$\begin{split} \|\Delta_{h}^{M-N} D^{\alpha} f\|_{L^{\infty}} \\ &\leq \sum_{j \leq j_{0}} \|\Delta_{h}^{M-N} D^{\alpha} f_{j}\|_{L^{\infty}} + \sum_{j=j_{0}+1}^{+\infty} 2^{M-N} \|D^{\alpha} f_{j}\|_{L^{\infty}} \\ &\leq C \sum_{j \leq j_{0}} |h|^{M-N} \sup_{|\beta|=M-N} \|D^{\beta+\alpha} f_{j}\|_{L^{\infty}} + C \sum_{j=j_{0}+1}^{+\infty} 2^{Nj} \sigma_{j} \\ &\leq C |h|^{M-N} 2^{Mj_{0}} \sigma_{j_{0}} + C 2^{N(j_{0}+1)} \sigma_{j_{0}+1} \\ &\leq C 2^{Nj_{0}} \sigma_{j_{0}}. \end{split}$$

# 2.10 Generalized Hölder-Zygmund spaces $\Lambda^{\sigma,\alpha}(\mathbb{R}^d)$ and generalized (real) interpolation of Sobolev spaces

Several recent results express the generalized Besov spaces as an interpolation of other spaces. For example, generalized Besov spaces can be obtained through an interpolation of classical Besov spaces (see [7]). In our framework, this means that generalized Hölder spaces can be obtained as an interpolation of classical Hölder spaces.

On the other hand, it is well known that classical Besov spaces  $B_{p,q}^s$   $(p,q \in [1, +\infty], s \in \mathbb{R})$  can be obtained through an interpolation of Sobolev spaces  $W_m^p(\mathbb{R}^d)$  ([3, 84]). A natural question to ask is whether this result is still true for generalized Hölder spaces. The goal of this section is to generalize theorem 76.

Classical concepts associated with interpolation spaces as well as the definition of Sobolev spaces are recalled in section 5.5. Let us now introduce some new concepts. In the sequel, we consider two Banach spaces  $A_0$  and  $A_1$ , which are continuously embedded in a topological vector space V. So, spaces  $A_0 \cap A_1$  and  $A_0 + A_1$  are well-defined Banach spaces. We recall that the operator J is defined for all t > 0 and  $a \in A_0 \cap A_1$  by

$$J(t,a) = \max\{\|a\|_{A_0}, t\|a\|_{A_1}\}.$$

Let us give the definition of the generalized *J*-method of interpolation.

**Definition 116.** Let  $\sigma = (\sigma_j)_{j \in \mathbb{Z}}$  and  $\psi = (\psi_j)_{j \in \mathbb{Z}}$  be two admissible sequences. We define the generalized interpolation space  $[A_0, A_1]^*_{\sigma, \psi, J}$  in the following way: we say that a belongs to  $[A_0, A_1]^*_{\sigma, \psi, J}$  if a can be written as  $a = \sum_{j \in \mathbb{Z}} u_j$  with convergence in  $A_0 + A_1$ , where  $u_j \in A_0 \cap A_1$  and  $(\sigma_j J(\psi_j, u_j))_{j \in \mathbb{Z}} \in l^{\infty}(\mathbb{Z})$ .

We recall that the operator K is defined for all t > 0 and  $a \in A_0 + A_1$  by

$$K(t,a) = \inf\{\|a_0\|_{A_0} + t\|a_1\|_{A_1} : a = a_0 + a_1\}.$$

Let us give the definition of the generalized K-method of interpolation.

**Definition 117.** Let  $\sigma = (\sigma_j)_{j \in \mathbb{Z}}$  and  $\psi = (\psi_j)_{j \in \mathbb{Z}}$  be two admissible sequences. The generalized interpolation space  $[A_0, A_1]^*_{\sigma, \psi, K}$  is defined in the following way: we say that a belongs to  $[A_0, A_1]^*_{\sigma, \psi, K}$  if  $a \in A_0 + A_1$  and  $(\sigma_j K(\psi_j, a))_{j \in \mathbb{Z}} \in l^{\infty}(\mathbb{Z})$ .

If  $\sigma_j = 2^{-j\alpha}$  and  $\psi_j = 2^j$ , one recovers the classical real interpolation spaces  $[A_0, A_1]_{\alpha, \infty, J}$ and  $[A_0, A_1]_{\alpha, \infty, K}$ .

The next result shows that the generalized J-method of interpolation and the generalized K-method of interpolation are equivalent under specific conditions.

**Proposition 118.** Let  $N, M \in \mathbb{N}_0$  and  $\sigma$  be an admissible sequence such that

$$N < \underline{s}(\sigma^{-1}) \le \overline{s}(\sigma^{-1}) < M.$$

If  $A_1$  is continuously embedded in  $A_0$ , one has

$$[A_0, A_1]^{\star}_{\theta, 2^{j(M-N)}, J} = [A_0, A_1]^{\star}_{\theta, 2^{j(M-N)}, K}$$

where  $\theta$  is the admissible sequence defined by

$$\theta_j = \begin{cases} 2^{jN} \sigma_{-j}^{-1} & \forall j \in -\mathbb{N}_0 \\ (\theta_{-j})^{-1} & \forall j \in \mathbb{N}^* . \end{cases}$$

*Proof.* Let  $f \in [A_0, A_1]^*_{\theta, 2^{j(M-N)}, J}$ . This function can be written as  $f = \sum_{j \in \mathbb{Z}} f_j$  where the series converges in  $A_0$  and where the functions  $f_j$  satisfy

$$||f_j||_{A_0} + 2^{j(M-N)} ||f_j||_{A_1} \le C\theta_j^{-1} \qquad \forall j \in \mathbb{Z}.$$

Let us set  $b_j = \sum_{l=-\infty}^{j-1} f_l$  and  $c_j = \sum_{l=j}^{+\infty} f_l$  for all  $j \in \mathbb{Z}$ . We have  $b_j \in A_0$  and  $c_j \in A_1$ . Let us prove that the inequality

$$\theta_j(\|b_j\|_{A_0} + 2^{j(M-N)}\|c_j\|_{A_1}) \le C \qquad \forall j \in \mathbb{Z}$$

holds.

1. If j < 0, then

$$\begin{split} \|b_{j}\|_{A_{0}} &\leq \sum_{l=-\infty}^{j-1} \|f_{l}\|_{A_{0}} \\ &\leq C \sum_{l=-j+1}^{+\infty} \theta_{-l}^{-1} = C \sum_{l=-j+1}^{+\infty} 2^{lN} \sigma_{l} \\ &\leq C 2^{-jN} \sigma_{-j} = C \theta_{j}^{-1}, \end{split}$$

and

$$\begin{aligned} \|c_{j}\|_{A_{1}} &\leq \sum_{l=j}^{+\infty} \|f_{l}\|_{A_{1}} \\ &\leq C \sum_{l=j}^{+\infty} \theta_{l}^{-1} 2^{-l(M-N)} \\ &\leq C \sum_{l=1}^{-j} 2^{l(M-N)} \theta_{-l}^{-1} + C \sum_{l=0}^{+\infty} 2^{-l(M-N)} \theta_{l}^{-1} \\ &\leq C 2^{-jM} \sigma_{-j} + C \leq C 2^{-j(M-N)} \theta_{j}^{-1}. \end{aligned}$$

2. If  $j \ge 0$ , then

$$\begin{split} \|b_j\|_{A_0} &\leq \sum_{l=-\infty}^0 \|f_l\|_{A_0} + C \sum_{l=1}^{j-1} \|f_l\|_{A_1} \\ &\leq C + C \sum_{l=1}^{j-1} 2^{-lM} \sigma_l^{-1} \\ &\leq C \leq C \theta_i^{-1}, \end{split}$$

 $\quad \text{and} \quad$ 

$$\begin{aligned} \|c_j\|_{A_1} &\leq \sum_{l=j}^{+\infty} \|f_l\|_{A_1} \leq C \sum_{l=j}^{+\infty} 2^{-lM} \sigma_l^{-1} \\ &\leq C 2^{-jM} \sigma_j^{-1}, \end{aligned}$$

where we have used the relation  $\overline{s}(\sigma^{-1}) < M$  in the last inequality.

Let  $f \in [A_0, A_1]^*_{\theta, 2^{j(M-N)}, K}$ ; for all  $j \in \mathbb{Z}$ , there exist  $b_j \in A_0$  and  $c_j \in A_1$  such that  $f = b_j + c_j$  and

$$||b_j||_{A_0} + 2^{j(M-N)} ||c_j||_{A_1} \le C\theta_j^{-1}.$$

Let us write  $b_0 = \sum_{j=-\infty}^{-1} (b_{j+1} - b_j)$ , with convergence in  $A_0$ . Similarly, let  $c_0 = \sum_{j=0}^{+\infty} (c_j - c_{j+1})$ , with convergence in  $A_1$ . Let us set

$$f_j = \begin{cases} b_{j+1} - b_j & \text{if } j \in -\mathbb{N}^*, \\ c_j - c_{j+1} & \text{if } j \in \mathbb{N}_0. \end{cases}$$

As  $b_{j+1} - b_j = c_j - c_{j+1}$  (for all  $j \in \mathbb{Z}$ ), we have  $f = \sum_{j \in \mathbb{Z}} f_j$  in  $A_0$ , where  $f_j \in A_1$  for all  $j \in \mathbb{Z}$ . Moreover, we have

$$||f_j||_{A_0} = ||b_{j+1} - b_j||_{A_0} \le C\theta_j^{-1}$$

and

$$||f_j||_{A_1} = ||c_{j+1} - c_j||_{A_1} \le C2^{-j(M-N)}\theta_j^{-1},$$

which leads to the conclusion.

The following theorem is the main result of this section, and show that generalized Hölder spaces can be obtained through a generalized interpolation of Sobolev spaces. It is a consequence of the characterization of generalized Hölder spaces by derivatives and by the Littlewood-Paley decomposition.

**Theorem 119.** (D.K., S. Nicolay) Let  $N, M \in \mathbb{N}_0$  and  $\sigma$  be an admissible sequence such that

$$N < \underline{s}(\sigma^{-1}) \le \overline{s}(\sigma^{-1}) < M.$$

We have

$$\Lambda^{\sigma,\overline{s}(\sigma^{-1})}(\mathbb{R}^d) = [W_N^{\infty}, W_M^{\infty}]^{\star}_{\theta,2^{j(M-N)},J} = [W_N^{\infty}, W_M^{\infty}]^{\star}_{\theta,2^{j(M-N)},K}$$

where  $\theta$  is the admissible sequence defined by

$$\theta_j = \begin{cases} 2^{jN} \sigma_{-j}^{-1} & \forall j \in -\mathbb{N}_0 \\ (\theta_{-j})^{-1} & \forall j \in \mathbb{N}^* . \end{cases}$$

*Proof.* Let us prove that we have  $\Lambda^{\sigma,\overline{s}(\sigma^{-1})}(\mathbb{R}^d) = [W_N^{\infty}, W_M^{\infty}]_{\theta,2^{j(M-N)},J}^{\star}$ . Let  $f \in \Lambda^{\sigma,\overline{s}(\sigma^{-1})}(\mathbb{R}^d)$ . We denote

$$u_j = \begin{cases} 0 & \text{if } j \in \mathbb{Z}, \ j > 1, \\ S_0(f) & \text{if } j = 1, \\ \Delta_{-j}(f) & \text{if } j \in \mathbb{Z}, \ j < 1. \end{cases}$$

By Bernstein's inequalities, the series  $\sum_{j \in \mathbb{Z}} u_j$  converges in  $W_{\infty}^N$  and we have  $u_j \in W_{\infty}^M$ . Moreover, we have  $\theta_j J(2^{j(M-N)}, u_j) \leq C$  by theorem 33.

Let  $f \in [W_N^{\infty}, W_M^{\infty}]_{\theta, 2^{j(M-N)}, J}^{\star}$ . Let us check that the conditions of proposition 92 are satisfied. Let  $(f_j)_{j \in \mathbb{Z}}$  be a sequence of functions of  $W_M^{\infty}(\mathbb{R}^d)$  such that  $\sum_{j \in \mathbb{Z}} f_j = f$  with convergence in  $W_N^{\infty}(\mathbb{R}^d)$  and such that

$$\theta_j J(2^{j(M-N)}, f_j) \in l^\infty(\mathbb{Z}).$$

By modifying the functions  $f_j$  on some negligible set, we can suppose that they belong to the space  $C^{M-1}(\mathbb{R}^d)$  (see remark 197). Let  $|\alpha| \leq N$ . We have

$$\sum_{l=0}^{+\infty} \|D^{\alpha} f_l\|_{L^{\infty}} \le C \sum_{l=0}^{+\infty} 2^{-l(M-N)} \theta_l^{-1} = C \sum_{l=0}^{+\infty} 2^{-lM} \sigma_l^{-1}$$

which is bounded because of the inequality  $\overline{s}(\sigma^{-1}) < M$ . Moreover,

$$\sum_{l=-\infty}^{-1} \|D^{\alpha} f_l\|_{L^{\infty}} \le C \sum_{l=-\infty}^{-1} \theta_l^{-1} = C \sum_{l=1}^{+\infty} 2^{lN} \sigma_l$$

Putting these inequalities together, we find that  $f \in C^N(\mathbb{R}^d)$  and  $D^{\alpha}f \in L^{\infty} \forall |\alpha| \leq N$ . Let  $h \in \mathbb{R}^d$  be such that  $|h| \leq 2^{-j}$  and  $|\alpha| = N$ . We note that

$$\Delta_h^{M-N} D^{\alpha} f = \sum_{l \in \mathbb{Z}} \Delta_h^{M-N} D^{\alpha} f_l \qquad \text{(uniformly)}.$$

By successive applications of the mean value theorem and by proposition 198, we have

$$\sum_{l=0}^{+\infty} \|\Delta_h^{M-N} D^{\alpha} f_l\|_{L^{\infty}} \le C|h|^{M-N} \sum_{l=0}^{+\infty} \|f_l\|_{W_M^{\infty}}$$
$$\le C 2^{-j(M-N)} \sum_{l=0}^{+\infty} 2^{-l(M-N)} \theta_l^{-1}$$
$$\le C 2^{-j(M-N)} \le C 2^{Nj} \sigma_j,$$

and

$$\sum_{l=-\infty}^{-1} \|\Delta_{h}^{M-N} D^{\alpha} f_{l}\|_{L^{\infty}} = \sum_{l=-j}^{-1} \|\Delta_{h}^{M-N} D^{\alpha} f_{l}\|_{L^{\infty}} + \sum_{l=-\infty}^{-j-1} \|\Delta_{h}^{M-N} D^{\alpha} f_{l}\|_{L^{\infty}}$$

$$\leq C|h|^{M-N} \sum_{l=-j}^{-1} \|f_{l}\|_{W_{M}^{\infty}} + C \sum_{l=-\infty}^{-j-1} \|f_{l}\|_{W_{N}^{\infty}}$$

$$\leq C2^{-j(M-N)} \sum_{l=-j}^{-1} 2^{-l(M-N)} \theta_{l}^{-1} + C \sum_{l=-\infty}^{-j-1} \theta_{l}^{-1}$$

$$\leq C2^{-j(M-N)} \sum_{l=1}^{j} 2^{lM} \sigma_{l} + C \sum_{l=j+1}^{+\infty} 2^{lN} \sigma_{l}$$

$$\leq C2^{jN} \sigma_{j}.$$

One can conclude, since we have  $\sup_{|h| \leq 2^{-j}} \|\Delta_h^{M-N} D^{\alpha} f\|_{L^{\infty}} \leq C 2^{Nj} \sigma_j$ .

In particular, the previous result can be applied to strong admissible sequences. The result can be restated as follows:

**Corollary 120.** Let  $\sigma$  be a strong admissible sequence of order  $N \in \mathbb{N}^*$  such that  $\overline{s}(\sigma^{-1}) < N$ . We have

$$\Lambda^{\sigma,\bar{s}(\sigma^{-1})}(\mathbb{R}^d) = [W_{N-1}^{\infty}, W_N^{\infty}]_{\theta,2^j,J}^{\star} = [W_{N-1}^{\infty}, W_N^{\infty}]_{\theta,2^j,K}^{\star}$$

where  $\theta$  is the admissible sequence defined by

$$\theta_j = \begin{cases} 2^{j(N-1)} \sigma_{-j}^{-1} & \forall j \in -\mathbb{N}_0, \\ (\theta_{-j})^{-1} & \forall j \in \mathbb{N}^*. \end{cases}$$

In this last result, the assumption  $N - 1 < \underline{s}(\sigma^{-1})$  is not necessary: inequality (2.7) is sufficient.

**Remark 121.** In particular, theorem 119 can be applied to classical Hölder spaces to obtain theorem 76. For  $\alpha > 0$ , let N and M be two natural numbers satisfying  $N < \alpha < M$ . We have

$$\Lambda^{\alpha}(\mathbb{R}^d) = [W_N^{\infty}, W_M^{\infty}]^{\star}_{\theta, 2^{j(M-N)}, J} = [W_N^{\infty}, W_M^{\infty}]^{\star}_{\theta, 2^{j(M-N)}, K}$$

where

$$\theta_j = 2^{-\alpha j} 2^{Nj} \qquad \forall j \in \mathbb{Z}$$

The following result, which is easy to prove, is taken from [84] (Prop.2.1.(A)).

**Lemma 122.** Let  $\lambda \in ]0,1[$ . For all  $\rho > 1$ , we have

$$[A_0, A_1]^{\star}_{2^{-j\lambda}, 2^j, J} = [A_0, A_1]^{\star}_{\rho^{-j\lambda}, \rho^j, J}.$$

So, if  $\lambda$  satisfies  $\alpha = (1 - \lambda)N + \lambda M$ , then

$$\Lambda^{\alpha}(\mathbb{R}^d) = [W_N^{\infty}, W_M^{\infty}]_{2^{-j\lambda}, 2^j, J}^{\star} = [W_N^{\infty}, W_M^{\infty}]_{2^{-j\lambda}, 2^j, K}^{\star}.$$

### 2.11 A weak result of Lions-Peetre type for spaces $\Lambda^{\sigma,\alpha}(\mathbb{R}^d)$

The results of this section are based on the idea of expressing a given function belonging to some Hölder-Zygmund space as a weighted sum of a more regular and a less regular function. This idea is an important concept from the real variables Lions-Peetre method of interpolation theory. Proof of the next result can be found in [73].

**Theorem 123.** Let  $0 < \alpha_1 < \alpha < \alpha_2 < +\infty$ . Then,  $f \in \Lambda^{\alpha}(\mathbb{R})$  if and only if there exists C > 0 such that for each  $0 < \lambda < 1$ , there exist  $F_1^{\lambda} \in \Lambda^{\alpha_1}(\mathbb{R})$ ,  $F_2^{\lambda} \in \Lambda^{\alpha_2}(\mathbb{R})$  satisfying  $f = F_1^{\lambda} + F_2^{\lambda}$  and

$$\begin{aligned} \|F_1^\lambda\|_{\Lambda^{\alpha_1}(\mathbb{R})} &\leq C\lambda^{\alpha-\alpha_1} \\ \|F_2^\lambda\|_{\Lambda^{\alpha_2}(\mathbb{R})} &\leq C\lambda^{\alpha-\alpha_2}. \end{aligned}$$

The aim of this section is to look how theorem 123 can be transposed in the general setting. Indeed, we have the following partial result:

**Proposition 124.** Let  $m \in \mathbb{N}^* \setminus \{1\}$ ,  $\alpha > 0$  satisfying  $1 \leq \alpha \leq m$ ,  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$  be an admissible sequence and  $f \in L^{\infty}(\mathbb{R})$  be a function such that

$$\sup_{|h| \le 2^{-j}} \|\Delta_h^m f\|_{L^{\infty}} \le C\sigma_j \quad \forall j \in \mathbb{N}_0.$$

If

$$\sum_{j=1}^{+\infty} 2^{j(m-\alpha)} \sigma_j < +\infty,$$

then, for all  $\lambda \in ]0,1[$ , there exist two functions  $F_1^{\lambda} \in \Lambda^{m-(\alpha-1)}(\mathbb{R}), F_2^{\lambda} \in \Lambda^{m-\alpha}(\mathbb{R})$  such that  $f = F_1^{\lambda} + F_2^{\lambda}$  and for  $K_{\lambda} = \lfloor 2\log_2(1/\lambda) \rfloor + 1$ , we have

$$\sup_{|h| \le 2^{-l}} \|\Delta_h^m F_2^{\lambda}\|_{L^{\infty}} \le C_1 2^{-l(m-\alpha)} \sum_{j=K_{\lambda}+1}^{+\infty} 2^{j(m-\alpha)} \sigma_j$$

and

$$\sup_{|h| \le 2^{-l}} \|\Delta_h^m F_1^\lambda\|_{L^{\infty}} \le C_2 2^{-l(m-\alpha+1)} \sum_{j=1}^{K_\lambda} 2^{j(m-\alpha+1)} \sigma_j$$

where  $C_1$  and  $C_2$  are two constants independent of  $\lambda$ .

Proof. Let  $\Phi$  be the function defined by proposition 77 and  $f_1 := f \star \Phi_{2^{-1}}, f_j := f \star (\Phi_{2^{-j}} - \Phi_{2^{-j+1}})$  (j > 1). By proposition 77, we have  $||f_j||_{L^{\infty}} \leq C\sigma_j$  for all  $j \in \mathbb{N}^*$ , where the constant C does not depend on j. We thus get

$$\sum_{j=1}^k ||f_j||_{L^{\infty}} \le C \sum_{j=1}^k \sigma_j,$$

for all  $k \in \mathbb{N}^*$ , which implies  $f = \sum_{j=1}^{+\infty} f_j$  (with uniform convergence). By the mean value theorem and lemma 78, one has

$$|\Delta_h^m f_j(x)| \le C|h|^m ||D^m f_j||_{L^{\infty}} \le C|h|^m 2^{mj} \sigma_j$$

and

$$|\Delta_h^m f_j(x)| \le 2^m ||f_j||_{L^{\infty}} \le 2^m C \sigma_j,$$

for all  $j \in \mathbb{N}^*$ . This implies

$$\begin{aligned} |\Delta_h^m f_j(x)| &= |\Delta_h^m f_j(x)|^{1-\alpha/m} |\Delta_h^m f_j(x)|^{\alpha/m} \\ &\leq C |h|^{m-\alpha} 2^{j(m-\alpha)} \sigma_j \end{aligned}$$

for all  $0 \leq \alpha \leq m$ .

Let  $\lambda \in ]0,1[$  and  $M \in \mathbb{N}^*$  satisfying  $M-1 \leq 2\log_2(1/\lambda) < M$ . We set  $F_1^{\lambda} := \sum_{j=1}^M f_j$ and  $F_2^{\lambda} := \sum_{j=M+1}^{+\infty} f_j$ . From the previous results, we have

$$|\Delta_h^m F_2^{\lambda}(x)| \le C|h|^{m-\alpha} \sum_{j=M+1}^{+\infty} 2^{j(m-\alpha)} \sigma_j$$

and

$$|\Delta_h^m F_1^{\lambda}(x)| \le C|h|^{m-\alpha+1} \sum_{j=1}^M 2^{j(m-\alpha+1)} \sigma_j.$$

r	-	-	-	-	
L					
н					
н					

**Remark 125.** Proposition 124 can easily be adapted to  $\mathbb{R}^d$ .

# Chapter 3

# Some applications of the generalized Hölder-Zygmund spaces

### 3.1 Generalized Hölder exponents

The main objective of this section is to give some conditions on admissible sequences  $\sigma^{(\alpha)}$  $(\alpha > 0)$  that lead to embedded generalized Hölder spaces, i.e. spaces such that  $\alpha < \beta$ implies  $\Lambda^{\sigma^{(\beta)},\beta}(\mathbb{R}^d) \subseteq \Lambda^{\sigma^{(\alpha)},\alpha}(\mathbb{R}^d)$ . For such spaces, we can define a generalized Hölder exponent of any function  $f \in L^{\infty}(\mathbb{R}^d)$  by

$$H_f^{\sigma^{(\alpha)}} = \sup\{\alpha > 0 : f \in \Lambda^{\sigma^{(\alpha)}, \alpha}(\mathbb{R}^d)\}.$$

This exponent gives information about the regularity of the function f.

#### 3.1.1 Preliminary results

The two main results of this section are the following. The first one is expressed in terms of  $d_0$  and the second one is expressed in terms of  $d_1$  (where  $d_0$  and  $d_1$  come from (1.2)).

**Proposition 126.** Let  $p \in [1, \infty]$ ,  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$  be an admissible sequence and  $f \in L^p(\mathbb{R}^d)$  satisfying

$$\sup_{|h| \le 2^{-j}} \|\Delta_h^M f\|_{L^p} \le C\sigma_j \qquad \forall j \in \mathbb{N}_0$$

with  $M \in \mathbb{N}^* \setminus \{1\}$ . We have three different cases:

1. if  $1 < 2^{M-1}d_0$ , then we have

$$\sup_{h|\leq 2^{-J}} \|\Delta_h^{M-1} f\|_{L^p} \le C\sigma_J + C2^{-J(M-1)} \qquad \forall J \in \mathbb{N}^*,$$
(3.1)

2. if  $1 > 2^{M-1}d_0$ , then we have

$$\sup_{|h| \le 2^{-J}} \|\Delta_h^{M-1} f\|_{L^p} \le C(2^{M-1} d_0)^{-J} \sigma_J + C2^{-J(M-1)} \qquad \forall J \in \mathbb{N}^*, \tag{3.2}$$

3. if  $1 = 2^{M-1}d_0 \Leftrightarrow d_0 = 2^{-(M-1)}$ , then we have

$$\sup_{|h| \le 2^{-J}} \|\Delta_h^{M-1} f\|_{L^p} \le C J \sigma_J + C 2^{-J(M-1)} \qquad \forall J \in \mathbb{N}^* \,.$$
(3.3)

**Proposition 127.** Let  $p \in [1, \infty]$ ,  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$  be an admissible sequence and  $f \in L^p(\mathbb{R}^d)$  satisfying

$$\sup_{|h| \le 2^{-j}} \|\Delta_h^M f\|_{L^p} \le C\sigma_j \qquad \forall j \in \mathbb{N}_0$$

with  $M \in \mathbb{N}^* \setminus \{1\}$ . We have three different cases:

1. if  $1 < 2^{M-1}d_1$ , then we have

$$\sup_{|h| \le 2^{-J}} \|\Delta_h^{M-1} f\|_{L^p} \le C d_1^J + C 2^{-J(M-1)} \qquad \forall J \in \mathbb{N}^*,$$
(3.4)

2. if  $1 > 2^{M-1}d_1$ , then we have

$$\sup_{|h| \le 2^{-J}} \|\Delta_h^{M-1} f\|_{L^p} \le C 2^{-(M-1)j} \qquad \forall J \in \mathbb{N}^*,$$
(3.5)

3. if  $1 = 2^{M-1}d_1 \Leftrightarrow d_1 = 2^{-(M-1)}$ , then we have

$$\sup_{|h| \le 2^{-J}} \|\Delta_h^{M-1} f\|_{L^p} \le CJ2^{-(M-1)J} \qquad \forall J \in \mathbb{N}^*.$$
(3.6)

The proposition 128 links finite differences of different orders. The proof can be obtained through an easy adaptation of proposition 69.

**Proposition 128.** Let  $m \in \mathbb{N}^*$ ,  $p \in [1, +\infty]$  and  $f : \mathbb{R}^d \to \mathbb{R}$ . We have

$$\|\Delta_h^m f\|_{L^p} \le \frac{m}{2} \|\Delta_h^{m+1} f\|_{L^p} + \frac{1}{2^m} \|\Delta_{2h}^m f\|_{L^p}$$

for all  $h \in \mathbb{R}^d$ .

*Proof of proposition 126.* Let us consider the following sum:

$$\sum_{j=0}^{J} 2^{-j(M-1)} \left( 2^{M-1} \sup_{|h| \le 2^{-J}} \|\Delta_{2^{j}h}^{M-1} f\|_{L^{p}} - \sup_{|h| \le 2^{-J}} \|\Delta_{2(2^{j}h)}^{M-1} f\|_{L^{p}} \right)$$
$$= 2^{M-1} \sup_{|h| \le 2^{-J}} \|\Delta_{h}^{M-1} f\|_{L^{p}} - 2^{-J(M-1)} \sup_{|h| \le 2^{-J}} \|\Delta_{2^{J+1}h}^{M-1} f\|_{L^{p}}, \quad J \in \mathbb{N}_{0}$$

For all  $J \in \mathbb{N}_0$ , we have

$$2^{M-1} \sup_{|h| \le 2^{-J}} \|\Delta_h^{M-1} f\|_{L^p} \le C \sum_{j=0}^J 2^{-j(M-1)} \sup_{|h| \le 2^{-(J-j)}} \|\Delta_h^M f\|_{L^p} + 2^{-J(M-1)} \sup_{|h| \le 2^{-J}} \|\Delta_{2^{J+1}h}^{M-1} f\|_{L^p} \le C \left(\sum_{j=0}^J (2^{-(M-1)} d_0^{-1})^j\right) \sigma_J + C 2^{-J(M-1)},$$

hence the conclusion.

Proof of proposition 127. We proceed similarly as in the proof of proposition 126. We have

$$2^{M-1} \sup_{|h| \le 2^{-J}} \|\Delta_h^{M-1} f\|_{L^p} \le C \sum_{j=0}^J 2^{-j(M-1)} \sup_{|h| \le 2^{-(J-j)}} \|\Delta_h^M f\|_{L^p} + 2^{-J(M-1)} \sup_{|h| \le 2^{-J}} \|\Delta_{2^{J+1}h}^{M-1} f\|_{L^p} \le C \left( \sum_{j=0}^J (2^{-(M-1)} d_1^{-1})^j \right) d_1^J + C 2^{-J(M-1)},$$

which ends the proof.

Let us note that each right member of inequalities (3.1)-(3.6) are new admissible sequences, by lemma 13.

Propositions 126 and 127 can be stated in terms of the Boyd index  $\overline{s}(\sigma^{-1})$  instead of  $d_0$  and  $d_1$ .

**Corollary 129.** Let  $p \in [1, \infty]$ ,  $\sigma = (\sigma_j)_{j \in \mathbb{N}_0}$  be an admissible sequence and  $f \in L^p(\mathbb{R}^d)$  satisfying

$$\sup_{|h| \le 2^{-j}} \|\Delta_h^M f\|_{L^p} \le C\sigma_j \qquad \forall j \in \mathbb{N}_0$$

with  $M \in \mathbb{N}^* \setminus \{1\}$ . We have the two following cases:

1. if  $\bar{s}(\sigma^{-1}) < M - 1$ , then

$$\sup_{|h| \le 2^{-J}} \|\Delta_h^{M-1} f\|_{L^p} \le C\sigma_J + C2^{-J(M-1)} \qquad \forall J \in \mathbb{N}^*,$$

2. if 
$$\bar{s}(\sigma^{-1}) \ge M - 1$$
, then

$$\sup_{|h| \le 2^{-J}} \|\Delta_h^{M-1} f\|_{L^p} \le C_{\varepsilon} 2^{J(\bar{s}(\sigma^{-1}) + \varepsilon - (M-1))} \sigma_J + C 2^{-J(M-1)} \qquad \forall J \in \mathbb{N}^*,$$

for all  $\varepsilon > 0$ .

*Proof.* Using (1.3), we have

$$2^{M-1} \sup_{|h| \le 2^{-J}} \|\Delta_h^{M-1} f\|_{L^p} \le C \sum_{j=0}^J 2^{-j(M-1)} \sup_{|h| \le 2^{-(J-j)}} \|\Delta_h^M f\|_{L^p} + 2^{-J(M-1)} \sup_{|h| \le 2^{-J}} \|\Delta_{2^{J+1}h}^{M-1} f\|_{L^p} \le C_{\varepsilon} \sigma_J \sum_{j=0}^J 2^{j(\overline{s}(\sigma^{-1}) + \varepsilon - (M-1))} + C 2^{-J(M-1)}$$

for all  $\varepsilon > 0$ , hence the conclusion follows.

Let us mention that corollary 129 expressed in terms of  $\overline{s}(\sigma^{-1})$  is a weaker result than propositions 126 and 127 stated in terms of  $d_0$  and  $d_1$ .

**Remark 130.** As a consequence, it is equivalent for finite differences in definition 40 to consider any bigger order than  $\lfloor \overline{s}(\sigma^{-1}) \rfloor + 1$  (we can use (1.3) to prove it). This result can also be obtained from theorem 33 without assumption  $\underline{s}(\sigma^{-1}) > 0$  being necessary.

### 3.1.2 Decreasing generalized Hölder spaces

The classical Hölder spaces  $\Lambda^{\alpha}(\mathbb{R}^d)$  are decreasing in the following sense: if  $\alpha < \beta$ , then  $\Lambda^{\beta}(\mathbb{R}^d) \subset \Lambda^{\alpha}(\mathbb{R}^d)$ . If  $\sigma^{(\alpha)}$  ( $\alpha > 0$ ) is a family of admissible sequences, let us set

$$\sigma^{(.)}: \alpha > 0 \mapsto \sigma^{(\alpha)}$$

The spaces  $\Lambda^{\sigma^{(\alpha)},\alpha}$  are not necessary embedded as the classical Hölder spaces.

**Definition 131.** A family of admissible sequence  $\sigma^{(.)}$  is *decreasing* if  $\alpha < \beta$  implies  $\Lambda^{\sigma^{(\beta)},\beta}(\mathbb{R}^d) \subseteq \Lambda^{\sigma^{(\alpha)},\alpha}(\mathbb{R}^d)$ .

A natural question arises: under which conditions can we obtain a family of decreasing admissible sequences? The following result gives an answer to that question:

**Proposition 132.** (D.K., S. Nicolay) A family of admissible sequences  $\sigma^{(.)}$  is decreasing if it satisfies the three following conditions:

1. for all  $m \in \mathbb{N}_0$  and  $\alpha, \beta > 0$  for which  $m \le \alpha < \beta < m + 1$ , there exist C, J > 0 such that

$$\sigma_j^{(\beta)} \le C \sigma_j^{(\alpha)} \quad \forall j \ge J;$$

2. for all  $m \in \mathbb{N}^*$ , there exists  $\varepsilon_0 > 0$  such that for all  $\varepsilon \in ]0, \varepsilon_0[$ , there exist C, J > 0 such that

$$2^{-jm} \le C\sigma_j^{(m-\varepsilon)} \quad \forall j \ge J;$$

3. for all  $m \in \mathbb{N}^*$ , at least one of the two following conditions is satisfied:

### DECREASING GENERALIZED HÖLDER SPACES

(a) • if  $1 < 2^m d_1^{(m)}$ , there exists  $\varepsilon_0 > 0$  such that for all  $\varepsilon \in ]0, \varepsilon_0[$ , there exist C, J > 0 such that

$$2^{-jm}(2^m d_1^{(m)})^j \le C\sigma_j^{(m-\varepsilon)} \quad \forall j \ge J;$$

• if  $1 > 2^m d_1^{(m)}$ , there exists  $\varepsilon_0 > 0$  such that for all  $\varepsilon \in ]0, \varepsilon_0[$ , there exist C, J > 0 such that

$$2^{-jm} \le C\sigma_j^{(m-\varepsilon)} \quad \forall j \ge J_j$$

• if  $1 = 2^m d_1^{(m)}$ , there exists  $\varepsilon_0 > 0$  such that for all  $\varepsilon \in ]0, \varepsilon_0[$ , there exist C, J > 0 such that

$$j2^{-jm} \le C\sigma_j^{(m-\varepsilon)} \quad \forall j \ge J.$$

(b) • if  $1 < 2^m d_0^{(m)}$ , there exists  $\varepsilon_0 > 0$  such that for all  $\varepsilon \in ]0, \varepsilon_0[$ , there exist C, J > 0 such that

$$\sigma_j^{(m)} \le C \sigma_j^{(m-\varepsilon)} \quad \forall j \ge J;$$

• if  $1 > 2^m d_0^{(m)}$ , there exists  $\varepsilon_0 > 0$  such that for all  $\varepsilon \in ]0, \varepsilon_0[$ , there exist C, J > 0 such that

$$\sigma_j^{(m)} (2^m d_0^{(m)})^{-j} \le C \sigma_j^{(m-\varepsilon)} \quad \forall j \ge J;$$

• if  $1 = 2^m d_0^{(m)}$ , there exists  $\varepsilon_0 > 0$  such that for all  $\varepsilon \in ]0, \varepsilon_0[$ , there exist C, J > 0 such that

$$j\sigma_j^{(m)} \le C\sigma_j^{(m-\varepsilon)} \quad \forall j \ge J.$$

(where  $d_0^{(m)}$  and  $d_1^{(m)}$  are some constants satisfying inequalities (1.2) for the admissible sequence  $\sigma^{(m)}$ ).

*Proof.* It is an immediate consequence of propositions 126 and 127.  $\Box$ 

**Remark 133.** In particular, conditions of proposition 132 imply that for all  $0 < \alpha < \beta$ , there exist C, J > 0 such that<sup>1</sup>

$$\sigma_j^{(\beta)} \le C \sigma_j^{(\alpha)} \quad \forall j \ge J.$$
(3.7)

Moreover, for  $0 < \alpha < m$ , there exists C > 0 such that

$$2^{-jm} \le C\sigma_j^{(\alpha)} \quad \forall j \in \mathbb{N}_0.$$
(3.8)

It is useless to check whether the conditions of the previous result are satisfied or not if the family of admissible sequences does not even satisfy these two simplified conditions.

<sup>&</sup>lt;sup>1</sup>This result is a consequence of condition 3 of proposition 132. This is obvious from the inequalities expressed in terms of  $d_0^{(m)}$ . For the ones expressed in terms of  $d_1^{(m)}$ , we proceed case by case, by rewriting inequalities exclusively in terms of  $d_1^{(m)}$ , and using relation (1.2).
**Remark 134.** At first glance, inequality (3.8) can be seen as a strong restriction, but it is not. Indeed, condition (1.2) in the definition of admissible sequences already requires a similar restriction<sup>2</sup>. Moreover, the example 135 shows that an admissible sequence with a high speed of convergence to 0 can lead to useless spaces.

**Example 135.** Consider the admissible sequences  $(2^{-j\delta\alpha})_j$  of usual Hölder spaces, and let us set  $\sigma_j^{(\alpha)} := 2^{-j\delta\alpha}$  with  $\delta > 0$  for all  $j \in \mathbb{N}_0$ . If  $\delta = 1$ , we recover the usual Hölder spaces. If  $0 < \delta < 1$ , it is easy to check that  $\Lambda^{\sigma^{(\alpha)},\alpha} = \Lambda^{\delta\alpha}(\mathbb{R}^d)$  (this is a consequence of remark 130, or also of theorem 33). Let us now consider the case  $\delta > 1$ . The spaces  $\Lambda^{\sigma^{(\alpha)},\alpha}$  do not satisfy the conditions of proposition 132. It is easy to check that if  $1 < \delta \leq 2$ , then the generated spaces are not embedded into one another. If  $\delta > 2$ , then there exists  $0 < \epsilon < 1$ such that for all  $\alpha > \epsilon$ , the space  $\Lambda^{\sigma^{(\alpha)},\alpha}$  is composed of constant functions<sup>3</sup>.

In particular, this example shows that we can construct admissible sequences such that condition 2 of proposition 132 is not satisfied, and such that the associated spaces are composed of constant functions. Those spaces are embedded (even equal). So, this shows that proposition 132 only gives sufficient conditions to construct decreasing families of admissible sequences.

As announced previously, the concept of decreasing families of admissible sequences allows the definition of a notion which characterizes the global regularity of functions.

**Definition 136.** Let  $\sigma^{(.)}$  be a decreasing family of admissible sequences. The generalized Hölder exponent associated with  $\sigma^{(.)}$  of a function  $f \in L^{\infty}(\mathbb{R}^d)$  is defined by

$$H_f^{\sigma^{(.)}} = \sup\{\alpha > 0 : f \in \Lambda^{\sigma^{(\alpha),\alpha}}\}.$$

**Example 137.** Let 0 < a < 1. Let us consider the admissible sequence  $\sigma$  defined by

$$\sigma_j = \frac{1}{j}$$
 for all  $j \in \mathbb{N}^*$ .

It is easy to check that any constants  $0 < d_0 < 1$  and  $d_1 \ge 1$  satisfy inequalities (1.2) for j sufficiently large. By lemma 14, the family  $(\sigma^{\alpha})_{\alpha>0}$  is a family of admissible sequences. We easily check, by using proposition 132, that it is a decreasing family of admissible sequences. The spaces  $\Lambda^{\sigma^{\alpha},\alpha}$  are made of functions such that their finite differences  $\Delta_h^{\lfloor\alpha\rfloor+1}$  have a slight decrease in the argument |h|. Moreover, it is easy to check that for all  $\alpha', \alpha > 0$ , we have  $\Lambda^{\alpha}(\mathbb{R}^d) \subset \Lambda^{\sigma^{\alpha'},\alpha'}$ . Let us remark that elements of those spaces are continuous.

**Example 138.** Let us consider a function  $g: \alpha > 0 \mapsto g(\alpha) \ge 0$ . Let us define the family of admissible sequences  $(\sigma^{(\alpha)})_{\alpha>0}$  by  $\sigma_j^{(\alpha)} := 2^{-j\alpha}j^{g(\alpha)}$  for all  $j \in \mathbb{N}^*$ . Proposition 132 proves that it is a decreasing family of admissible sequences. Let us notice that the condition 3.(b) of this result is satisfied but not the 3.(a) if g > 0.

<sup>&</sup>lt;sup>2</sup>It is important to remember that all convergences of exponential types, like  $(2^{2^{j}})_{j}$ , can not define an admissible sequence.

<sup>&</sup>lt;sup>3</sup>This is proved in remark 85.

**Example 139.** Let us define the family of sequences  $(\sigma^{(\alpha)})_{\alpha\geq 0}$  in the following way:

- if  $\alpha \in \mathbb{N}^*$ ,  $\sigma_j^{(\alpha)} := 2^{-j\alpha} \frac{1}{j}$  for all  $j \in \mathbb{N}^*$ ;
- if  $\alpha \in ]0, +\infty[\backslash \mathbb{N}^*, \sigma_j^{(\alpha)} := 2^{-j\alpha} \ \forall j \in \mathbb{N}^*.$

(

It is a family of admissible sequences. Condition 3.(b) of proposition 132 is not satisfied contrary to condition 3.(a). It is a decreasing family of admissible sequences.

**Remark 140.** The two previous examples show that conditions 3.(a) and 3.(b) of proposition 132 are complementary, i.e. one is not the consequence of the other.

**Example 141.** Proposition 132 gives an idea to construct recursively a decreasing family of admissible sequences. For all  $m \in \mathbb{N}^*$ , we would like to define

$$\sigma_j^{(m+1)} := \sigma_j^{(m)} \left( 2^{m+1} d_0^{(m+1)} \right)^j.$$

Such a definition is not complete because it depends on itself through the factor  $d_0^{(m+1)}$ . We could then ask what are the values of  $d_0^{(m+1)}$  which satisfy (1.2). It is easy to check that a sufficient condition is given by  $(d_0^{(m)})^{-1} \leq 2^{m+1}$   $(m \in \mathbb{N}^*)$ . If we set an initial condition  $\sigma^{(1)}$  such that  $(d_0^{(1)})^{-1} \leq 2^2$ , we can construct a family of admissible sequences by setting

$$\sigma_{j}^{(m)} = \sigma_{j}^{(m-1)} \left(2^{m} d_{0}^{(m)}\right)^{j}$$

$$= \sigma_{j}^{(m-2)} \left(2^{m-1} d_{0}^{(m-1)}\right)^{j} \left(2^{m} d_{0}^{(m)}\right)^{j}$$

$$= \cdots$$

$$= \sigma_{j}^{(1)} 2^{j \sum_{n=2}^{m} n} \left(\prod_{n=2}^{m} d_{0}^{(n)}\right)^{j}.$$

Let us study the conditions under which this construction leads to a decreasing family of admissible sequences. By construction, if we consider values of  $d_0^{(m)}$  in the interval  $[2^{-(m+1)}, 2^{-m}]$ , condition 3 of proposition 132 is satisfied. The last condition to check is

$$2^{-j(m+1)} \le C\sigma_j^{(m)} \quad \forall j \in \mathbb{N}_0.$$

To validate such a condition, we choose  $\sigma^{(1)}$  such that  $2^{-2j} \leq C\sigma_j^{(1)}$  for all j. Then, we find

$$\sigma_{j}^{(m)} = \sigma_{j}^{(1)} 2^{j \sum_{n=2}^{m} n} \left( \prod_{n=2}^{m} d_{0}^{(n)} \right)^{j}$$
  

$$\geq \sigma_{j}^{(1)} \prod_{n=2}^{m} \left( d_{0}^{(n-1)} \right)^{-j} \left( \prod_{n=2}^{m} d_{0}^{(n)} \right)^{j}$$
  

$$\geq C 2^{-2j} \left( \frac{d_{0}^{(m)}}{d_{0}^{(1)}} \right)^{j}.$$

So, a sufficient condition to apply proposition 132 is that for all  $m \ge 2$ , there exists C > 0 such that

$$2^{-j(m-1)} \le C \left(\frac{d_0^{(m)}}{d_0^{(1)}}\right)^j \quad \forall j \in \mathbb{N}_0.$$

Let us consider the particular case where  $d_0^{(m)} := 2^{m+1}$  for all  $m \in \mathbb{N}^*$ . We easily check that the family of sequences  $(\sigma^{(m)})_{m \in \mathbb{N}^*}$  satisfies the previous sufficient conditions. We can thus apply proposition 132. It is a decreasing family of admissible sequences and we note that

$$\sigma_j^{(m)} = \sigma_j^{(1)} 2^{-(m-1)j} \ge C 2^{-(m+1)j} \quad \forall j \in \mathbb{N}_0 \,.$$

### 3.2 Another definition of Hölder-Zygmund spaces

In this section, we prove that definition 6 is equivalent to the following one: a function f defined on  $\mathbb{R}^d$  belongs to the space  $\Lambda^{\alpha}(\mathbb{R}^d)$  ( $\alpha > 0$ ) if it is bounded almost everywhere and satisfies

$$\sup_{\substack{|h| \le 2^{-j} \\ x \in \mathbb{R}^d}} |\Delta_h^{\lfloor \alpha \rfloor + 1} f(x)| \le C 2^{-j\alpha} \qquad \forall j \in \mathbb{N}_0.$$
(3.9)

This means that the Lebesgue-measurability can be omitted in the definition of Hölder spaces if a slightly more restrictive inequality is imposed.

This is a consequence of the developments discussed in section 3.1. This result is not obvious at first glance as there exist non-measurable functions satisfying inequality (3.9).

**Lemma 142.** Let  $\alpha > 0$  and  $m \in \mathbb{N}^*$ . If the function  $f : \mathbb{R}^d \to \mathbb{R}$  is bounded almost everywhere and satisfies  $|\Delta_h^m f(x)| \leq C|h|^{\alpha} \ \forall x, h \in \mathbb{R}^d$ , then it is bounded.

*Proof.* Let us note that

$$\begin{split} |f(x)| - |\sum_{j=0}^{m-1} (-1)^j \binom{m}{j} f(x+mh-jh)| &\leq \left| |f(x)| - |\sum_{j=0}^{m-1} (-1)^j \binom{m}{j} f(x+mh-jh)| \right| \\ &\leq |\sum_{j=0}^m (-1)^j \binom{m}{j} f(x+mh-jh)| \\ &\leq |\Delta_h^m f(x)| \\ &\leq C|h|^{\alpha}, \end{split}$$

for all  $x, h \in \mathbb{R}^d$ . Let  $E \subseteq \mathbb{R}^d$  be a set such that its complement is negligible and such that  $|f(y)| \leq C'$  for all  $y \in E$ . Let also  $x \in E^c$  and  $h \in \mathbb{R}^d$  such that  $|h| \leq 1$  and  $|f(x + (m - j)h)| \leq C'$  for all  $j \in \{0, 1, ..., m - 1\}$  (such a real number h exists because the set E is the complementary of a negligible set). Putting these inequalities together, we find that  $|f(x)| \leq C''$ .

Let f be a function defined on  $\mathbb{R}^d$  and satisfying (3.9). Let us prove the result for  $\alpha \in \mathbb{N}^*$  and  $\alpha > 1$  (if  $\alpha < 1$ , the function is continuous and so measurable). Using results of section 3.1 (which can easily be adapted to inequality (3.9) thanks to lemma 142), we have

$$\sup_{\substack{|h| \le 2^{-j} \\ x \in \mathbb{R}^d}} |\Delta_h^{\alpha} f(x)| \le Cj 2^{-j\alpha} \qquad \forall j \in \mathbb{N}^*.$$
(3.10)

By induction, we can reduce the order of the finite difference to 1 to find that the function is continuous on  $\mathbb{R}^d$ , which ends the proof.

**Remark 143.** The equivalence of definitions proved in this section concerns the classical Hölder-Zygmund spaces. However, the same proof can also be applied to spaces  $\Lambda^{\sigma,\alpha}(\mathbb{R}^d)$  to obtain similar results on generalized Hölder spaces, under some technical conditions on the sequence  $\sigma$ .

# 3.3 The uniform irregular Hölder spaces $I^{\alpha}(\mathbb{R}^d)$ expressed as a particular case of generalized Hölder spaces $\Lambda^{\sigma,\alpha}(\mathbb{R}^d)$

The irregular Hölder spaces, introduced in [31, 32, 33], allow to study the irregularity of functions. Roughly speaking, they are the counterpart of Hölder spaces, and are obtained by reversing the inequality controlling the finite difference.

**Definition 144.** Let  $\alpha > 0$  and  $f \in L^{\infty}(\mathbb{R}^d)$ . A function f belongs to the *irregular Hölder* space  $I^{\alpha}(\mathbb{R}^d)$  if there exists a constant C > 0 such that

$$\sup_{|h| \le 2^{-j}} \|\Delta_h^{\lfloor \alpha \rfloor + 1} f\|_{L^{\infty}} \ge C 2^{-j\alpha} \qquad \forall j \in \mathbb{N}_0.$$

$$(3.11)$$

We can define an Hölder exponent linked to these spaces in the following way.

**Definition 145.** The upper global Hölder exponent (or uniform irregularity exponent) of a function  $f \in L^{\infty}(\mathbb{R}^d)$  is defined by

$$\overline{H_f} = \inf\{\alpha > 0 : f \in I^{\alpha}(\mathbb{R}^d)\}.$$

The aim of this section is to show that these spaces can be expressed in terms of generalized Hölder spaces. This allows to apply our previous results.

Let us remark that the inequality (3.11) is not equivalent to  $f \notin \Lambda^{\alpha}(\mathbb{R}^d)$ . Indeed, if  $f \notin I^{\alpha}(\mathbb{R}^d)$ , then for every C > 0, there exists a strictly increasing sequence  $(j_n)_{n \in \mathbb{N}^*}$  of integers such that

$$\sup_{|h| \le 2^{-j_n}} \|\Delta_h^{\lfloor \alpha \rfloor + 1} f\|_{L^{\infty}} \le C 2^{-j_n \alpha} \qquad \forall n \in \mathbb{N}^*.$$
(3.12)

This inequality is a weaker condition than the one used for usual Hölder spaces. We are led to the following natural definition.

**Definition 146.** Let  $\alpha > 0$  and  $f \in L^{\infty}(\mathbb{R}^d)$ . We say that f belongs to the weak Hölder space  $C^{\alpha}_w(\mathbb{R}^d)$  if  $f \notin I^{\alpha}(\mathbb{R}^d)$ .

The sequence  $(2^{-j_n\alpha})_{n\in\mathbb{N}^*}$  can be a non-admissible sequence (e.g. if  $j_n$  is of exponential type, see section 1.2). Moreover,  $|h| \leq 2^{-j_n}$  seems closer to the spaces  $\Lambda^{\alpha}_{\sigma,N}(\mathbb{R}^d)$  with  $N_n = 2^{j_n}$   $(n \in \mathbb{N}^*)$  than the spaces  $\Lambda^{\sigma,\alpha}(\mathbb{R}^d)$ . Again, the sequence  $N_j$  can be non-admissible. Nevertheless, the following result shows that this can be reduced to a particular case of generalized Hölder spaces of type  $\Lambda^{\sigma,\alpha}(\mathbb{R}^d)$ , for some admissible sequence  $\sigma$ .

**Proposition 147.** Let  $\alpha > 0$ ,  $M = \lfloor \alpha \rfloor + 1$  and  $f \in L^{\infty}(\mathbb{R}^d)$ . We have  $f \in C^{\alpha}_w(\mathbb{R}^d)$  if and only if for every C > 0, there exists a strictly increasing sequence  $(j_n)_{n \in \mathbb{N}^*}$  of integers (depending on C) such that  $f \in \Lambda^{\sigma((j_n)_n),\alpha}(\mathbb{R}^d)$  where  $\sigma((j_n)_n)$  is the admissible sequence defined by

$$\sigma((j_n)_n)_j = \inf\{2^{-\alpha j_n}; 2^{(M-\alpha)j_{n+1}}2^{-jM}\} \quad \forall j \in \{j_n, ..., j_{n+1}-1\}, \quad \forall n \in \mathbb{N}^*.$$

*Proof.* 1. Let C > 0 and  $(j_n)_n$  the sequence associated to C such that

$$\sup_{|h| \le 2^{-j_n}} \|\Delta_h^{\lfloor \alpha \rfloor + 1} f\|_{L^{\infty}} \le C 2^{-j_n \alpha} \qquad \forall n \in \mathbb{N}^*.$$

Let us prove that we have

$$\sup_{|h| \le 2^{-j}} \|\Delta_h^{\lfloor \alpha \rfloor + 1} f\|_{L^{\infty}} \le C\sigma((j_n)_n)_j \qquad \forall j \in \mathbb{N}^*.$$

Let  $n \in \mathbb{N}^*$  and  $j \in \{j_n, ..., j_{n+1} - 1\}$ . We have

$$\sup_{|h| \le 2^{-j}} \|\Delta_h^{\lfloor \alpha \rfloor + 1} f\|_{L^{\infty}} \le \sup_{|h| \le 2^{-j_n}} \|\Delta_h^{\lfloor \alpha \rfloor + 1} f\|_{L^{\infty}} \le C 2^{-\alpha j_n}.$$

We note that<sup>4</sup>

$$\sup_{|h| \le 2r} \|\Delta_h^{\lfloor \alpha \rfloor + 1} f\|_{L^{\infty}} \le 2^M \sup_{|h| \le r} \|\Delta_h^{\lfloor \alpha \rfloor + 1} f\|_{L^{\infty}} \quad \forall r > 0.$$

Using this result, we find

$$\sup_{\substack{|h| \le 2^{-j_{n+1}}2^{(j_{n+1}-j)}}} \|\Delta_h^{\lfloor \alpha \rfloor + 1} f\|_{L^{\infty}} \le 2^{(j_{n+1}-j)M} \sup_{\substack{|h| \le 2^{-j_{n+1}}}} \|\Delta_h^{\lfloor \alpha \rfloor + 1} f\|_{L^{\infty}} \le C 2^{(M-\alpha)j_{n+1}} 2^{-jM}.$$

2. Let us prove that  $(\sigma((j_n)_n)_j)_{j \in \mathbb{N}^*}$  is an admissible sequence with associated constants  $d_0 = 2^{-M}$  and  $d_1 = 1$  (these constants are optimal without any additional assumptions on the sequence  $(j_n)_n$ ), i.e. we have

$$2^{-M}\sigma((j_n)_n)_j \le \sigma((j_n)_n)_{j+1} \le \sigma((j_n)_n)_j \quad \forall j \in \mathbb{N}^*.$$

Let  $n \in \mathbb{N}^*$ . The result is immediate for  $j \in \{j_n, ..., j_{n+1}-2\}$ . Let us prove the result for  $j = j_{n+1} - 1$ . In order to simplify the notations, let us denote  $\sigma((j_n)_n)_j$  by  $\sigma_j$ . We consider the following different cases:

(a) if 
$$\sigma_{j_{n+1}-1} = 2^{-\alpha j_n}$$
 and  $\sigma_{j_{n+1}} = 2^{-\alpha j_{n+1}}$ , we have  $\sigma_{j_{n+1}} < \sigma_{j_{n+1}-1}$  and  
 $\sigma_{j_{n+1}-1} \le 2^{(M-\alpha)j_{n+1}}2^{-(j_{n+1}-1)M} = 2^{-\alpha j_{n+1}}2^M = \sigma_{j_{n+1}}2^M$ ,

(b) if  $\sigma_{j_{n+1}-1} = 2^{-\alpha j_n}$  and  $\sigma_{j_{n+1}} = 2^{(M-\alpha)j_{n+2}}2^{-j_{n+1}M}$ , we have

$$\sigma_{j_{n+1}} \le 2^{-\alpha j_{n+1}} \le 2^{-\alpha j_n} = \sigma_{j_{n+1}-1}$$

and

$$\sigma_{j_{n+1}-1} \leq 2^{-\alpha j_{n+1}} 2^M$$
  
=  $\sigma_{j_{n+1}} 2^{(M-\alpha)(j_{n+1}-j_{n+2})} 2^M$   
 $\leq 2^M \sigma_{j_{n+1}},$ 

(c) if  $\sigma_{j_{n+1}-1} = 2^{-\alpha j_{n+1}} 2^M$  and  $\sigma_{j_{n+1}} = 2^{-\alpha j_{n+1}}$ , we have  $\sigma_{j_{n+1}-1} = \sigma_{j_{n+1}} 2^M$ , (d) if  $\sigma_{j_{n+1}-1} = 2^{-\alpha j_{n+1}} 2^M$  and  $\sigma_{j_{n+1}} = 2^{(M-\alpha)j_{n+2}} 2^{-j_{n+1}M}$ , we have

$$\sigma_{j_{n+1}-1} = 2^{(M-\alpha)j_{n+1}} 2^{-j_{n+1}M} 2^M$$
$$\leq 2^{(M-\alpha)j_{n+2}} 2^{-j_{n+1}M} 2^M = \sigma_{j_{n+1}} 2^M$$

and

$$\sigma_{j_{n+1}} \le 2^{-\alpha j_{n+1}} = \sigma_{j_{n+1}-1} 2^{-M}.$$

This ends the proof.

**Remark 148.** We proved that  $d_1 = 1$  in proposition 147. The sequence  $\sigma((j_n)_n)$  is thus non-increasing.

70

<sup>&</sup>lt;sup>4</sup>This is a consequence of lemma 67.

#### **3.4** Application to financial models

**Example 149.** Let us set  $\sigma_j := (2^{-j})^{\frac{1}{2}} |\log|\log(2^{-j})||^{\frac{1}{2}}$  for all  $j \in \mathbb{N}^*$ . This is a strong admissible sequence of order 1. A. Khintchine proved in [72] that the trajectories of Brownian motions belong almost surely to  $\Lambda^{\sigma,\alpha}(\mathbb{R})$  (where  $0 < \alpha < 1$ ). So, its wavelet coefficients satisfy

$$\begin{cases} |C_k| \leq C & \forall k \in \mathbb{Z}^d \\ |c_{j,k}^i| \leq C\sigma_j & \forall j \in \mathbb{N}_0, \forall i \in \{1, \dots, 2^d - 1\}, \forall k \in \mathbb{Z}^d. \end{cases}$$

It is known that the trajectories of a Brownian motion do not belong to  $\Lambda^{1/2}(\mathbb{R})$  ([98, 102]). This example is a typical case where the generalized Hölder-Zygmund spaces give more information than what is given by classical Hölder spaces. Let us note that, in the classical case, even if the Brownian motion does not belong to  $\Lambda^{1/2}(\mathbb{R})$ , the usual Hölder exponent is still 1/2, which demonstrates the lack of accuracy of the classical Hölder spaces in this particular case.

**Example 150.** Let us consider the *geometric Brownian motion*. This stochastic process is a basic example used in financial models of stock indices, and is used in the famous Black and Scholes model ([62]). It is given by the following stochastic differential equation:

$$dS(t) = \mu S(t)dt + \sigma S(t)dW(t)$$

where  $(W(t))_{t\geq 0}$  is a Brownian motion,  $\mu \in \mathbb{R}$  is a real number determining the mean value of the stock and  $\sigma$  represents the volatility of the stock index. A solution of this stochastic equation is given by

$$S(t) = S(0)e^{(\mu - \sigma^2/2)t + \sigma W(t)}, \qquad t \ge 0,$$

where S(0) > 0 is a constant. Let us remark that we can rewrite this process as

$$S(t) = f(W(t))$$

where f is an infinitely continuously differentiable function on  $\mathbb{R}$ . So, we have

$$|f(W(t)) - f(W(s))| \le \max_{z \in K} |f'(z)| |W(t) - W(s)|$$

by the mean value theorem for some compact set K. We can conclude from example 149 that the trajectories of S(t) belong almost surely to  $\Lambda^{\sigma,\alpha}(\mathbb{R})$  (where  $\sigma$  is the sequence defined in example 149). Moreover, the trajectories do not belong to  $\Lambda^{1/2}(\mathbb{R})$  (otherwise we could apply the mean value theorem to the logarithm of S(t) in order to prove that the trajectories of the Brownian motion also belong to  $\Lambda^{1/2}(\mathbb{R})$ ).

**Example 151.** Let us consider the *Hull and White one-factor model*. This model was introduced in 1990 by John C. Hull and Alan White to model interest rates, and is popular



Figure 3.1: Simulations of trajectories of the stochastic process S, with  $\mu = 0,49\%$  and  $\sigma = 19,7\%$ . The estimation of parameters  $\mu$  and  $\sigma$  is based on observed values of Eurostoxx 50 during the year 2011 (these observed values are represented in grey in this graphic).

among financial market and actuarial fields ([23]). It is expressed by the following stochastic differential equation:

$$dr(t) = a(\theta(t) - r(t))dt + \sigma dW(t).$$

In this equation, the stochastic process r(t) should be understood as a short rate model and  $(W(t))_{t\geq 0}$  is a Brownian motion. All other factors are deterministic: the function  $\theta(t)$  is the asymptotic interest rate (which can vary over time because of macroeconomical changes in the future), a > 0 represents the attractive force towards which the interest rate converges to  $\theta(t)$ , and  $\sigma$  represents the volatility of the interest rate. The solution of this stochastic equation is given by

$$r(t) = r(0)e^{-at} + e^{-at}a \int_0^t \theta(s)e^{as}ds + \sigma e^{-at} \int_0^t e^{as}dW(s),$$

#### APPLICATION TO FINANCIAL MODELS

where r(0) > 0 represents the initial observed value of the interest rate. When  $\theta$  is a constant function, then the model is more commonly called the *Ornstein-Uhlenbeck* stochastic process or the Vasicek model. Under the Hull and White model, the price at time t of a zero-coupon bond that gives one unit of monetary at time T(T > t) can be written as

$$P(t,T) = e^{A(t,T) - B(t,T)r(t)},$$

where A(t,T) and B(t,T) are two regular deterministic functions. This expression can be used to fit the model to the observed initial yield curve and to predict values of bonds. More information about those models can be found in [23]. In practice, it is common to use the Nelson-Siegel or Svensson model to describe the behaviour of  $\theta$ . These models write the function  $\theta$  as exponential polynomials, and are used to calibrate the model to market data (see e.g. [75]). So, the function  $\theta$  can be assumed to be infinitely continuously differentiable. Let us set

$$g(t) := r(0)e^{-at} + e^{-at}a \int_0^t \theta(s)e^{as}ds \in C^\infty(\mathbb{R}).$$

By the mean value theorem, we find

$$|\Delta_h^1 g(t)| \le |h| \sup_K |g'|$$

where  $t, t + h \in K$  for some compact set K. Let us define the stochastic process Y by

$$Y(t) = \int_0^t e^{as} dW(s) \quad (t \ge 0).$$

Let us remark that

$$Y(t) = e^{at}W(t) - a\int_0^t e^{as}W(s)ds$$

by the integration by parts theorem for stochastic integrals (this classical result is a consequence of Ito's formula). Using the Leibniz formula for finite difference, we have

$$\Delta_h^1(e^{at}W(t)) = \Delta_h^1(e^{at})W(t+h) + e^{at}\Delta_h^1(W(t)).$$

By example 149, we have  $t \mapsto e^{at}W(t) \in \Lambda^{\sigma,\alpha}(\mathbb{R})$  with  $\sigma_j := (2^{-j})^{\frac{1}{2}} |\log|\log(2^{-j})||^{\frac{1}{2}}$   $(j \in \mathbb{N}^*)$ and  $0 < \alpha < 1$ . Therefore, by proposition 92, we get  $t \mapsto \int_0^t e^{as}W(s)ds \in \Lambda^{2^{-j}\sigma_j,1+\alpha}$  (in the sense that its trajectories belong almost surely to this space). Through the Leibniz formula and linearity of finite difference, one finally gets that  $r \in \Lambda^{\sigma,\alpha}$ . The trajectories of r do not belong to  $\Lambda^{1/2}$ . Otherwise, we could prove that the trajectories of the Brownian motion

$$W(t) = e^{-at}(Y(t) + a \int_0^t e^{as} W(s) ds)$$

also belong to this space, which is in contradiction with the results stated in example 149. We thus get that the trajectories of  $t \mapsto P(t,T)$  (with T > t) belong almost surely to  $\Lambda^{\sigma,\alpha}$  and do not belong to  $\Lambda^{1/2}$  (the same ideas as the ones exposed in example 150 can be applied).



Figure 3.2: Simulations of trajectories of the stochastic process r, with a = 9,46% and  $\sigma = 0,21\%$ . The estimation of parameters a and  $\sigma$  is based on observed values of Euribor 1 Week of year 2011.

**Example 152.** Let us define the stochastic process  $(X(t))_{t\geq 0}$  by

$$X(t) = \int_0^t W(s) ds.$$

This stochastic process is a Gaussian process and a martingale. It represents the area between a Brownian motion and the horizontal axis. Using the same arguments as the ones exposed in example 151, we easily check that the trajectories of X belong almost surely to  $\Lambda^{\sigma,1+\alpha}$  with  $\sigma_j := (2^{-j})^{\frac{3}{2}} |\log|\log(2^{-j})||^{\frac{1}{2}}$   $(j \in \mathbb{N}^*)$  and  $0 < \alpha < 1$ .

The reader should note that all ideas expressed in this section could be used to determine the generalized Hölder exponent of most stochastic processes which are derived from stochastic integration.

### Chapter 4

## Characterizations and properties of pointwise generalized Hölder-Zygmund spaces

The study of global regularity of a function can be made through the use of generalized Hölder-Zygmund spaces. For many functions used as models for signals, however, the regularity can change drastically from one point to another. It is therefore natural to define a notion of pointwise regularity. For example, the function  $f(x) = x^{m+1} \sin(x^{-m})$  $(m \in \mathbb{N}_0)$  is clearly irregular at x = 0 although it admits a continuous extension on  $\mathbb{R}$  by f(0) = 0. This function is represented in figure 4.1 for m = 2. This function is clearly infinitely continuously differentiable on  $\mathbb{R}_0$  (but is not continuously differentiable on  $\mathbb{R}$ ). It could be interesting in this case to determine "how much" the function is irregular at x = 0. Another example is given by the Takagi function. We treat this case in section 4.6.

For those purposes, we introduce in this section pointwise Hölder spaces and their generalized versions. Later on, we also look at some characterizations of these spaces. Their properties are very similar to their global versions, although these spaces can not be manipulated in the same way.

In the sequel, the notation  $||f||_E$  stands for  $||f||_E = \sup_{x \in E} |f(x)|$ .

# 4.1 Definition of pointwise generalized Hölder-Zygmund spaces $\Lambda^{M}_{\sigma,N}(x_0)$ and $\Lambda^{\sigma,M}(x_0)$

Let us recall the definition of classical pointwise Hölder spaces, which are denoted by  $\Lambda^{\alpha}(x_0)$  $(\alpha > 0, x_0 \in \mathbb{R}^d).$ 

**Definition 153.** Let  $\alpha > 0$  and  $x_0 \in \mathbb{R}^d$ . We say that a function  $f \in L^{\infty}_{loc}(\mathbb{R}^d)$  belongs to



Figure 4.1: Graphic of the function  $f(x) = x^3 \sin(x^{-2})$ . The red dot represents the continuous extension at x = 0 of the function f.

the space  $\Lambda^{\alpha}(x_0)$  if there exist two constants C, J > 0 such that<sup>1</sup>

$$\inf_{P \in \mathbb{P}_{\lfloor \alpha \rfloor}} \|f - P\|_{B(x_0, 2^{-j})} \le C 2^{-j\alpha} \quad \forall j \ge J.$$

By analogy with the global Hölder-Zygmund spaces, we generalize this definition as follows.

**Definition 154.** Let  $M \in \mathbb{N}_0$ ,  $\sigma$  and N be two admissible sequences, and  $x_0 \in \mathbb{R}^d$ . We say that a function  $f \in L^{\infty}_{loc}(\mathbb{R}^d)$  belongs to the space  $\Lambda^M_{\sigma,N}(x_0)$  if there exist two constants C, J > 0 such that

$$\inf_{P \in \mathbb{P}_M} \|f - P\|_{B(x_0, N_j^{-1})} \le C\sigma_j \quad \forall j \ge J.$$

In the case  $N_j = 2^j$   $(j \in \mathbb{N}_0)$ , we denote these spaces by  $\Lambda^{\sigma,M}(x_0) = \Lambda^M_{\sigma,N}(x_0)$ .

**Remark 155.** 1. The previous definition can be rewritten in the following way. A function  $f \in L^{\infty}_{loc}(\mathbb{R}^d)$  belongs to the space  $\Lambda^M_{\sigma,N}(x_0)$  if there exist two constants

<sup>&</sup>lt;sup>1</sup>We remind the reader that the notation  $\mathbb{P}_M$  refers to the set of polynomials defined on  $\mathbb{R}^d$  of degree less or equal to M.

C, J > 0 such that for all  $j \ge J$ , there exists a polynomial  $P_{x_0,j}$  of degree less or equal to M which satisfies

$$\sup_{|h| \le N_j^{-1}} |f(x_0 + h) - P_{x_0,j}(h)| \le C\sigma_j.$$
(4.1)

2. Let  $\sigma$  be an admissible sequence such that  $\underline{s}(\sigma^{-1}) > 0$ . If a function f belongs to a generalized Hölder space  $\Lambda^{\sigma,M}(\mathbb{R}^d)$ , then it belongs in particular to the pointwise space  $\Lambda^{\sigma,M}(x_0)$  for all  $x_0 \in \mathbb{R}^d$ . The converse is not true: if a function  $f \in L^{\infty}(\mathbb{R}^d)$ belongs to  $\Lambda^{\sigma,M}(x_0)$  for all  $x_0 \in \mathbb{R}^d$ , then it doesn't necessary belong<sup>2</sup> to  $\Lambda^{\sigma,M}(\mathbb{R}^d)$ (the constant C depends on  $x_0$ ).

**Remark 156.** We have introduced a generalization of global and pointwise Hölder-Zygmund spaces in terms of admissible sequences. Thus, it is natural to complete our study of Hölder-Zygmund spaces with the local point of view of these spaces. We propose a definition of local Hölder spaces that can be considered as a particular case of the global spaces. This means that all of our results which are true for the global case can be transposed to the local case.

**Definition 157.** Let  $\alpha > 0$ ,  $\sigma$  be an admissible sequence and  $x_0 \in \mathbb{R}^d$ . A function  $f \in L^{\infty}_{loc}(\mathbb{R}^d)$  belongs to the space  $\Lambda^{\sigma,\alpha}_{loc}(x_0)$  if there exists a function  $g \in \Lambda^{\sigma,\alpha}(\mathbb{R}^d)$  such that f = g on a neighbourhood of  $x_0$ , i.e. there exists C > 0 such that

$$\sup_{|h| \le 2^{-j}} \|\Delta_h^{\lfloor \alpha \rfloor + 1} g\|_{L^{\infty}(\mathbb{R}^d)} \le C \sigma_j \quad \forall j \in \mathbb{N}_0.$$

In particular, there exists a neighbourhood  $\nu$  of  $x_0$  such that

$$\sup_{|h| \le 2^{-j}} \|\Delta_h^{\lfloor \alpha \rfloor + 1} f\|_{L^{\infty}(\nu)} \le C\sigma_j \quad \forall j \in \mathbb{N}_0.$$

## 4.2 A characterization of pointwise generalized Hölder-Zygmund spaces in terms of finite differences

The definition of generalized pointwise Hölder spaces is expressed in terms of polynomials. Because of the analogy between pointwise Hölder spaces and global Hölder spaces, it is natural to wonder whether these spaces can be expressed in terms of finite differences. The result here below provides the answer.

**Proposition 158.** (D.K., S. Nicolay) Let  $M \in \mathbb{N}_0$ ,  $\sigma$  and N be two admissible sequences such that  $N_j \to +\infty$ . If  $f \in L^{\infty}_{loc}(\mathbb{R}^d)$  is a function which is continuous in a neighbourhood of  $x_0 \in \mathbb{R}^d$ , then the following conditions are equivalent:

<sup>&</sup>lt;sup>2</sup>Such an example is given in remark 164.

- 1.  $f \in \Lambda^M_{\sigma,N}(x_0),$
- 2. there exist C > 0 and J > 0 such that

$$\sup_{|h| \le N_j^{-1}} \|\Delta_h^{M+1} f\|_{B_h(x_0, N_j^{-1})} \le C\sigma_j \quad \forall j \ge J.$$

*Proof.* Let us prove that 2.  $\Rightarrow$  1. Let r > 0 such that f is continuous in  $B(x_0, r)$  and  $J' \ge J$  such that  $N_j^{-1} \le r$  for all  $j \ge J'$ . Because f is bounded on  $B(x_0, r)$ , using Whitney theorem ([24], theorem 1'), we have

$$\inf_{P \in \mathbb{P}_M} \|f - P\|_{B(x_0, N_j^{-1})} \le C \sup_{|h| \le N_j^{-1}} \|\Delta_h^{M+1} f\|_{B_h(x_0, N_j^{-1})} \quad \forall j \ge J'.$$

Let us prove that  $1. \Rightarrow 2$ . By assumption, there exists  $J \in \mathbb{N}_0$  such that, for all  $j \ge J$ , there exists a polynomial  $P_{x_0,j}$  of degree less or equal to M satisfying

$$\sup_{|h| \le N_j^{-1}} |f(x_0 + h) - P_{x_0,j}(h)| \le C\sigma_j.$$

Let  $j \geq J$ ,  $x, h \in \mathbb{R}^d$  such that  $[x, x + (M+1)h] \subseteq B(x_0, N_j^{-1})$  and let us set  $Q_{x_0,j}(.) = P_{x_0,j}(.-x_0)$ . We have

$$\begin{aligned} |\Delta_h^{M+1} f(x)| &= |\Delta_h^{M+1} (f - Q_{x_0,j})(x)| \\ &\leq 2^{M+1} \sup_{y \in B(x_0, N_j^{-1})} |f(y) - Q_{x_0,j}(y)|. \end{aligned}$$

So, we find

$$\sup_{|h| \le N_j^{-1}} \|\Delta_h^{M+1} f\|_{B_h(x_0, N_j^{-1})} \le 2^{M+1} C \sigma_j.$$

Remark 159. Proposition 158 is the pointwise version of theorem 88.

# 4.3 A characterization of pointwise generalized Hölder-Zygmund spaces in terms of Taylor decomposition

In section 2.7, we have seen that, under some assumptions on  $\sigma$ , generalized Hölder-Zygmund spaces admit a Taylor decomposition of their elements (see corollary 105). The goal of this section is to prove a similar property for pointwise spaces.

Let us recall the following multidimensional Markov inequality: let  $p \in [0, +\infty]$ ,  $i \in \{1, ..., d\}$  and  $S \subset \mathbb{R}^d$  be a bounded convex set with non-empty interior; we have

$$||D_iP||_{L^p(S)} \le Cn^2 ||P||_{L^p(S)}$$

for every polynomials of degree less or equal to n-1, where the constant C > 0 only depends on S and p, and does not depend on the polynomial P or n (see [44], theorem 4.1.). We deduce the following inequality: if  $x_0 \in \mathbb{R}^d$ , then we have

$$||D_iP||_{L^{\infty}(B(x_0,R))} \le CR^{-1}n^2 ||P||_{L^{\infty}(B(x_0,R))}$$
(4.2)

for all R > 0 and for all polynomials of degree less or equal to n - 1.

We need a lemma to prove the main result of this section.

**Lemma 160.** Let  $M \in \mathbb{N}_0$ ,  $\sigma$  and N be two admissible sequences such that  $M < \underline{s}(\sigma^{-1})\overline{s}(N)^{-1}$ and  $\underline{N}_1 \geq 1$ . If  $f \in \Lambda^M_{\sigma,N}(x_0)$ , then the sequence  $(P_{x_0,j})_{j \in \mathbb{N}_0}$  composed of polynomials satisfying (4.1) verifies

$$|D^{\beta}P_{x_0,l}(x_0) - D^{\beta}P_{x_0,j}(x_0)| \le CN_j^{|\beta|}\sigma_j \quad \forall j \le l, \forall |\beta| \le M.$$

*Proof.* Using Markov's inequality (4.2), we have

$$\begin{split} \|D^{\beta}(P_{x_{0},j} - P_{x_{0},j+1})\|_{B(x_{0},N_{j+1}^{-1})} \\ &\leq CN_{j}^{|\beta|}\|P_{x_{0},j} - P_{x_{0},j+1}\|_{B(x_{0},N_{j+1}^{-1})} \\ &\leq CN_{j}^{|\beta|}\left(\|P_{x_{0},j} - f\|_{B(x_{0},N_{j}^{-1})} + \|P_{x_{0},j+1} - f\|_{B(x_{0},N_{j+1}^{-1})}\right) \\ &\leq CN_{j}^{|\beta|}(\sigma_{j} + \sigma_{j+1}) \\ &\leq CN_{j}^{|\beta|}\sigma_{j} \end{split}$$

for all  $|\beta| \leq M$  and  $j \in \mathbb{N}_0$ . If  $j \leq l$ , we find

$$\begin{split} \|D^{\beta}(P_{x_{0},j} - P_{x_{0},l})\|_{B(x_{0},N_{l}^{-1})} \\ &\leq \sum_{k=j}^{l-1} \|D^{\beta}(P_{x_{0},k} - P_{x_{0},k+1})\|_{B(x_{0},N_{l}^{-1})} \\ &\leq \sum_{k=j}^{l-1} \|D^{\beta}(P_{x_{0},k} - P_{x_{0},k+1})\|_{B(x_{0},N_{k+1}^{-1})} \\ &\leq C \sum_{k=j}^{l-1} N_{k}^{|\beta|} \sigma_{k} \\ &\leq C N_{j}^{|\beta|} \sigma_{j}. \end{split}$$

**Remark 161.** Under the assumption of lemma 160, the sequence  $(D^{\beta}P_{x_0,j}(x_0))_{j\in\mathbb{N}_0}$  ( $|\beta| \leq M$ ) is a Cauchy sequence. Let  $D^{\beta}f(x_0)$  denote its limit. Its value is called the  $\beta$ -th Peano's

derivative of f at  $x_0$  (see [42]), and is also called the  $\beta$ -th "De la Vallée-Poussin" derivative of f at  $x_0$ . It does not depend on the chosen polynomial sequence  $(D^{\beta}P_{x_0,j}(x_0))_{j\in\mathbb{N}_0}$ satisfying (4.1). If  $(D^{\beta}P'_{x_0,j}(x_0))_{j\in\mathbb{N}_0}$  is another polynomial sequence satisfying (4.1), then we have

$$|D^{\beta}P_{x_{0},j}'(x_{0}) - D^{\beta}f(x_{0})| \le |D^{\beta}P_{x_{0},j}'(x_{0}) - D^{\beta}P_{x_{0},j}(x_{0})| + |D^{\beta}P_{x_{0},j}(x_{0}) - D^{\beta}f(x_{0})|.$$

Using Markov's inequality, we obtain

$$\begin{split} \|D^{\beta}(P_{x_{0},j} - P'_{x_{0},j})\|_{B(x_{0},N_{j}^{-1})} \\ &\leq CN_{j}^{|\beta|}\|P_{x_{0},j} - P'_{x_{0},j}\|_{B(x_{0},N_{j}^{-1})} \\ &\leq CN_{j}^{|\beta|}\left(\|P_{x_{0},j} - f\|_{B(x_{0},N_{j}^{-1})} + \|P'_{x_{0},j} - f\|_{B(x_{0},N_{j}^{-1})}\right) \\ &\leq CN_{j}^{|\beta|}\sigma_{j} \to 0 \quad \text{if } j \to +\infty, \end{split}$$

so the conclusion. The proof of the next result justifies its name as "derivative".

**Theorem 162.** (D.K., S. Nicolay) Let  $M \in \mathbb{N}_0$ ,  $\sigma$  and N be two admissible sequences such that  $M \leq \underline{s}(\sigma^{-1})\overline{s}(N)^{-1}$  and  $\underline{N}_1 > 1$ . The following assertions are equivalent:

- 1.  $f \in \Lambda^M_{\sigma,N}(x_0),$
- 2. there exist a positive constant C and a polynomial  $P_{x_0}$  of degree less or equal to M such that

$$\sup_{|h| \le N_j^{-1}} |f(x_0 + h) - P_{x_0}(h)| \le C\sigma_j \qquad \forall j \in \mathbb{N}_0.$$

*Proof.* The proof of  $2 \Rightarrow 1$  is immediate. Let us prove that  $1 \Rightarrow 2$ . Let  $(P_{x_0,j})_{j \in \mathbb{N}_0}$  be a sequence of polynomials of degree less or equal to M satisfying (4.1). We set

$$P_{x_0}(x) = \sum_{|\beta| \le M} D^{\beta} f(x_0) \frac{(x - x_0)^{\beta}}{|\beta|!}$$

for  $x \in B(x_0, N_0^{-1})$ . We have

$$\begin{aligned} \|P_{x_0} - P_{x_0,j}\|_{B(x_0,N_j^{-1})} &= \left\| \sum_{|\beta| \le M} \left( D^{\beta} f(x_0) - D^{\beta} P_{x_0,j}(x_0) \right) \frac{(x-x_0)^{\beta}}{|\beta|!} \right\|_{B(x_0,N_j^{-1})} \\ &\le \sum_{|\beta| \le M} |D^{\beta} f(x_0) - D^{\beta} P_{x_0,j}(x_0)| N_j^{-|\beta|}. \end{aligned}$$

Using lemma 160, we have

$$|D^{\beta}P_{x_0,l}(x_0) - D^{\beta}P_{x_0,j}(x_0)| \le CN_j^{|\beta|}\sigma_j \quad \forall j \le l,$$

and, by taking  $l \to +\infty$ , we find

$$|D^{\beta}f(x_{0}) - D^{\beta}P_{x_{0},j}(x_{0})| \le CN_{j}^{|\beta|}\sigma_{j}.$$

So, we obtain

$$||P_{x_0} - P_{x_0,j}||_{B(x_0,N_j^{-1})} \le C\sigma_j \quad \forall j \in \mathbb{N}_0.$$

The conclusion follows immediately from

$$||f - P_{x_0}||_{B(x_0, N_j^{-1})} \le ||f - P_{x_0, j}||_{B(x_0, N_j^{-1})} + ||P_{x_0 j} - P_{x_0}||_{B(x_0, N_j^{-1})} \le C\sigma_j$$

for all  $j \in \mathbb{N}_0$ .

**Remark 163.** Under the assumptions of theorem 162, a function f of  $\Lambda^M_{\sigma,N}(x_0)$  admits a *Taylor decomposition* around the point  $x_0$ , expressed in terms of its Peano derivatives at  $x_0$ : for x in a neighbourhood of  $x_0$ , we have

$$f(x) = \sum_{|\beta| \le M} D^{\beta} f(x_0) \frac{(x - x_0)^{\beta}}{|\beta|!} + R(x - x_0),$$

where  $\sup_{|h| \leq N_j^{-1}} |R(h)| \leq C\sigma_j$  for j sufficiently large. Under the assumptions of theorem 162, if f is M-times continuously differentiable around  $x_0$ , then the Peano derivatives at  $x_0$  (up to the order M) coincide with classical derivatives. This justifies their name.

**Remark 164.** We conclude this section by a general remark concerning Peano derivatives. There exist functions which are not differentiable but admit Peano derivatives. Here is a basic example.

Let m > 1 and  $f : \mathbb{R} \to \mathbb{R}$  be the function defined by  $f(x) = x^{m+1} \sin(x^{-m})$  at  $x \neq 0$ and f(0) = 0. We can check that f admits Peano derivatives up to the order m in all points of  $\mathbb{R}$  but its first derivative is not continuous at 0. Moreover, Peano derivatives of f at 0 are equal to 0 up to the order m, so  $f \in \Lambda^{(2^{-j(m+1)})_{j,m}}(0)$  by theorem 162. This space is not equal to any classical pointwise Hölder spaces. In particular, we find that  $f \in \Lambda^{\alpha}(0)$  for all  $\alpha < m+1$ , but  $f \notin \Lambda^{1+\varepsilon}(\mathbb{R})$  for all  $\varepsilon > 0$  (otherwise the function f would be continuously differentiable according to proposition 92).

### 4.4 A characterization of pointwise generalized Hölder-Zygmund spaces in terms of wavelet coefficients

We proved in section 2.9 that generalized global Hölder-Zygmund spaces can be characterized in terms of wavelet coefficients. The goal of this section is to prove a similar result for pointwise Hölder spaces.

We consider Daubechies wavelets in the sequel. Moreover we suppose that such wavelets are sufficiently regular and have sufficiently many vanishing moments<sup>3</sup>. We let  $j_0$  denote a natural number satisfying

$$supp \psi^{i} \subseteq B(0, \leq 2^{j_{0}}) \quad \forall i \in \{1, ..., 2^{d} - 1\}.$$

First, let us introduce some notations. A dyadic cube of scale  $j \in \mathbb{N}_0$  is a cube that can be written as

$$\lambda = \prod_{i=1}^{d} \left[ \frac{k_i}{2^j}, \frac{k_i + 1}{2^j} \right[$$

where  $k = (k_1, ..., k_d) \in \mathbb{Z}^d$ . We consider the indices  $i \in \{1, ..., 2^d - 1\}$ ,  $j \in \mathbb{Z}$ ,  $k \in \mathbb{Z}^d$ which characterize the wavelet coefficients  $c_{j,k}^i$ . We can suppose that *i* takes its value in the set  $\{0, 1\}^d \setminus (0, ..., 0)$  without loss of generality. We let  $\lambda = \lambda(i, j, k) = \frac{k}{2^j} + \frac{i}{2^{j+1}} + [0, \frac{1}{2^{j+1}})^d$ denote the dyadic cube associated with the wavelet coefficient  $c_\lambda = c_{j,k}^i$ .

**Definition 165.** The wavelet leaders are defined by

$$d_{\lambda} = \sup_{\lambda' \subseteq \lambda} |c_{\lambda'}|$$

If  $f \in L^{\infty}(\mathbb{R}^d)$ , the wavelet leaders are finite because

$$|c_{\lambda}| \leq 2^{dj} \int_{\mathbb{R}^d} |f(x)| |\psi_{\lambda}(x)| dx \leq C ||f||_{L^{\infty}(\mathbb{R}^d)}.$$

**Definition 166.** Two dyadic cubes  $\lambda_1$  and  $\lambda_2$  are said to be adjacent if they have the same scale and if dist $(\lambda_1, \lambda_2) = 0$  (so, a dyadic cube is adjacent to itself). Let  $\lambda_j(x_0)$  denote the dyadic cube of side  $2^{-j}$  containing  $x_0$  and  $3\lambda$  denote the set of the  $3^d$  dyadic cubes which are adjacent to  $\lambda$ ; then

$$d_j(x_0) = \sup_{\lambda' \in 3\lambda_j(x_0))} d_{\lambda'}.$$

The characterization of generalized pointwise Hölder spaces is expressed in terms of  $d_j(x_0)$   $(j \in \mathbb{N})$ . The factor 3 in the definition of  $d_j(x_0)$  is needed for technical requirements, so that the characterization remains true even for limit cases.

The wavelet characterization needs an assumption on the global regularity of the considered functions. This assumption is a little more restrictive than continuity on  $\mathbb{R}^d$ . It is expressed by the following definition.

<sup>&</sup>lt;sup>3</sup>Using the notations from the sequel, we suppose that the wavelets belong to the space  $C^{M+1}(\mathbb{R}^d)$  and have M + 1 vanishing moments.

**Definition 167.** A function f is said to be uniformly Hölder if there exists  $\varepsilon > 0$  such that  $f \in \Lambda^{\varepsilon}(\mathbb{R}^d)$ .

**Remark 168.** One can easily check that a function f is uniformly Hölder if and only if there exists an admissible sequence  $\sigma$  such that  $\underline{s}(\sigma^{-1}) > 0$  and  $f \in \Lambda^{\sigma,M}(\mathbb{R}^d)$  for a certain natural number M.

**Theorem 169. (D.K., S. Nicolay)** Let  $M \in \mathbb{N}_0$ ,  $x_0 \in \mathbb{R}^d$  and  $\sigma$  be an admissible sequence. If  $f \in \Lambda^{\sigma,M}(x_0)$ , then there exist C > 0 and  $J \in \mathbb{N}_0$  such that

$$d_j(x_0) \le C\sigma_j \qquad \forall j \ge J. \tag{4.3}$$

Conversely, let us suppose that  $\sigma_j \to 0$  if  $j \to +\infty$ . If f is uniformly Hölder and if (4.3) is satisfied, then  $f \in \Lambda^{\sigma',M}(x_0)$  where  $\sigma'$  is a new admissible sequence defined by  $\sigma'_j = \sigma_j |\log_2(\sigma_j)| \ (j \in \mathbb{N}_0)$  and M is a natural number satisfying  $M + 1 > \overline{s}(\sigma^{-1})$ .

*Proof.* Let us suppose that  $f \in \Lambda^{\sigma,M}(x_0)$ ,  $k_0 \in \mathbb{N}_0$  be such that  $2^{\max(j_0;d)+3} + 4d \leq 2^{k_0}$  and let also  $j \geq k_0 + 1$  and  $\lambda = \lambda(i, j', k') \subset 3\lambda_j(x_0)$  (in particular we have  $j' \geq j - 2$ ). We find

$$\begin{aligned} |c_{\lambda}| &= \left| 2^{dj'} \int_{\mathbb{R}^{d}} f(x)\psi^{i}(2^{j'}x - k')dx \right| \\ &= \left| 2^{dj'} \int_{\mathbb{R}^{d}} (f(x) - P_{x_{0},j-k_{0}}(x))\psi^{i}(2^{j'}x - k')dx \right| \\ &= \left| 2^{dj'} \int_{B(\frac{k'}{2^{j'}},2^{j_{0}-j'})} (f(x) - P_{x_{0},j-k_{0}}(x))\psi^{i}(2^{j'}x - k')dx \right| \\ &\leq 2^{dj'} \int_{B(x_{0},2^{-(j-k_{0})})} |f(x) - P_{x_{0},j-k_{0}}(x)||\psi^{i}(2^{j'}x - k')|dx \\ &\leq C\sigma_{j-k_{0}}2^{dj'} \int_{\mathbb{R}^{d}} |\psi^{i}(2^{j'}x - k')|dx \leq C\sigma_{j}. \end{aligned}$$

Conversely, let us suppose that the wavelet coefficients satisfy (4.3) and that there exists  $\varepsilon > 0$  such that  $f \in \Lambda^{\varepsilon}(\mathbb{R}^d)$ . We proved in theorem 115 that the functions  $f_j$   $(j \in \mathbb{N}_0 \cup \{-1\})$  have the same regularity as wavelets and that  $f = \sum_{j=-1}^{+\infty} f_j$  uniformly on  $\mathbb{R}^d$ . Let us set

$$P_{x_0,J}(x-x_0) := \sum_{|\beta| \le M} \frac{(x-x_0)^{\beta}}{|\beta|!} \sum_{j=-1}^J D^{\beta} f_j(x_0).$$

This is a polynomial of degree less or equal to M. Let  $n_d$  be a natural number such that, for each dyadic number  $\frac{k}{2^j}$   $(k \in \mathbb{Z}^d, j \in \mathbb{N}_0)$ , for each  $R \geq 2^{-j}$  and  $x \in \mathbb{R}^d$  such that  $\frac{k}{2^j} \in B(x, R)$ , the dyadic cube  $\frac{k}{2^j} + \frac{i}{2^{j+1}} + [0, \frac{1}{2^j}]^d$  is included in the ball  $B(x, 2^{n_d}R)$ . Let  $m_d$  be a natural number such that each ball  $B(x, 2^{-j})$   $(x \in \mathbb{R}^d, j \in \mathbb{N}_0)$  is included in a dyadic cube (whose vertices have dyadic coordinates) with a side length  $2^{m_d}2^{-j}$ . Finally, let  $J \geq \sup\{J', j_0 + n_d + m_d + 1\}$ , where  $J' \in \mathbb{N}_0$  is such that  $\sigma_j < 1$  for all  $j \geq J'$ . We have

$$\|f - P_{x_0,J}\|_{B(x_0,2^{-J})} \le \sum_{j=-1}^{J} \|f_j(x) - \sum_{|\beta| \le M} \frac{(x-x_0)^{\beta}}{|\beta|!} D^{\beta} f_j(x_0)\|_{B(x_0,2^{-J})} + \sum_{j=J+1}^{+\infty} \|f_j\|_{B(x_0,2^{-J})}.$$

Let  $J_1 \in \mathbb{N}_0$  denote the unique natural number satisfying  $2^{-\varepsilon J_1} \leq \sigma_J < 2^{-\varepsilon (J_1-1)}$ .

1. Firstly, we study the term

$$\sum_{j=-1}^{J} \|f_j(x) - \sum_{|\beta| \le M} \frac{(x-x_0)^{\beta}}{|\beta|!} D^{\beta} f_j(x_0)\|_{B(x_0, 2^{-J})}.$$

Let  $j \leq J$ . By the Taylor decomposition, we find

$$\|f_j(x) - \sum_{|\beta| \le M} \frac{(x - x_0)^{\beta}}{|\beta|!} D^{\beta} f_j(x_0) \|_{B(x_0, 2^{-J})}$$
$$\le C 2^{-J(M+1)} \sup_{|\beta| = M+1} \|D^{\beta} f_j\|_{B(x_0, 2^{-J})}.$$

If  $|\beta| = M + 1$ , we prove that we have

$$\|D^{\beta}f_{j}\|_{B(x_{0},2^{-J})} \leq C\sigma_{j}2^{j(M+1)}\sup_{i}\|\sum_{k\in\mathbb{Z}^{d}}D^{\beta}\psi^{i}(2^{j}x-k)\|_{B(x_{0},2^{-J})}.$$

Indeed, if  $x \in B(x_0, 2^{-J})$ , we have

$$|D^{\beta}f_{j}(x)| \leq \sum_{i} \sum_{k \in \mathbb{Z}^{d}} |c_{j,k}^{i}| 2^{j(M+1)} |D^{\beta}\psi^{i}(2^{j}x-k)|$$
  
= 
$$\sum_{i} \sum_{\substack{k \in \mathbb{Z}^{d} \\ k2^{-j} \in B(x,2^{-(j-j_{0})})}} |c_{j,k}^{i}| 2^{j(M+1)} |D^{\beta}\psi^{i}(2^{j}x-k)|.$$

Each wavelet coefficient  $c_{j,k}^i = c_{\lambda}$  in the last sum is such that the associated dyadic cube is included in  $B(x, 2^{-(j-j_0-n_d)})$ . If  $j \ge j_0 + n_d + m_d + 1$ , then we have

$$|c_{j,k}^i| \le C\sigma_{j-j_0-n_d-m_d-1} \le C\sigma_j.$$

Otherwise, we have  $|c_{j,k}^i| \leq C \leq C'\sigma_j$  because the function f is uniformly Hölder, where  $C' = \sup\{C/\sigma_j : j \in \{0, ..., j_0 + n_d + m_d\}\}$ . This proves the announced inequality. This also implies

$$\|D^{\beta}f_{j}\|_{B(x_{0},2^{-J})} \leq C\sigma_{j}2^{j(M+1)},$$

#### A CHARACTERIZATION IN TERMS OF WAVELET COEFFICIENTS

because  $D^{\beta}\psi^{i}$  has compact support and is continuous on  $\mathbb{R}^{d}$ . We obtain

$$\sum_{j=-1}^{J} \|f_j(x) - \sum_{|\beta| \le M} \frac{(x-x_0)^{\beta}}{|\beta|!} D^{\beta} f_j(x_0) \|_{B(x_0, 2^{-J})} \le C 2^{-J(M+1)} \sum_{j=-1}^{J} \sigma_j 2^{j(M+1)} \le C \sigma_J.$$

2. We consider the term

$$\sum_{j=J_1+1}^{+\infty} \|f_j\|_{B(x_0,2^{-J})}.$$

Using the proof of theorem 115, we have

$$\sum_{j=J_1+1}^{+\infty} \|f_j\|_{B(x_0,2^{-J})} \leq \sum_{j=J_1+1}^{+\infty} \|f_j\|_{\mathbb{R}^d}$$
$$\leq C \sum_{j=J_1+1}^{+\infty} 2^{-\varepsilon j}$$
$$\leq C 2^{-\varepsilon J_1}$$
$$\leq C \sigma_J.$$

3. Let us consider the term given by

$$\sum_{j=J+1}^{J_1} \|f_j\|_{B(x_0,2^{-J})}.$$

Let  $j \in \{J + 1, ..., J_1\}$  and  $x \in B(x_0, 2^{-J})$ . We have

$$|f_j(x)| \le \sum_i \sum_{\substack{k \in \mathbb{Z}^d \\ k2^{-j} \in B(x, 2^{-(j-j_0)})}} |c_{j,k}^i \psi^i (2^j x - k)|.$$
(4.4)

If  $j \ge J + j_0 + n_d$ , then the wavelet coefficients in (4.4) are such that the associated dyadic cubes satisfy

$$\lambda = \lambda(i, j, k) \subseteq B(x, 2^{-(j-j_0-n_d)}) \subseteq B(x_0, 2^{-(J-1)})$$

and

$$|c_{j,k}^i| = |c_{\lambda}| \le C\sigma_{J-m_d-1} \le C\sigma_J$$

If  $j < J + j_0 + n_d$ , then we have

$$\lambda(i,j,k) \subseteq B(x,2^{-(j-j_0-n_d)}) \subseteq B(x_0,2^{-(j-j_0-n_d-1)}),$$

which implies

$$|c_{j,k}^i| = |c_{\lambda}| \le C\sigma_{j-j_0-m_d-m_d-1} \le C\sigma_j \le C\sigma_J.$$

Finally, we find

$$\sum_{j=J+1}^{J_1} \|f_j\|_{B(x_0, 2^{-J})} \le CJ_1 \sigma_J \le C |\log_2(\sigma_J)| \sigma_J,$$

which proves the result.

**Remark 170.** Because  $\sigma'_j \to 0$  if  $\sigma_j \to 0$ , the previous result always gives information on the pointwise regularity of f at  $x_0$  (under the assumptions of the previous theorem).

## 4.5 A characterization of pointwise generalized Hölder-Zygmund spaces in terms of the convolution product

We have seen in section 2.4 that global generalized Hölder-Zygmund spaces can be characterized through approximations with a convolution product of their own elements with a smooth function. The aim of this section is to prove a similar characterization for pointwise spaces.

We have the following result, using the same proof as in lemma 78.

**Lemma 171.** Let  $N \in \mathbb{N}_0$ ,  $\rho \in D(\mathbb{R}^d)$ ,  $\sigma = (\sigma_j)_{j \in \mathbb{N}^*}$  be an admissible sequence and  $f \in L^1_{loc}(\mathbb{R}^d)$  be a function satisfying

$$\sup_{k\geq j} \|f\star\rho_{2^{-k}} - f\|_{L^{\infty}(B(x_0,2^{-j}))} \leq C\sigma_j \quad \forall j\in\mathbb{N}_0.$$

For all  $\beta \in \mathbb{N}_0^d$  such that  $|\beta| \leq N$ , we have

$$\|D^{\beta} (f \star \rho_{2^{-j}} - f \star \rho_{2^{-(j-1)}})\|_{B(x_0, 2^{-j})} \le C 2^{jN} \sigma_j \qquad \forall j \in \mathbb{N}^*.$$

The characterization of pointwise Hölder spaces in terms of the convolution product can be written in the following way (the notation  $\Phi_i$  refers to  $\Phi_{2^{-j}}$ ).

**Theorem 172.** (D.K., S. Nicolay) Let  $M \in \mathbb{N}_0$  and  $\sigma$  be an admissible sequence. If  $f \in \Lambda^{\sigma,M}(x_0)$ , then there exists a function  $\Phi \in D(\mathbb{R}^d)$  such that

$$\sup_{k \ge j} \|f - f \star \Phi_k\|_{B(x_0, 2^{-j})} \le C\sigma_j, \quad \forall j \in \mathbb{N}_0.$$

$$(4.5)$$

Conversely, let us suppose that  $\sigma \to 0$ . If the function  $f \in L^{\infty}(\mathbb{R}^d)$  satisfies (4.5) and

$$\sup_{j\in\mathbb{N}_0} \left( 2^{\alpha j} \sup_{k\geq j} \|f \star \Phi_k - f\|_{L^{\infty}} \right) < +\infty$$

for a certain  $\alpha > 0$ , then  $f \in \Lambda^{\sigma,M}(x_0)$  for all natural number  $M \in \mathbb{N}_0$  such that  $M + 1 > \overline{s}(\sigma^{-1})$ .

*Proof.* Let  $f \in \Lambda^{\sigma,M}(x_0)$  and  $\Phi$  be the function defined in section 2.2. Using a similar proof as in proposition 77, we find

$$f \star \Phi_k(x) - f(x) = C \int \Delta_{2^{-k_t}}^{M'+1} f(x) \rho(t) dt, \quad \forall x \in \mathbb{R}^d,$$

for a natural number  $M' \geq M$ . This proves equality (4.5). Let us prove the converse part. By assumption, there exists  $\alpha < 1$  such that  $f \in \Lambda^{\alpha}(\mathbb{R}^d)$ . If  $(f_j)_j$  denotes the sequence of functions which are defined in proposition 80, we have

$$f = \sum_{j=1}^{+\infty} f_j$$
 uniformly on  $\mathbb{R}^d$ ,

because f is uniformly Hölder. So, we also have

$$\Delta_h^{M+1} f = \sum_{j=1}^{+\infty} \Delta_h^{M+1} f_j \quad \text{uniformly on } \mathbb{R}^d, \text{ for all } h \in \mathbb{R}^d.$$

Let  $n_0 \in \mathbb{N}_0$  such that  $M + 1 \leq 2^{n_0}$ ,  $h \in \mathbb{R}^d$  such that  $|h| \leq 2^{-(j+n_0)}$ , and  $j_0 \in \mathbb{N}_0$  such that  $2^{-(j_0+1)\alpha} \leq \sigma_j < 2^{-j_0\alpha}$ . We have

$$\begin{split} \|\Delta_h^{M+1} f\|_{B(x_0, 2^{-j})} &\leq \sum_{k=1}^{j-1} \|\Delta_h^{M+1} f_k\|_{B(x_0, 2^{-j})} + \|\sum_{k=j}^{j_0} \Delta_h^{M+1} f_k\|_{B(x_0, 2^{-j})} \\ &+ \sum_{k=j_0+1}^{+\infty} \|\Delta_h^{M+1} f_k\|_{B(x_0, 2^{-j})}. \end{split}$$

1. We find

$$\sum_{k=1}^{j-1} \|\Delta_h^{M+1} f_k\|_{B(x_0, 2^{-j})} \le \sum_{k=1}^{j-1} C|h|^{M+1} \sup_{|\beta|=M+1} \|D^{\beta} f_k\|_{B(x_0, 2^{-(j-1)})}$$
$$\le C 2^{-j(M+1)} \sum_{k=1}^{j-1} 2^{k(M+1)} \sigma_k$$
$$\le C \sigma_j,$$

using the mean value theorem for the first inequality, lemma 171 for the second one, and the assumption  $M + 1 > \overline{s}(\sigma^{-1})$  in the last one.

2. We have

$$\begin{split} \| \sum_{k=j}^{j_0} \Delta_h^{M+1} f_k \|_{B(x_0, 2^{-j})} &= \| \Delta_h^{M+1} (f \star \Phi_{j_0} - f \star \Phi_{j-1}) \|_{B(x_0, 2^{-j})} \\ &\leq C \| f \star \Phi_{j_0} - f \star \Phi_{j-1}) \|_{B(x_0, 2^{-(j-1)})} \\ &\leq C \left( \| f \star \Phi_{j_0} - f \|_{\mathbb{R}^d} + \| f - f \star \Phi_{j-1}) \|_{B(x_0, 2^{-(j-1)})} \right) \\ &\leq C (2^{-j_0 \alpha} + \sigma_{j-1}) \leq C \sigma_j. \end{split}$$

3. Finally, as  $f \in \Lambda^{\alpha}(\mathbb{R}^d)$ , we have

$$\sum_{k=j_0+1}^{+\infty} \|\Delta_h^{M+1} f_k\|_{B(x_0,2^{-j})} \le C \sum_{\substack{k=j_0+1}}^{+\infty} \|f_k\|_{\mathbb{R}^d}$$
$$\le C \sum_{\substack{k=j_0+1}}^{+\infty} 2^{-k\alpha}$$
$$\le C 2^{-j_0\alpha} \le C\sigma_j.$$

The conclusion follows from

$$\sup_{|h| \le 2^{-(j+n_0)}} \|\Delta_h^{M+1} f\|_{B(x_0, 2^{-(j+n_0)})} \le \sup_{|h| \le 2^{-(j+n_0)}} \|\Delta_h^{M+1} f\|_{B(x_0, 2^{-j})} \le C\sigma_j \le C\sigma_{j+n_0}.$$

**Remark 173.** The two following results, which are slight modifications of lemma 171 and theorem 172, can be proved similarly.

**Lemma 174.** Let  $N \in \mathbb{N}_0$ ,  $\rho \in D(\mathbb{R}^d)$  be a function such that its support is included in the ball B(0, 1/4),  $\sigma = (\sigma_j)_{j \in \mathbb{N}^*}$  be an admissible sequence and  $f \in L^1_{loc}(\mathbb{R}^d)$  be a function satisfying

 $\|f \star \rho_{2^{-j}} - f\|_{L^{\infty}(B(x_0, 2^{-j}))} \le C\sigma_j \quad \forall j \in \mathbb{N}^*.$ 

Then, for all  $\beta \in \mathbb{N}_0^d$  such that  $|\beta| \leq N$ , we have

$$\|D^{\beta} (f \star \rho_{2^{-j}} - f \star \rho_{2^{-(j-1)}})\|_{B(x_0, 2^{-(j+1)})} \le C 2^{jN} \sigma_j \qquad \forall j \in \mathbb{N}^*.$$

**Theorem 175.** Let  $M \in \mathbb{N}_0$  and  $\sigma$  be an admissible sequence. If  $f \in \Lambda^{\sigma,M}(x_0)$ , then there exists a function  $\Phi \in D(\mathbb{R}^d)$  whose support is included in the ball B(0, 1/4) and such that

$$\|f - f \star \Phi_j\|_{B(x_0, 2^{-j})} \le C\sigma_j, \quad \forall j \in \mathbb{N}_0.$$

$$(4.6)$$

Conversely, let us suppose that  $\sigma \to 0$ . If the function  $f \in L^{\infty}(\mathbb{R}^d)$  satisfies (4.6) and

$$\sup_{j\in\mathbb{N}_0} \left( 2^{\alpha j} \sup_{k\geq j} \|f \star \Phi_k - f\|_{L^{\infty}} \right) < +\infty$$

for some  $\alpha > 0$  (for some function  $\Phi \in D(\mathbb{R}^d)$  with compact support included in the ball B(0, 1/4)), then  $f \in \Lambda^{\sigma, M}(x_0)$  for all  $M \in \mathbb{N}_0$  such that  $M + 1 > \overline{s}(\sigma^{-1})$ .

### 4.6 An application of generalized Hölder-Zygmund spaces: the Takagi function

In 1903, T. Takagi published an example of a continuous but nowhere differentiable function ([115]). Other authors rediscovered this function later under different forms, which inspired and fascinated many mathematicians (this is still the case nowadays). This function has many singular properties. As the Weierstrass function, the Takagi function is continuous but nowhere differentiable. It is used as a tool in many mathematical areas, such as classical real analysis, multifractal analysis, combinatorics and number theory. For example, let us cite its use in the characterization of zero sets of continuous nowhere differentiable functions ([87, 107]) and its use as a key element for solving the binary digital sum problem ([38, 123]).

Let us recall its formal definition.

**Definition 176.** The *Takagi function* is defined by

$$T(x) = \sum_{n=0}^{+\infty} \frac{1}{2^n} \phi(2^n x), \quad x \in [0, 1],$$

where  $\phi(x) = \operatorname{dist}(x, \mathbb{Z})$ . Figure 4.2 represents the graphic of T.

The goal of this section is to determine the global and the pointwise regularity of the Takagi function.



Figure 4.2: Graphic of the Takagi function T defined on [0, 1].

It is known that  $T \in \Lambda^{\alpha}(\mathbb{R})$  for  $\alpha \in ]0,1[$  but  $T \notin \Lambda^{\alpha,(2^{-j})_j}(\mathbb{R})$  because T is not Lipschitz ([110]). However, there exist some points  $x \in [0,1]$  called *slow points* such that  $T \in \Lambda^{\alpha,(2^{-j})_j}(x)$  for  $\alpha \in ]0,1[$  ([1]). In [10], it is shown that  $T \in \Lambda^{\alpha,\sigma}(\mathbb{R})$  where  $\sigma_j = 2^{-j}j$  $(j \in \mathbb{N}^*)$  and  $T \notin \Lambda^1(\mathbb{R})$  for  $\alpha \in ]0,1[$ . Moreover, the sequence  $\sigma$  is the best estimate for Tto be an element of a generalized Hölder-Zygmund space.

This example shows that generalized Hölder-Zygmund spaces can provide interesting additional information besides classical Hölder-Zygmund spaces. It also shows that pointwise generalized Hölder-Zygmund spaces complete the study of the global regularity in some cases.

A global survey about the Takagi function is done in [5].

#### 4.7 Generalized pointwise Hölder exponent

The goal of this section is to give some conditions on admissible sequences that lead to embedded generalized pointwise Hölder spaces. If we have embedded pointwise spaces, we can define an Hölder exponent which gives information on the pointwise regularity at  $x_0$ . Our aim is to prove an analogous result to proposition 132 obtained in section 3.1. However, we need to completely change the ideas behind the proofs.

Notation 177. Let  $M \in \mathbb{N}_0$ . Let  $P^{(M)}$  denote the part of degree less or equal to M of a polynomial P.

First, let us consider the following result.

**Proposition 178.** Let  $M \in \mathbb{N}^*$  and  $\sigma$  be an admissible sequence such that  $\underline{s}(\sigma^{-1}) > M$ . We have

$$\Lambda^{\sigma,M}(x_0) \subseteq \Lambda^{(2^{-jM})_j,M-1}(x_0).$$

*Proof.* Let  $P_{x_0}$  be the polynomial given by theorem 162. It is given by

$$P_{x_0}(x) = \sum_{|\beta| \le M} D^{\beta} f(x_0) \frac{(x - x_0)^{\beta}}{|\beta|!}.$$

Let us note that

$$\left| f(x_0+h) - \sum_{|\beta| \le M-1} D^{\beta} f(x_0) \frac{h^{\beta}}{|\beta|!} \right| \le C\sigma_j + \left| \sum_{|\beta|=M} D^{\beta} f(x_0) \frac{h^{\beta}}{|\beta|!} \right|$$

for all  $|h| \leq 2^{-j}$ . We have

$$\inf_{P \in \mathbb{P}_{M-1}} \|f - P\|_{B(x_0, 2^{-j})} \le \|f - P_{x_0}^{(M-1)}\|_{B(x_0, 2^{-j})} \le C(\sigma_j + 2^{-jM}) \le C' 2^{-jM},$$

because  $2^{jM}\sigma_j \to 0$ .

<sup>&</sup>lt;sup>4</sup>We can consider that the Takagi function is defined on  $\mathbb{R}$  by multiplying it with an adequate regular function with compact support. For example, one can use  $\rho \in D(\mathbb{R})$  such that  $\rho = 1$  on  $[\varepsilon, 1 - \varepsilon]$  (where  $\varepsilon > 0$  is sufficiently small) and equal to 0 outside of ]0, 1[).

This result is interesting but does not give enough accurate information. This is a consequence of the assumption expressed in terms of Boyd indices among other things<sup>5</sup>. The two following results give more information than proposition 178.

**Proposition 179.** Let  $f \in \Lambda^{\sigma,M}(x_0)$  and  $(P_{x_0,j})_{j\in\mathbb{N}_0}$  be a sequence of polynomials associated with the function f (given by the definition of the space  $\Lambda^{\sigma,M}(x_0)$ ).

1. If  $2^M d_1 < 1$ , then

$$\|f - P_{x_0,j}^{(M-1)}\|_{B(x_0,2^{-j})} \le C\left(\sigma_j + 2^{-jM}\right) \quad \forall j \in \mathbb{N}_0.$$

2. If  $2^M d_1 > 1$ , then

$$|f - P_{x_0,j}^{(M-1)}||_{B(x_0,2^{-j})} \le C \left(\sigma_j + 2^{-jM} (2^M d_1)^j\right) \quad \forall j \in \mathbb{N}_0.$$

3. If  $2^M d_1 = 1$ , then

$$||f - P_{x_{0,j}}^{(M-1)}||_{B(x_{0},2^{-j})} \le C(\sigma_{j} + 2^{-jM}j) \quad \forall j \in \mathbb{N}_{0}$$

**Proposition 180.** Let  $f \in \Lambda^{\sigma,M}(x_0)$  and  $(P_{x_0,j})_{j\in\mathbb{N}_0}$  be a sequence of polynomials associated to the function f (given by the definition of the space  $\Lambda^{\sigma,M}(x_0)$ ).

- 1. If  $2^M d_0 < 1$ , then  $\|f - P_{x_0,j}^{(M-1)}\|_{B(x_0,2^{-j})} \le C \left(\sigma_j (2^M d_0)^{-j} + 2^{-jM}\right) \quad \forall j \in \mathbb{N}_0.$
- 2. If  $2^M d_0 > 1$ , then

$$||f - P_{x_0,j}^{(M-1)}||_{B(x_0,2^{-j})} \le C(\sigma_j + 2^{-jM}) \quad \forall j \in \mathbb{N}_0.$$

3. If  $2^M d_0 = 1$ , then

$$\|f - P_{x_0,j}^{(M-1)}\|_{B(x_0,2^{-j})} \le C\left(\sigma_j j + 2^{-jM}\right) \quad \forall j \in \mathbb{N}_0.$$

These two results are mainly a consequence of the following lemma:

**Lemma 181.** Let  $f \in \Lambda^{\sigma,M}(x_0)$  and  $(P_{x_0,j})_{j\in\mathbb{N}_0}$  be the sequence of polynomials associated to the function f. If  $a_{x_0,j}^{\beta}$  denotes the  $\beta^{th}$  coefficient of the polynomial  $P_{x_0,j}$ , then we have

$$\sup_{|\beta|=M} |a_{x_0,j}^{\beta}| \le C\left(\sum_{k=1}^{j-1} (2^M d_1)^k + 1\right), \quad \forall j \in \mathbb{N}_0,$$

and

$$\sup_{\beta|=M} |a_{x_0,j}^{\beta}| \le C\left(\sigma_j d_0^{-j} \sum_{k=1}^{j-1} (2^M d_0)^k + 1\right), \quad \forall j \in \mathbb{N}_0.$$

<sup>&</sup>lt;sup>5</sup>Let us remind that in section 3.1, we obtained two types of results for global spaces: the first one is expressed in terms of Boyd indices and the second one is expressed in terms of the factors  $d_0$  and  $d_1$  (given by definition of admissible sequences). The first one is a weaker result.

Proof of lemma 181. For all  $|\beta| \leq M$ , we have, using Markov's inequality (4.2),

$$\begin{split} \|D^{\beta}(P_{x_{0},j} - P_{x_{0},j+1})\|_{B(x_{0},2^{-(j+1)})} \\ &\leq C2^{jM} \|P_{x_{0},j} - P_{x_{0},j+1}\|_{B(x_{0},2^{-(j+1)})} \\ &\leq C2^{jM} \left( \|P_{x_{0},j} - f\|_{B(x_{0},2^{-j})} + \|P_{x_{0},j+1} - f\|_{B(x_{0},2^{-(j+1)})} \right) \\ &\leq C2^{jM} \sigma_{j}, \end{split}$$

which implies

$$\begin{split} \|D^{\beta}(P_{x_{0},1} - P_{x_{0},j})\|_{B(x_{0},2^{-j})} \\ &\leq \sum_{k=1}^{j-1} \|D^{\beta}(P_{x_{0},k} - P_{x_{0},k+1})\|_{B(x_{0},2^{-j})} \\ &\leq \sum_{k=1}^{j-1} \|D^{\beta}(P_{x_{0},k} - P_{x_{0},k+1})\|_{B(x_{0},2^{-(k+1)})} \\ &\leq C \sum_{k=1}^{j-1} 2^{kM} \sigma_{k} \leq C \sigma_{0} \sum_{k=1}^{j-1} (2^{M} d_{1})^{k}, \end{split}$$

for all  $j \in \mathbb{N}_0$ . Let  $\beta \in \mathbb{N}_0^d$  such that  $|\beta| = M$ . We have

$$\begin{split} \|D^{\beta}(P_{x_{0},j} - P_{x_{0},1})\|_{B(x_{0},2^{-j})} &= \beta! |a_{x_{0},j}^{\beta} - a_{x_{0},1}^{\beta}| \\ &\geq |a_{x_{0},j}^{\beta}|\beta! - |a_{x_{0},1}^{\beta}|\beta!. \end{split}$$

We thus get the first inequality of the result. To prove the second inequality, we notice that

$$\|D^{\beta}(P_{x_{0},1}-P_{x_{0},j})\|_{B(x_{0},2^{-j})} \le C \sum_{k=1}^{j-1} 2^{kM} \sigma_{k} \le C \sigma_{j} d_{0}^{-j} \sum_{k=1}^{j-1} (2^{M} d_{0})^{k}.$$

Proposition 179 is a consequence of the first inequality of lemma 181. Proposition 180 is a consequence of the second inequality of lemma 181.

**Definition 182.** Let  $x_0 \in \mathbb{R}^d$ . A family of admissible sequences  $\sigma^{(.)}$  is said to be  $x_0$ decreasing if  $\alpha < \beta$  implies  $\Lambda^{\sigma^{(\beta)}, \lfloor \beta \rfloor}(x_0) \subseteq \Lambda^{\sigma^{(\alpha)}, \lfloor \alpha \rfloor}(x_0)$ .

Corollary 183. (D.K., S. Nicolay) A family of admissible sequences  $\sigma^{(.)}$  is  $x_0$ -decreasing if it satisfies the following conditions:

1. if  $m \leq \alpha < \beta < m + 1$  for some natural number m, there exist C, J > 0 such that

$$\sigma_j^{(\beta)} \le C \sigma_j^{(\alpha)} \quad \forall j \ge J,$$

#### GENERALIZED POINTWISE HÖLDER EXPONENT

- 2. for all  $m \in \mathbb{N}^*$ , at least one of the two following conditions is satisfied:
  - (a) there exists  $\varepsilon_0 > 0$  such that for all  $\varepsilon \in ]0, \varepsilon_0[$ , there exist C, J > 0 such that

$$\sigma_j^{(m)} \le C \sigma_j^{(m-\varepsilon)} \quad \forall j \ge J,$$

• if  $1 < 2^m d_1^{(m)}$ , then there exists  $\varepsilon_0 > 0$  such that for all  $\varepsilon \in ]0, \varepsilon_0[$ , there exist C, J > 0 such that

$$2^{-jm}(2^m d_1^{(m)})^j \le C\sigma_j^{(m-\varepsilon)} \quad \forall j \ge J,$$

• if  $1 > 2^m d_1^{(m)}$ , then there exists  $\varepsilon_0 > 0$  such that for all  $\varepsilon \in ]0, \varepsilon_0[$ , there exist C, J > 0 such that

$$2^{-jm} \le C\sigma_j^{(m-\varepsilon)} \quad \forall j \ge J,$$

• if  $1 = 2^m d_1^{(m)}$ , then there exists  $\varepsilon_0 > 0$  such that for all  $\varepsilon \in ]0, \varepsilon_0[$ , there exist C, J > 0 such that

$$j2^{-jm} \le C\sigma_j^{(m-\varepsilon)} \quad \forall j \ge J,$$

(b) • there exists  $\varepsilon_0 > 0$  such that for all  $\varepsilon \in ]0, \varepsilon_0[$ , there exist C, J > 0 such that

$$2^{-jm} \le C\sigma_j^{(m-\varepsilon)} \quad \forall j \ge J,$$

• if  $1 < 2^m d_0^{(m)}$ , then there exists  $\varepsilon_0 > 0$  such that for all  $\varepsilon \in ]0, \varepsilon_0[$ , there exist C, J > 0 such that

$$\sigma_j^{(m)} \le C \sigma_j^{(m-\varepsilon)} \quad \forall j \ge J,$$

• if  $1 > 2^m d_0^{(m)}$ , then there exists  $\varepsilon_0 > 0$  such that for all  $\varepsilon \in ]0, \varepsilon_0[$ , there exist C, J > 0 such that

$$\sigma_j^{(m)} (2^m d_0^{(m)})^{-j} \le C \sigma_j^{(m-\varepsilon)} \quad \forall j \ge J,$$

• if  $1 = 2^m d_0^{(m)}$ , then there exists  $\varepsilon_0 > 0$  such that for all  $\varepsilon \in ]0, \varepsilon_0[$ , there exist C, J > 0 such that

$$j\sigma_j^{(m)} \le C\sigma_j^{(m-\varepsilon)} \quad \forall j \ge J,$$

(where we let  $d_0^{(m)}$  and  $d_1^{(m)}$  denote some constants satisfying the inequality (1.2) associated to  $\sigma^{(m)}$ ).

**Remark 184.** The conditions of corollary 183 imply that, for all  $m \in \mathbb{N}^*$ , there exists  $\varepsilon_0 > 0$  such that for all  $\varepsilon \in ]0, \varepsilon_0[$ , we have

$$2^{-jm} \le C_{\varepsilon} \sigma_j^{(m-\varepsilon)}$$
 and  $\sigma_j^{(m)} \le C_{\varepsilon} \sigma_j^{(m-\varepsilon)} \quad \forall j,$ 

for some constant  $C_{\varepsilon} > 0$ . If the family of sequences does not even satisfy these two simplified conditions, then it is not worthwhile checking whether the conditions of the corollary are satisfied.

**Remark 185.** One can easily check that the sufficient conditions described in corollary 183 are equivalent to the ones obtained in proposition 132. It is remarkable that pointwise spaces maintain all main properties of global spaces, even though the proofs need to be completely adapted. One main difference is that we do not have classical derivatives as a tool in the pointwise case.

**Example 186.** In particular, corollary 183 can be used to prove that classical pointwise Hölder spaces are embedded: if  $0 < \alpha < \beta$ , we have

$$\Lambda^{\beta}(x_0) = \Lambda^{(2^{-j\beta})_j, \lfloor\beta\rfloor}(x_0) \subseteq \Lambda^{\alpha}(x_0) = \Lambda^{(2^{-j\alpha})_j, \lfloor\alpha\rfloor}(x_0).$$

As announced in the introduction, the concept of  $x_0$ -decreasing family of admissible sequences leads to the concept of generalized Hölder exponent at  $x_0$ .

**Definition 187.** Let  $\sigma^{(.)}$  be a  $x_0$ -decreasing family of admissible sequences. The generalized Hölder exponent at  $x_0$  associated with  $\sigma^{(.)}$  of a function  $f \in L^{\infty}_{loc}(\mathbb{R}^d)$  is defined by

$$h_f^{\sigma^{(.)}}(x_0) = \sup\{\alpha > 0 : f \in \Lambda^{\sigma^{(\alpha)}, \lfloor \alpha \rfloor}(x_0)\}.$$

## Chapter 5

## Appendix

#### 5.1About subadditive sequences

**Definition 188.** A sequence  $(a_n)_{n \in \mathbb{N}^*}$  of real numbers is said to be *subadditive* if it satisfies the inequality

$$a_{n+m} \leq a_n + a_m \qquad \forall n, m \in \mathbb{N}^*.$$

The following lemma is attributed to M. Fekete ([52]). We show a proof that is relatively simple here below.

**Lemma 189** (Fekete (1923)). For every subadditive sequence  $(a_n)_{n \in \mathbb{N}^*}$ , the limit  $\lim_{n \to +\infty} \frac{a_n}{n}$ exists and is equal to  $\inf_{n \in \mathbb{N}^*} \frac{a_n}{n}$  (the limit can be equal to  $-\infty$ ).

*Proof.* We set  $\gamma = \inf\left(\frac{a_n}{n}\right)$ . Let us at first suppose that  $\gamma > -\infty$  (so  $\gamma$  is a real number). For all  $\varepsilon > 0$ , there exists  $k \in \mathbb{N}_0$  such that  $a_k \leq (\gamma + \varepsilon)k$ . We also have

$$a_{nl} \leq na_l \quad \forall n, l \in \mathbb{N}^*$$

by subadditivity, which implies

$$\frac{a_{nl}}{nl} \leq \frac{a_l}{l} \quad \forall n, l \in \mathbb{N}^* \,.$$

For all  $n \in \mathbb{N}^*$ , if  $l \in \mathbb{N}_0$  is such that n < lk, then we have

$$\inf_{m \ge n} \frac{a_m}{m} \le \frac{a_{lk}}{lk} \le \gamma + \varepsilon$$

which implies  $\gamma = \liminf \frac{a_n}{n}$ . For all  $m \in \mathbb{N}_0$ , we write m = kn + j where  $0 \le j < k$  so that

$$a_m = a_{nk+j} \le a_{kn} + a_j \le (\gamma + \varepsilon)kn + a_j$$

which implies

$$\frac{a_m}{m} \le (\gamma + \varepsilon) \frac{kn}{m} + \sup_{0 \le j \le k} \frac{a_j}{m}.$$

So, we find

$$\sup_{m \ge n'} \frac{a_m}{m} \le \sup_{m \ge n'} \left( (\gamma + \varepsilon) \frac{kn}{m} \right) + \sup_{m \ge n'} \sup_{0 \le j < k} \frac{a_j}{m} \le \sup_{m \ge n'} \left( (\gamma + \varepsilon) \frac{kn}{m} \right) + \sup_{m \ge n'} \sup_{0 \le j < k} \frac{ja_1}{m}$$

Let us prove that  $\lim_{n'\to+\infty} \sup_{m\geq n'} \sup_{0\leq j< k} \frac{ja_1}{m} \leq 0$ . If  $a_1 \leq 0$ , the result is obvious and if  $a_1 > 0$ , this results from

$$\sup_{0 \le j < k} \frac{ja_1}{m} \le \frac{k}{nk} a_1 = \frac{a_1}{n} \to 0 \quad \text{if} \quad n' \to +\infty$$

Moreover, we have

$$\lim_{n'\to+\infty}\sup_{m\geq n'}\left((\gamma+\varepsilon)\frac{kn}{m}\right)\leq\gamma+\varepsilon.$$

Indeed, if  $\gamma + \varepsilon > 0$ , this results immediately from  $\frac{kn}{m} \leq 1$ . If  $\gamma + \varepsilon < 0$ , this results from

$$(\gamma + \varepsilon)\frac{kn}{m} \le (\gamma + \varepsilon)\frac{kn}{k(n+1)} = (\gamma + \varepsilon)\frac{n}{(n+1)} \to (\gamma + \varepsilon) \quad \text{if} \quad n' \to +\infty.$$

We have proved that

$$\limsup \frac{a_m}{m} \le \gamma + \varepsilon = \liminf \frac{a_m}{m} + \varepsilon.$$

Now, let us consider the case  $\gamma = -\infty$ . Let us prove that

$$\limsup \frac{a_n}{n} = -\infty.$$

For all N > 0, there exists  $k \in \mathbb{N}_0$  such that  $\frac{a_k}{k} \leq -N$ . Applying the same logic as the ideas developed in the first part of the proof, we have

$$\frac{a_{kn}}{kn} \le \frac{a_k}{k} \le -N$$

and if we keep the notation of the decomposition m = kn + j where  $0 \le j < k$ , we have

$$\sup_{m \ge n'} \frac{a_m}{m} \le \sup_{m \ge n'} \left(\frac{a_{nk}}{m}\right) + \sup_{m \ge n'} \left(\frac{a_j}{m}\right)$$
$$\le \sup_{m \ge n'} \left(\frac{-Nkn}{m}\right) + \sup_{m \ge n'} \left(j\frac{a_1}{m}\right)$$
$$\le \sup_{m \ge n'} \left(\frac{-Nn}{n+1}\right) + \sup_{m \ge n'} \left(j\frac{a_1}{m}\right).$$

If we separate the case  $a_1 \ge 0$  from the case  $a_1 < 0$ , we conclude that the second term of this last inequality converges to 0 if n' goes to infinity. Hence the conclusion follows.  $\Box$ 

An analogous result to this lemma is true for the superadditives sequences, i.e. the sequences satisfying  $a_{n+m} \ge a_n + a_m \quad \forall m, n \in \mathbb{N}^*$ .

#### 5.2 Some inequalities from S. Bernstein

**Proposition 190.** Let  $k \in \mathbb{N}_0$  and  $R_1$ ,  $R_2 \in \mathbb{R}$  such that  $0 < R_1 < R_2$ . There exists a positive constant C (which only depends on  $R_1$ ,  $R_2$ , k and d) such that for every function  $u \in L^{\infty}(\mathbb{R}^d)$ , we have

$$supp\mathcal{F}u \subseteq B(0, R_1\lambda) \Rightarrow \sup_{|\alpha|=k} \|D^{\alpha}u\|_{L^{\infty}} \le C\lambda^k \|u\|_{L^{\infty}},$$
(5.1)

 $supp\mathcal{F}u \subseteq B(0, \leq R_2\lambda) \setminus B(0, R_1\lambda) \Rightarrow C^{-1}\lambda^k \|u\|_{L^{\infty}} \leq \sup_{|\alpha|=k} \|D^{\alpha}u\|_{L^{\infty}} \leq C\lambda^k \|u\|_{L^{\infty}}.$  (5.2)

#### 5.3 About the Littlewood-Paley decomposition

The aim of this section is to prove certain results presented in section 2.8.

Proposition 191. We have

$$Id = S_0 + \Delta_0 + \Delta_1 + \dots,$$

with convergence in  $\mathcal{S}'(\mathbb{R}^d)$ .

*Proof.* Let  $u \in \mathcal{S}'(\mathbb{R}^d)$  and  $f \in \mathcal{S}(\mathbb{R}^d)$ . We have

$$(S_0 u + \sum_{j=0}^{N} \Delta_j u)(f) = S_{N+1} u(f)$$
  
=  $\mathcal{F}^{-1}(\hat{\varphi}(2^{-(N+1)}\xi)\mathcal{F}u)(f)$   
=  $u(\mathcal{F}(\hat{\varphi}(2^{-(N+1)}\xi)\mathcal{F}^{-1}f))$   
=  $u(\mathcal{F}^{-1}(\hat{\varphi}(-2^{-(N+1)}\xi)\mathcal{F}f)).$ 

Let us prove that we have

$$\mathcal{F}^{-1}(\hat{\varphi}(-2^{-(N+1)}\xi)\mathcal{F}f) \to f \text{ in } \mathcal{S}(\mathbb{R}^d)$$

which is equivalent to

$$\hat{\varphi}(-2^{-(N+1)}\xi)\mathcal{F}f \to \mathcal{F}f \quad \text{in } \mathcal{S}(\mathbb{R}^d).$$

Let  $k, M \in \mathbb{N}_0$ . It is sufficient to prove that

$$\sup_{x \in \mathbb{R}^d} (1+|x|)^M |D^{\alpha} \left( (\hat{\varphi}(-2^{-(N+1)}x) - 1)\mathcal{F}f \right)(x) | \longrightarrow 0$$

for all  $|\alpha| = k$ . Let  $0 \neq \beta \leq \alpha$ . We have

$$\begin{split} \sup_{x \in \mathbb{R}^d} (1+|x|)^M |D^{\beta}(\hat{\varphi}(-2^{-(N+1)}x)-1)D^{\alpha-\beta}\mathcal{F}f(x)| \\ &= \sup_{x \in \mathbb{R}^d} (1+|x|)^M |D^{\beta}\hat{\varphi}(-2^{-(N+1)}x)2^{-(N+1)|\beta|}D^{\alpha-\beta}\mathcal{F}f(x)| \\ &\leq C2^{-(N+1)|\beta|} \sup_{x \in \mathbb{R}^d} (1+|x|)^M |D^{\alpha-\beta}\mathcal{F}f(x)|. \end{split}$$

Moreover,

$$\sup_{x \in \mathbb{R}^d} (1+|x|)^M |(\hat{\varphi}(-2^{-(N+1)}x)-1)D^{\alpha}\mathcal{F}f(x)| \\ \leq C \sup_{|x| \ge 2^N} (1+|x|)^M |D^{\alpha}\mathcal{F}f(x)| \longrightarrow 0 \quad \text{as } N \to +\infty.$$

By using the Leibniz formula, we obtain the desired result.

**Lemma 192.** If  $f \in L^p(\mathbb{R}^d)$  where  $p \in [1, +\infty]$ , then the functions  $S_j(f)$  and  $\Delta_j(f)$  belong to the space  $L^p(\mathbb{R}^d)$  and

$$S_j(f) = 2^{jd}\varphi(2^j \cdot) \star f$$
 and  $\Delta_j(f) = 2^{jd}\psi(2^j \cdot) \star f$ 

for all  $j \in \mathbb{Z}$ .

*Proof.* Let us prove the result for  $S_j(f)$  (the proof can easily be adapted to  $\Delta_j(f)$ ). Since the function  $2^{jd}\varphi(2^j\cdot)$  belongs to the Schwartz space, it belongs in particular to the spaces  $L^q(\mathbb{R}^d)$  for all  $q \in [1, +\infty]$ . Using Hausdorff-Young inequalities, the function defined by  $2^{jd}\varphi(2^j\cdot) \star f$  exists, belongs to  $L^p(\mathbb{R}^d)$  and, by a classical property of the Fourier transform of tempered distributions, it satisfies

$$\mathcal{F}(2^{jd}\varphi(2^j\cdot)\star f) = \mathcal{F}(2^{jd}\varphi(2^j\cdot))\mathcal{F}f = \hat{\varphi}(2^{-j}\xi)\mathcal{F}f.$$

#### 5.4 Some reminders about wavelets

In this section, we recall the concept of wavelets as well as some of their basic properties.

The wavelet bases in  $L^2(\mathbb{R}^d)$  can be obtained from a function  $\phi$  and from  $2^d - 1$  functions  $\psi^1, ..., \psi^{2^d-1}$ , which are all assumed to be sufficiently regular. The function  $\phi$  is often called the scaling function or "father" wavelet, and the functions  $\psi^i$  are often called the "mother" wavelets. We define the following functions by translations and dilatations of functions  $\phi$  and  $\psi^i$ :

$$\phi_k(x) = \phi(x-k), \quad \psi_{j,k}^i(x) = \psi^i(2^j x - k)$$

for all  $i \in \{1, ..., 2^d - 1\}, j \in \mathbb{N}_0, k \in \mathbb{Z}^d$ .

Under some conditions, the set  $\{\phi_k\} \cup \{2^{jd/2}\psi_{j,k}^i\}_{i,j,k}$  defines an orthonormal basis of  $L^2(\mathbb{R}^d)$  and forms what is called a *multiresolution analysis* or a *multiresolution approxima*tion ([37, 93]). We recall this concept here below:

**Definition 193.** A sequence  $(V_j)_{j \in \mathbb{Z}}$  of closed linear subspaces of  $L^2(\mathbb{R}^d)$  is a multiresolution analysis if the six following properties are satisfied:

1. the space  $V_j$  is invariant by translation proportional to the scale  $2^{-j}$ : for all  $j \in \mathbb{Z}$ ,  $k \in \mathbb{Z}^d$ , the function f belongs to the space  $V_j$  if and only if  $f(.-2^{-j}k)$  also belongs to the space  $V_j$ ;

#### SOME REMINDERS ABOUT WAVELETS

- 2. for all  $j \in \mathbb{Z}$ , we have  $V_j \subset V_{j+1}$ ;
- 3. for all  $j \in \mathbb{Z}$ , the function f belongs to the space  $V_{j+1}$  if and only if f(./2) belongs to the space  $V_j$ ;
- 4. we have

$$\bigcap_{j=-\infty}^{+\infty} V_j = \{0\};$$

5. we have

$$\overline{\bigcup_{j=-\infty}^{+\infty} V_j} = L^2(\mathbb{R}^d);$$

6. there exists a function  $\theta$  such that the sequence  $\{\theta(.-k)\}_{k\in\mathbb{Z}^d}$  is a Riesz basis of the space  $V_0$ , i.e. it is a sequence of elements of  $V_0$  such that there exist  $C_1$ ,  $C_2 > 0$  such that, for every sequence of scalars  $(\alpha_k)_{k\in\mathbb{Z}^d} \in l^2(\mathbb{Z})$ , we have

$$C_1\left(\sum_k |\alpha_k|^2\right)^{1/2} \le \|\sum_k \alpha_k \theta(.-k)\|_{L^2(\mathbb{R}^d)} \le C_2\left(\sum_k |\alpha_k|^2\right)^{1/2}, \quad (5.3)$$

and the vector space of finite sums  $\sum_{k} \alpha_k \theta(.-k)$  (on which the inequality (5.3) is tested) is dense in  $V_0$ .

If the functions  $\phi$  and  $\psi^i$ , which constitute the multiresolution analysis, belong to  $C^r(\mathbb{R}^d)$  and if the derivatives of order less or equal to r are rapidly decreasing, we say that the multiresolution analysis is of regularity r ([64]). We define the *wavelet coefficients* of a function f of  $L^2(\mathbb{R}^d)$  by

$$C_{k} = \int_{\mathbb{R}^{d}} f(x)\phi_{k}(x)dx, \quad c_{j,k}^{i} = 2^{jd} \int_{\mathbb{R}^{d}} f(x)\psi_{j,k}^{i}(x)dx$$
(5.4)

for all  $j \in \mathbb{N}_0, i \in \{1, ..., 2^d - 1\}, k \in \mathbb{Z}^d$ .

Hence, any function f of  $L^2(\mathbb{R}^d)$  can be decomposed in the basis in the following way:

$$f = \sum_{k \in \mathbb{Z}^d} C_k \phi_k + \sum_{i=1}^{2^d - 1} \sum_{j \in \mathbb{N}_0} \sum_{k \in \mathbb{Z}^d} c^i_{j,k} \psi^i_{j,k}.$$
 (5.5)

For more information concerning the multiresolution analysis and the wavelet decomposition, the reader can refer to [37, 89, 93].

Under some classical regularity conditions on the multiresolution analysis, the formulae (5.4) and (5.5) are still true in a more general framework than  $L^2(\mathbb{R}^d)$  ([93]). The formulae should be understood as a dual product between sufficiently regular functions (the wavelets)

and distributions (f in our case). Let us recall that if the multiresolution analysis is of regularity r, the wavelets have a corresponding number of vanishing moments ([93]):

if 
$$|\alpha| < r$$
, we have  $\int_{\mathbb{R}^d} \psi^i(x) x^{\alpha} dx = 0.$ 

Therefore, if the wavelets belong to the Schwartz class, all their moments vanish.

#### 5.5 Some reminders about the interpolation theory

In this section, we recall some of the classical concepts of the interpolation theory.

In the sequel, we consider two Banach spaces  $A_0$  and  $A_1$ , which are continuously embedded in a topological vector space V, so that spaces  $A_0 \cap A_1$  and  $A_0 + A_1$  are well-defined Banach spaces. Let us define the operator J for all t > 0 and  $a \in A_0 \cap A_1$  by

$$J(t,a) = \max\{\|a\|_{A_0}, t\|a\|_{A_1}\}.$$

Let us give the definition of the *J*-method of interpolation.

**Definition 194.** Let  $0 < \theta < 1$  and  $1 \leq q \leq +\infty$ . We define the *interpolation space*  $[A_0, A_1]_{\theta,q,J}$  in the following way: we say that a belongs to  $[A_0, A_1]_{\theta,q,J}$  if a can be written as  $a = \sum_{j \in \mathbb{Z}} u_j$  with convergence in  $A_0 + A_1$ , where  $u_j \in A_0 \cap A_1$  and  $(2^{-j\theta}J(2^j, u_j))_{j \in \mathbb{Z}} \in l^q(\mathbb{Z})$ .

Let us define the operator K for all t > 0 and  $a \in A_0 + A_1$  by

$$K(t,a) = \inf\{\|a_0\|_{A_0} + t\|a_1\|_{A_1} : a = a_0 + a_1\}.$$

Let us give the definition of the K-method of interpolation.

**Definition 195.** Let  $0 < \theta < 1$  and  $1 \le q \le +\infty$ . The interpolation space  $[A_0, A_1]_{\theta,q,K}$  is defined in the following way: we say that a belongs to  $[A_0, A_1]_{\theta,q,K}$  if  $a \in A_0 + A_1$  and  $(2^{-j\theta}K(2^j, a))_{j\in\mathbb{Z}} \in l^q(\mathbb{Z}).$ 

Let us recall the definition of the Sobolev spaces.

**Definition 196.** Let  $p \in [1, +\infty]$  and  $m \in \mathbb{N}_0$ . The Sobolev space  $W_m^p(\mathbb{R}^d)$  is defined by

$$W_m^p(\mathbb{R}^d) = \{ f \in L^p(\mathbb{R}^d) : D^\alpha f \in L^p(\mathbb{R}^d) \quad \forall |\alpha| \le m \}.$$

We define a norm on this space by

$$||f||_{W^p_m} = \sum_{|\alpha| \le m} ||D^{\alpha}f||_{L^p}.$$

The derivatives in this definition should be understood in a weak sense.
**Remark 197.** Even if the Sobolev spaces are defined in terms of derivatives in the weak sense, it is still possible to link them with the classical concept of derivatives up to a certain order. Indeed, the result below links Sobolev spaces and classical Hölder spaces together, and is generally attributed to Morrey ([114, 116]).

**Proposition 198.** Let 
$$p \in [d, +\infty]$$
; we have  $W_1^p(\mathbb{R}^d) \subset \Lambda^{\gamma}(\mathbb{R}^d)$  where  $\gamma = 1 - \frac{d}{p}$ , and

$$|f(x) - f(y)| \le C|x - y|^{\gamma} ||f||_{W_1^p} \quad \forall f \in W_1^p(\mathbb{R}^d).$$

As a consequence, if  $f \in W_m^{\infty}(\mathbb{R}^d)$ , then the function f can be modified on a negligible set so that it belongs to the space  $C^{m-1}(\mathbb{R}^d)$ , and its derivatives  $D^{\alpha}f$  for  $|\alpha| = m - 1$  are differentiable almost everywhere on  $\mathbb{R}^d$ . Moreover, proposition 198 gives a result similar to the mean value theorem for these almost everywhere differentiable functions.

For more information about the classical theory of interpolation spaces, the reader can refer to [17, 86].

## Index

Definition J-Method of interpolation, 100 K-Method of interpolation, 100  $x_0$ -decreasing family of admissible sequences, 92"De la Vallée-Poussin" derivative, 80 Adjacent dyadic cubes, 82 Admissible sequence, 7 Almost increasing sequence, 10 Besov spaces, 5 Boyd indices, 7 Central finite difference, 22 Decreasing family of admissible sequences, 63 Dyadic cube of scale  $j \in \mathbb{N}_0, 82$ Finite difference, 5 Forward finite difference, 5 Generalized J-Method of interpolation, 53Generalized K-Method of interpolation, 53Generalized Besov spaces, 11, 13 Generalized Hölder exponent, 60, 65 Generalized Hölder exponent at  $x_0$ , 94 Generalized Hölder-Zygmund spaces, 17, 18 Generalized interpolation spaces, 53 Generalized pointwise Hölder spaces, 76 Geometric Brownian motion, 71 Hölder exponent, 2 Hölder exponent at  $x_0, 2$ Hölder-Zygmund spaces, 6 Hull and White one-factor model, 72 Interpolation spaces, 100 Irregular Hölder space, 68

Littlewood-Paley decomposition, 43, 97 Local generalized Hölder-Zygmund spaces, 77Modulus of continuity, 17 Multiresolution analysis, 98 Multiresolution approximation, 98 Ornstein-Uhlenbeck stochastic process, 73 Peano's derivative, 80 Pointwise Hölder spaces, 75 Riesz basis, 99 Sequence of bounded growth, 10 Slowly varying function, 9 Sobolev space, 100 Strong admissible sequence of order  $N \in$  $\mathbb{N}^*, 38$ Strongly increasing sequence, 10 Subadditive sequences, 95 Superadditive sequences, 96 Takagi function, 89 Uniform irregularity exponent, 69 Uniformly Hölder function, 83 Upper and lower Boyd indices, 7 Upper global Hölder exponent, 69 Vasicek model, 73 Wavelet coefficients, 99 Wavelet leaders, 82 Weak Hölder space, 69

## Operators

 $H_f, 2$  $H_f^{\sigma^{(.)}}, 60, 65$ J, 53, 100K, 53, 100 $S_j, 43, 97$  $\Delta_h^1, 5, 22$ 

 $\Delta_h^m$ , 5, 22  $\Delta_i, 43, 97$  $\delta_h^1$ , 22  $\underline{\delta_h^m}, 22$  $\overline{H_f}$ , 69  $\overline{\sigma}_j, 7$  $\overline{s}(\sigma), 7$  $\underline{\sigma}_i, 7$  $\underline{s}(\sigma), 7$  $h_f(x_0), 2$  $h_f^{\sigma^{(.)}}(x_0), 94$ Other notations  $3\lambda, 3\lambda_j(x_0), 82$  $B_h(x,r), 35, 77$  $C_k, c^i_{j,k}, 29, 99$  $P^{(M)}, 90$ W(t), 71[a, b], 35 $\binom{m}{j}$ , 22, 30  $\lambda = \lambda(i, j, k), 82$  $\lambda_i(x_0), 82$  $\mathbb{P}_m, 27, 35, 76$  $\phi_k, \, \psi^i_{j,k}, \, 98$  $\rho_{\delta}, \Phi_{\delta}, 27, 29, 86$  $\sigma^{(.)}, 63$  $c_{\lambda}, 82$  $d_0, d_1, 7$  $d_{\lambda}, 82$  $d_i(x_0), 82$ Spaces  $B^{\phi}_{p,q}(\mathbb{R}^d), \, 11$  $B_{p,q}^{\sigma}(\mathbb{R}^d), 13$  $B_{p,q}^{\sigma}(\mathbb{R}^d), 11$  $B_{p,q}^{s}(\mathbb{R}^d), 5$  $C^{\alpha}_w(\mathbb{R}^d), 69$  $I^{\alpha}(\mathbb{R}^d), 68$  $W^{p}_{m}(\mathbb{R}^{d}), 29, 53, 100$  $[A_0, A_1]^{\star}_{\sigma, \psi, J}, 53$  $[A_0, A_1]^{\star}_{\sigma, \psi, K}, 53$  $[A_0, A_1]_{\theta,q,J}, 29, 100$  $[A_0, A_1]_{\theta,q,K}, 29, 100$ 

 $\begin{array}{l} \Lambda^{\alpha}(\mathbb{R}^{d}), \ 6 \\ \Lambda^{\alpha}(x_{0}), \ 75 \\ \Lambda^{\omega,\alpha}(\mathbb{R}^{d}), \ 18 \\ \Lambda^{\sigma,M}(x_{0}), \ 76 \\ \Lambda^{\sigma,\alpha}(\mathbb{R}^{d}), \ 17 \\ \Lambda^{M}_{\sigma,N}(x_{0}), \ 76 \\ \Lambda^{\alpha}_{\sigma,N}(\mathbb{R}^{d}), \ 17 \\ \Lambda^{\sigma,\alpha}_{loc}(x_{0}), \ 77 \\ \Phi(\mathbb{R}^{d}), \ 4 \\ \Phi^{N,J}(\mathbb{R}^{d}), \ 12 \\ \mathcal{B}, \ 11 \\ l^{q,\phi}(E), \ 11 \end{array}$ 

## Bibliography

- [1] S. Abbott, J. M. Anderson, and L. D. Pitt. Slow points for functions in the Zygmund class  $\Lambda_d^*$ . Real Analysis Exchange, 32(1):145–170, 2006-2007.
- [2] D. R. Adams. Function Spaces and Potential Theory. Springer-Verlag Berlin, 2000.
- [3] R. A. Adams. Sobolev Spaces. Academic Press, 1975.
- [4] R. Aebi. Itô's formula for non-smooth functions. Publications of the Research Institute for Mathematical Sciences, 28(4), 1992.
- [5] P. C. Allaart and K. Kawamura. The Takagi function: a survey. *Real Analysis Exchange*, 37(1):1–54, 2011.
- [6] A. Almeida. Wavelet bases in generalized Besov spaces. *Journal of Mathematical Analysis and Applications*, 2005.
- [7] A. Almeida. Wavelet representation in Besov spaces of generalized smoothness. In FSDONA 2004: Function Spaces, Differential Operators and Nonlinear Analysis, pages 7–18, 2005.
- [8] A. Almeida and A. Caetano. Real interpolation of generalized Besov-Hardy spaces and applications. *Journal of Fourier Analysis and Applications*, 17, 2011.
- [9] A. Almeida and P. Hästö. Besov spaces with variable smoothness and integrability. Journal of Functional Analysis, 258:1628–1655, 2010.
- [10] J. M. Anderson and L. D. Pitt. Probabilistic behavior of functions in the Zygmund spaces  $\Lambda^*$  and  $\lambda^*$ . London Mathematical Society, 59(3):558–592, 1989.
- [11] P. Andersson. Characterization of pointwise Hölder regularity. Applied and computational harmonic analysis, 4, 1997.
- [12] A. Arneodo, B. Audit, E.-B. Brodie of Brodie, S. Nicolay, M. Touchon, Y. D'Aubenton-Carafa, M. Huvet, and C. Thermes. Fractals and wavelets: What can we learn on transcription and replication from wavelet-based multifractal analysis of DNA sequences? In Robert A. Meyers, editor, *Encyclopedia of Complexity* and Systems Science, pages 3893–3924. Springer, 2009.

- [13] A. Arneodo and S. Jaffard. L'analyse multifractale des signaux. Image des Mathématiques, 2004.
- [14] A. Ayache and M. S. Taqqu. Multifractional processes with random exponent. Publications Mathemàtiques, 49, 2005.
- [15] S. Banach. Sur l'équation fonctionnelle f(x+y) = f(x) + f(y). Fundamenta Mathematicae, 1:123–124, 1920.
- [16] F. Bastin. Notes du cours d'Analyse III 2ème partie (Université de Liège), 2006.
- [17] J. Bergh and J. L. Löfström. Interpolation spaces: An introduction. Springer-Verlag, 1976.
- [18] O. V. Besov. On a family of function spaces. Embedding theorems and applications. Doklady Akademii Nauk SSSR, 126, 1959.
- [19] O. V. Besov. On a family of functions spaces in connection with embeddings and extensions. *Trudy Matematicheskogo Instituta imeni V. A. Steklova*, 60, 1961.
- [20] O. V. Besov, V. P. Il'in, and S. M. Nikol'skij. Integral Representations of Functions and Embedding Theorems. Moskva, Nauka, 1975.
- [21] O. V. Besov, L. D. Kudrjavcev, and P. I. Lizorkin. Studies on the theory of spaces of differentiable functions of several variables. *Proceedings of the Steklov Institute of Mathematics*, 182, 1990.
- [22] P. Borwein and T. Erdélyi. Polynomials and Polynomial Inequalities. Springer-Verlag New York, 1995.
- [23] D. Brigo and F. Mercurio. Interest rate models Theory and Practice: With Smile, Inflation and Credit. Springer, 2006.
- [24] Yu. A. Brudnyi. A multidimensional analog of a theorem of Whitney. Matematicheskii Sbornik, 1970.
- [25] V. Burenkov. Extension theorems for Sobolev spaces. Operator Theory: Advances and Applications, 109:187–200, 1999.
- [26] A. Caetano and W. Farkas. Local growth envelopes of Besov spaces of generalized smoothness. Zeitschrift für Analysis und Ihre Anwendungen, 25(3), 2006.
- [27] A. M. Caetano and S. Lopez. Homogeneity, non-smooth atoms and Besov spaces of generalised smoothness on quasi-metric spaces. *Dissertationes Mathematicae*, (460), 2009.
- [28] A. M. Caetano and S. D. Moura. Local growth envelopes of spaces of generalized smoothness: the critical case. *Mathematical Inequalities and Applications*, 7(4), 2004.

- [29] P.-L. Chow and J.-L. Jiang. Stochastic partial differential equations in Hölder spaces. Probability Theory and Related Fields, 99(1), 1994.
- [30] M. Clausel. Quelques notions d'irrégularité uniforme et ponctuelle: le point de vue ondelettes. PhD thesis, Université Paris XII, 2008.
- [31] M. Clausel and S. Nicolay. Some prevalent results about strongly monoholder functions. Nonlinearity, 23, 2010.
- [32] M. Clausel and S. Nicolay. Wavelet Techniques for Pointwise Anti-Hölderian Irregularity. Constructive Approximation, 33, 2011.
- [33] M. Clausel and S. Nicolay. A Wavelet Characterization for the Upper Global Hölder index. Journal of Fourier Analysis and Applications, 18(4), 2012.
- [34] S. Czerwik. On the stability of the quadratic mapping in normed spaces. Abh. Math. Sem. Univ. Hamburg, 62:59-64, 1992.
- [35] W. Dahmen and A. Kunoth. Multilevel preconditioning. Numerische Mathematik, 63, 1992.
- [36] I. Daubechies. Orthonormal bases of compactly supported wavelets. Communications on pure and applied mathematics, 41(7):909–996, 1988.
- [37] I. Daubechies. Ten Lectures on Wavelets. CBMS-NSF Reg. Conf. Series in Applied Math. SIAM, 1992.
- [38] H. Delange. Sur la fonction sommatoire de la fonction "somme des chiffres". Enseignement Mathematique, 21:31-47, 1975.
- [39] R. A. DeVore and G. G. Lorentz. Constructive Approximation. Springer-Verlag, 1993.
- [40] R. A. DeVore and B. J. Lucier. Wavelets. Acta Numerica, 1991.
- [41] R. A. DeVore and V. A. Popov. Interpolation of Besov spaces. Transactions of the A. M. S., 305(1), 1988.
- [42] R. A. DeVore and R. C. Sharpley. Maximal functions measuring smoothness. Memoirs of the American Mathematical Society, 47(293), January 1984.
- [43] S. Dispa. Beyond Besov spaces, S<sup>ν</sup> spaces: Topology and prevalent properties. PhD thesis, Université de Liège, 2006.
- [44] Z. Ditzian. Multivariate Bernstein and Markov inequalities. Journal of Approximation Theory, 70(3):273-283, 1992.

- [45] B. R. Ebanks, Pl. Kannappan, and P. . Sahoo. A common generalization of functional equations characterizing normed and quasi-inner-product spaces. *Canadian Mathematical Bulletin*, 1992.
- [46] D. E. Edmunds, P. Gurka, and B. Opic. On Embeddings of Logarithmic Bessel Potential Spaces. Journal of Functional Analysis, 146(1):116-150, 1997.
- [47] D. E. Edmunds and D. Haroske. Spaces of Lipschitz type, embeddings and entropy numbers. *Dissertationes Mathematicae*, 380:1–43, 1999.
- [48] L. C. Evans. *Partial Differential Equations*. American Mathematical Society, 1998.
- [49] W. Farkas. Function spaces of generalised smoothness and pseudo-differential operators associated to a continuous negative definite function, 2002.
- [50] W. Farkas and H.-G. Leopold. Characterisations of function spaces of generalised smoothness. Annali Di Matematica Pura Ed Applicata, 185:1–62, 2006.
- [51] H. Fejzić. On generalized Peano and Peano derivatives. Fundamenta Mathematicae, 143, 1994.
- [52] M. Fekete. Uber die Verteilung der Wurzeln bei gewissen algebraischen Gleichungen mit ganzzahligen Koeffizienten. Mathematische Zeitschrift, 17:228-249, 1923.
- [53] A. Fischer. Differentiability of Peano derivatives. Proceedings of the American Mathematical Society, 136(5), 2008.
- [54] I. M. Gel'fand and G. E. Shilov. Generalized Functions: Properties and Operations, volume 1. Academic Press Inc, 1964.
- [55] I. M. Gel'fand and G. E. Shilov. Generalized Functions: Spaces of Fundamental and Generalized Functions, volume 2. Academic Press Inc, 1968.
- [56] D. Gilbarg and N. S. Trudinger. Elliptic Partial Differential Equations of Second Order. Classics in Mathematics. Springer-Verlag, 2001.
- [57] L. Grafakos. Classical Fourier Analysis. Springer New York, 2008.
- [58] G. Hamel. Eine basis aller Zalhen und die unstetigen lösungen der Funktionalgleichung f(x+y) = f(x) + f(y). Math. Ann., 60:459–462, 1905.
- [59] Q. Han and F. Lin. Elliptic Partial Differential Equations. Courant Institute of Mathematical Sciences, New-York, second edition, 2011.
- [60] D. Haroske and S. D. Moura. Continuity envelopes of spaces of generalised smoothness, entropy and approximation numbers. *Journal of Approximation Theory*, 128(2):151–174, 2004.

- [61] D. Haroske and H. Triebel. Wavelet bases and entropy numbers in weighted function spaces. Mathematische Nachrichten, 278(1-2):108-132, 2005.
- [62] J. C. Hull. Options, Futures, and Other Derivatives. Prentice Hall, 9th edition, 2014.
- [63] S. Jaffard. Pointwise smoothness, two microlocalization and wavelet coefficients. *Publicacions Matemàtiques*, 35:155–168, 1991.
- [64] S. Jaffard. Wavelet Techniques in Multifractal Analysis. In Fractal Geometry and Applications: Multifractals, probability and statistical mechanics, applications, volume 72. American Mathematical Society, 2004.
- [65] S. Jaffard and Y. Meyer. Wavelet Methods for Pointwise Regularity and Local Oscillations of Functions. *Memoirs of the American Mathematical Society*, 1996.
- [66] S. Jaffard and Y. Meyer. On the pointwise regularity of functions in critical Besov spaces. Journal of Functional Analysis, 175(2):415–434, 2000.
- [67] S. Jaffard and S. Nicolay. Pointwise smoothness of space-filling functions. Applied and Computational Harmonic Analysis, 26, 2009.
- [68] K. Jordán. Calculus of Finite Differences. AMS Chelsea Publishing Series. Chelsea Publishing Company, 1965.
- [69] N. Kalton, S. Mayboroda, and M. Mitrea. Interpolation of Hardy-Sobolev-Besov-Triebel-Lizorkin spaces and applications to problems in partial differential equations. In Interpolation theory and applications : a conference in honor of Michael Cwikel, March 29-31, 2006, and AMS special session on interpolation theory and applications, AMS sectional meeting, Florida International University, April 1-2, 2006, Miami, Florida, 2006.
- [70] Pl. Kannappan. Functional Equations and Inequalities with Applications. Springer-Verlag, 2009.
- [71] H. Kestelman. On the functionnal equation f(x + y) = f(x) + f(y). Fundamenta Mathematicae, 1947.
- [72] A. Khintchine. Über einen Satz der Wahrscheinlichkeitsrechnung. Fund. Math., 6:9-20, 1924.
- [73] S.G. Krantz. Lipschitz spaces, smoothness of functions, and approximation theory. Expositiones Mathematicae, 1983.
- [74] D. Kreit. Constructions de Mouvements Browniens et Généralisations aux Feuilles Browniennes. Master's thesis, Université de Liège (ULG), 2009.

- [75] D. Kreit. Copules pour processus stochastiques financiers: étude de la dépendance entre risque action et risque taux d'intérêt. Master's thesis, Université Catholique de Louvain (UCL), 2013.
- [76] D. Kreit and S. Nicolay. Some characterizations of generalized Hölder spaces. Mathematische Nachrichten, 285:2157–2172, 2012.
- [77] D. Kreit and S. Nicolay. Characterizations of the elements of generalized Hölder-Zygmund spaces by means of their representation. *Journal of Approximation Theory*, 172:23–36, 2013.
- [78] D. Kreit and S. Nicolay. Generalized pointwise Hölder spaces. ArXiv eprints:1307.3140, 2013.
- [79] T. Kühn, H.-G. Leopold, W. Sickel, and L. Skrzypczak. Entropy numbers of embeddings of weighted Besov spaces. *Constructive Approximation*, 23, 2005.
- [80] T. Kühn, H.-G. Leopold, W. Sickel, and L. Skrzypczak. Entropy numbers of embeddings of weighted Besov spaces II. Proceedings of the Edinburgh Mathematical Society (Series 2), 49(02):331-359, 2006.
- [81] T. Kühn, H.-G. Leopold, W. Sickel, and L. Skrzypczak. Entropy numbers of embeddings of weighted Besov spaces III. Weights of logarithmic type. *Constructive Approximation*, 255, 2007.
- [82] P. G. Laird and R. McCann. On some characterization of polynomials. Am. Math. Mon., 91(2):114–116, 1984.
- [83] P. G. Lemarié and Y. Meyer. Ondelettes et bases hilbertiennes. Revista Matematica Iberoamericana, 2(1-2), 1986.
- [84] P. G. Lemarié-Rieusset. Recent Developments in the Navier-Stokes Problem, volume 431 of Chapman & Hall/CRC Research Notes in Math. Chapman & Hall/CRC, Boca Raton, FL, 2002.
- [85] H. G. Leopold. Embeddings and entropy numbers in Besov spaces of generalized smoothness. In Function Spaces: The Fifth Conference, volume 213 of Lecture notes in pure and applied mathematics. M. Dekker, 2000.
- [86] J. L. Lions and J. Peetre. Sur une classe d'espaces d'interpolation. Publications mathématiques de l'I.H.É.S., 19, 1964.
- [87] J. S. Lipinski. On zeros of a continuous nowhere differentiable function. American Mathematical Monthly, 73(2):166-168, 1966.
- [88] G. G. Lorentz. Approximation of Functions. Holt, Rinehart and Winston, Inc, 1966.

- [89] S. Mallat. A Wavelet Tour of Signal Processing, Third Edition: The Sparse Way. Academic Press, 2009.
- [90] X. Mao. Stochastic Differential Equations and Applications. Horwood Publishing Limited, 1997.
- [91] A. Marchaud. Sur les dérivées et sur les différences des fonctions de variables réelles. Journal de mathématiques pures et appliquées, 6:337–426, 1927.
- [92] J. Marcinkiewicz and A. Zygmund. On the differentiability of functions and summability of trigonometrical series. *Fundamenta Mathematicae*, 26, 1936.
- [93] Y. Meyer. Ondelettes et opérateurs I. Actualités Mathématiques. Hermann, 1990.
- [94] Y. Meyer, F. Sellan, and M. S. Taqqu. Wavelets, generalized white noise and fractional integration: the synthesis of fractional Brownian motion. *Journal of Fourier Analysis and Applications*, 5:465–494, 1999.
- [95] S. E. A. Mohammed. Stochastic Functional Differential Equations. Pitman Publishing Limited, 1984.
- [96] M.S. Moslehian and T.M. Rassias. Stability of functionnal equations in nonarchimedean spaces. Applicable Analysis and Discrete Mathematics, 1:325–334, 2007.
- [97] S. D. Moura. On some characterizations of Besov spaces of generalized smoothness. Mathematische Nachrichten, 280, 2007.
- [98] P. Mörters and Y. Peres. Brownian Motion. Cambridge University Press, 2010.
- [99] B .Opic and W. Trebels. Bessel Potentials with Logarithmic Components and Sobolev-Type Embeddings. Analysis Mathematica, 26(4):299–319, 2000.
- [100] A. Ostrowski. Jahresberichte d. Deutscher Mathematiker Vereinigung, 38:56, 1929.
- [101] B. H. Qui. Weighted Besov and Triebel spaces: Interpolation by the real method. *Hiroshima Mathematical Journal*, 12:581–605, 1982.
- [102] D. Revuz and M. Yor. Continuous Martingales and Brownian Motion. Springer, 1999.
- [103] C. H. Richardson. An Introduction to the Calculus of Finite Difference. Van Nostrand, 1954.
- [104] W. Rudin. Functional Analysis. McGraw-Hill, Inc, 1991.
- [105] S. Ruziewicz. Une application de l'équation fonctionnelle f(x+y) = f(x) + f(y) à la décomposition de la droite en ensembles superposables, non mesurables. Fundamenta Mathematicae, pages 92–95, 1924.

- [106] J. Schmets. Analyse mathématique, introduction aux espaces fonctionnels. Notes du cours de la seconde candidature en sciences mathématiques ou en sciences physiques (Université de Liège), 2004-2005.
- [107] S. R. Schubert. On a function of Van Der Waerden. American Mathematical Monthly, 70(4):402, 1963.
- [108] L. Schwartz. Théorie des distributions, volume 1. Hermann, 1950.
- [109] L. Schwartz. Théorie des distributions, volume 2. Hermann, 1951.
- [110] A. Shidfar and K. Sabetfakhri. On the continuity of Van Der Waerden's function in the Hölder sense. American Mathematical Monthly, 93(5):375–376, 1986.
- [111] W. Sierpiński. Sur l'équation fonctionnelle f(x + y) = f(x) + f(y). Fundamenta Mathematicae, 1920.
- [112] W. Sierpiński. Sur une propriété des fonctions de M. Hamel. Fundamenta Mathematicae, 1924.
- [113] F. Skof. Proprieta locali e approssimazione di operatori. Rend. Sem. Mat. Fis. Milano, 53:113-129, 1983.
- [114] E. M. Stein. Singular Integrals and Differentiability Properties of Functions. Princeton University Press, 1970.
- [115] T. Takagi. A simple example of the continuous function without derivatives. Proceedings of the Physico-Mathematical Society of Japan, 1:176–177, 1903.
- [116] L. Tartar. An Introduction to Sobolev Spaces and Interpolation Spaces. Springer-Verlag Berlin Heidelberg, 2007.
- [117] H. Triebel. Interpolation Theory, Function spaces, Differential Operators. North-Holland Publishing company, 1978.
- [118] H. Triebel. Theory of Function Spaces. Birkhäuser Verlag, 1983.
- [119] H. Triebel. Theory of Function Spaces II. Birkhäuser Verlag, 1992.
- [120] H. Triebel. A note on wavelet bases in function spaces. Banach Center Publications, 2004.
- [121] H. Triebel. Theory of Function Spaces III. Birkhäuser Verlag, 2006.
- [122] H. Triebel. Bases in Function Spaces, Sampling, Discrepancy, Numerical Integration. European Mathematical Society, 2010.
- [123] J. R. Trollope. An explicit expression for binary digital sums. Mathematics Magazine, 41:21-25, 1968.

- [124] S. M. Ulam. Problems in Modern Mathematics. Wiley, 1960.
- [125] S. E. Whittaker and G. Robinson. The calculus of observations. Blackie And Son Limited, 1924.
- [126] D. Xu, Z. Yang, and Y. Huang. Existence-uniqueness and continuation theorems for stochastic functional differential equations. *Journal of Differential Equations*, 245(6), 2008.
- [127] A. Zygmund. Smooth functions. Duke Mathematical Journal, 12(1), 1945.